## Understanding the environmental impact of product returns in the fashion industry

The development of a Composite Indicator to provide insights into the environmental and economic performance of product returns for fashion companies

T.J. Veen July 2023





## Understanding the environmental impact of product returns in the fashion industry

#### The development of a Composite Indicator to provide insights into the environmental and economic performance of product returns for fashion companies

by

#### T.J. Veen

to obtain the degree of Master of Science at the Delft University of Technology to be defended publicly on Tuesday, July 11, 2023.

#### Master thesis

	Author	T.J. (Tijn) Veen
	Student number e	4554655
	MSc	
	University	Transport, Infrastructure, and Logistics
	Date defended	Delft University of Technology
		July 11, 2023
Thesis committee	First supervisor	
	Second supervisor	Dr. J.M. (Jaap) Vleugel
	Formal chair	Dr. W.W.A. (Wouter) Beelaerts van Blokland
		Prof. Dr. R.R. (Rudy) Negenborn
	External supervisor	D. (Daniel) Visser
	External supervisor	M. (Myrthe) Rustemeijer
	External supervisor	M. (Mahi) Natani





## Preface

This thesis marked the end of my studies and was written to complete the Master's degree in Transport, Infrastructure, and Logistics at the Delft University of Technology. This research was performed together with Metyis, a data-driven strategy consultancy, who allowed me to combine my interests in sustainability and fashion with my academic skills.

Metyis has allowed me to expand research on improving the fashion industry's environmental impact, which is highly relevant as it is known for its large footprint. My thesis aimed to provide insight into the impact created by product returns on the performance of fashion companies and thereby help them improve their environmental footprint while remaining economically viable.

The interactive collaboration with Metyis has shown me what is needed for companies to effectuate change, as both environmental and economic dimensions are relevant in decision-making. As a result, this research fills scientific research gaps and contributes to practice. The findings of this research will be used by Metyis as a foundation for new sustainable business development and assist companies in improving the impact of their product returns.

This thesis would not have been realized and completed without my graduation committee. First, I would like to thank Daniel Visser for your dedication, support, valuable feedback, and for sharing your expertise. Besides, your talent for connecting the red line and helping to structure my thoughts is something I am very thankful for and have taught me a lot. Secondly, I want to thank Myrthe Rustemeijer for sharing knowledge of the logistics within the fashion supply chain and helping me reach out to experts for interviews. I am also thankful to Mahi Natani for supporting me while writing my thesis. Besides, I am very thankful for everybody at Metyis who helped me to obtain information and made my time at Metyis enjoyable. I am thankful for the opportunities and resources provided by Metyis.

Further, I want to thank all my supervisors from the University for their guidance during my research. Firstly, I would like to thank Jaap Vleugel for your enthusiasm, support, and valuable advice throughout the process, which has helped me academically and personally. Next, I want to thank Wouter Beelaerts van Blokland for your supportive feedback and useful insights and for challenging me to think outside the box. Also, I am grateful for the guidance of Rudy Negenborn for your critical observations and valuable suggestions, which have supported me in improving and refining my research.

Last but not least, a big thank you to my family, friends, and E for always being there for support and motivation throughout my studies.

I hope you enjoy reading this research, and it will make you think twice about the environmental impact you create before returning an item.

> T.J. Veen Amsterdam, July 2023

### Summary

The fashion industry is a worldwide driver for economic growth [81]. With a growth rate of 9.93% per year (CAGR 2023-2027) [172], further growth is expected in the years to come, particularly within e-commerce [98, 81, 22]. At the same time, the fashion industry's impact ranks the fourth largest on climate change within the European Union (EU), and action is needed to lower its environmental footprint [63, 72, 182]. Most research and efforts focus on sustainability practices in the forward value chain to comply with European Commission (EC) regulations and increased customer pressure [11, 15]. Less attention is paid to returns, although these have become an increased problem in this industry, particularly [36, 183, 15, 66]. High return rates between 25% and 60% in fashion are observed, which are expected to increase further due to the industries' growth [50, 175, 36, 4, 36]. Besides the economic losses for companies due to returns [175], these returns create additional Greenhouse Gas (GHG) emissions, thereby negatively affecting society and the environmental footprint of companies [36, 175, 50, 145, 191].

Returns may account for approximately 12% of the total GHG emissions of fashion companies their operations [9], and they may yearly lose roughly 17% of 1 billion dollars in sales to returns [3, 135]. The growing problem of returns should be tackled by both lowering the environmental impact of the process of unavoidable returns and the number of returns [36, 63], leading to environmental gains for society and cost savings for companies [194]. To accomplish this and effectuate change, companies need transparency and understanding of these returns' environmental and economic impact on their performances [22, 202]. This knowledge is essential for companies to design, analyze, and implement effective strategies to improve their environmental impact while remaining economically viable [23, 36, 182, 202]. However, the impact of product returns is poorly understood, and there is a lack of accurate methods to assess this impact [183, 36].

This research addresses this gap by developing a new performance measurement tool for fashion companies to measure, analyze, and evaluate returns' environmental and economic impact. A new Composite Indicator (CI) was constructed that included all return activities and considered the differences in the return methods. Companies can make well-informed decisions to improve their returns' impact, as environmental and economic performance are considered [182]. In addition, a new benchmark based on GHG reduction targets of the EC for 2030 was designed, developed, and integrated into this CI, allowing companies to compare their performance relative to the targets they must comply with by 2030. With this, a transparent tool was developed where customers and governments know which- and how the environmental impact of a product return is measured for a company's performance. The main question of this research was:

#### "How can a Composite Indicator be constructed to provide insight into the environmental and economic performance of product returns for companies within the fashion industry and help improve their sustainability?"

The CI was constructed in five steps, where the performance is measured using indicators. First, it was determined which environmental and economic indicators needed to be included in this CI to measure the impact of product returns on a company's performance. This was done using a literature study and expert consultation, and the findings were combined into a theoretical framework. Second, the indicators were normalized by rescaling. Third, the Analytical Hierarchy Process (AHP) technique was applied to weigh each indicator. The indicators were aggregated into one measure in the fourth step using the multiplication technique, followed by evaluation of the robustness of the CI using a Uncertainty Analysis (UA) and a Sensitivity Analysis (SA). Lastly, a benchmark was designed using literature, reports, and expert consultation and developed and integrated into the constructed CI, where AHP weighting was replaced by Equal Weights (EW).

The constructed CI was verified and validated using a case study of a fashion retailer. The results showed that fashion companies could lower their environmental footprint- and costs by using the insights from the CI. Analysis of the performance showed considerable differences in the environmental and economic performance of the return methods. For example, from the analyzed performance of this fashion retailer, it follows that the return method of home pick-up is preferred as it performs best on environmental and economic performance, and in-store returns should be avoided as an alternative for returns. Inspection of each indicator

showed that for home pick-up, transport emissions per return are due to the more efficient transport networks compared to the other return methods. However, home pick-up is more costly as longer distances are driven with relatively expensive distribution vans. In-store returns have lower transport costs as customers bring the product to the brick-and-mortar store, and short-haul transport with expensive distribution vans can be avoided but score worst on transport emissions as no consolidated shipments are used. The obtained insights can be used to define strategies for companies to improve their impact and encourage conscious return behavior of customers. This leads to environmental and economic benefits for a company, as more conscious return choices are likely to be made, and opportunistic return behavior can be limited [194, 22, 23]. Further, the performance was compared to a benchmark that expressed EU target values for 2030. With these insights, knowledge is gained on how far the company is in reaching the goals of the EC to combat climate change. Scenarios were designed using the PESTLE framework, which showed the practical usage of the CI to identify improvement areas and economic risks on their performance. Insights based on the data used in this research show that the most considerable improvements in overall performance can be achieved when fewer products are returned due to more conscious behavior and more packaging material is recycled. Also, charging a return fee to discourage customers from returning items leads to performance improvements and is crucial to lowering costs when governmental financial support is lacking. The insights gathered from evaluat-

ing the results of the CI can be used by companies to make more informed decisions, prioritize improvement

efforts, and design actions to improve sustainability while ensuring economic feasibility.

The main recommendation of this research is for companies to offer return methods to customers based on their performance and avoid the worst-performing methods as an alternative to returning a product to improve their environmental impact. It is recommended that governments use this transparent tool to compare companies' performances sector-wide. Future research should explore how- and where companies in the purchasing- and return process can use the findings of this study best to encourage customers' conscious return decisions- and behavior most effectively. Next, it is recommended to design return policies using financial discouragements based on the findings of this research. The effectiveness of these policies as mitigation strategies for returns in practice should be explored, where research should be conducted into the effect of implementing financial discouragements on customer demand- and, thereby, the company's profitability. Following, investigating a company's decision-making process on the final destination of a returned product is recommended, as this affects performance. Future research should be done to increase the adoption potential of textile recycling, where both the technical and practical feasibility should be considered for fashion companies. The final recommendation for future research is to investigate governmental policies' role in lowering the number of product returns in fashion and to identify which approach would be most effective, considering legislation and financial support. Here, the designed benchmark and CI of this research are recommended to be included to investigate the effectiveness of implementing these policies and monitor the effects on companies' performances.

## Contents

1	Intr	roduction 1
	1.1	Research background
	1.2	Research problem definition
		1.2.1 Scientific research problems
		1.2.2 Practical research problems   3
	1.3	Research scope   3
		1.3.1 System boundaries
		1.3.2 Practical limitations
		1.3.3 What environmental impact is assessed
		1.3.4 What economic impact is assessed
		1.3.5 Conceptual understanding of the research
	1.4	Research objective
	1.5	Research questions
	1.6	Research design.   6
		1.6.1   Research set-up   6
		1.6.2 Research outline   7
		1.6.3 Expected research output   7
		1.6.4   Data sources   8
2	Pro	duct returns and their impact 9
	2.1	Why fashion products are returned
	2.2	How the process of product returns works
	2.3	How a customer can return a product
	2.4	How a product return creates environmental impact
		2.4.1 How the environmental impact is measured
		2.4.2 Phase I. Return preparation
		2.4.3 Phase II. Collection
		2.4.4 Phase III. Product handling and processing
		2.4.5 Phase IV. Recovery and disposition of products
	2.5	How a product return creates costs
	2.6	Main findings chapter 2
3	Met	21
	3.1	Why performance should be measured
		How performance is measured by indicators
	3.3	How a Composite Indicator can be constructed
		3.3.1 Phase 1. Identification and selection of indicators
		3.3.2 Phase 2. Normalization of indicators
		3.3.3 Phase 3. Weighting of indicators
		3.3.4 Phase 4. Aggregation of indicators
		3.3.5 Phase 5. Post-analysis evaluation of the CI
	3.4	How a construction method can be selected
	3.5	Why benchmarking should be performed
	3.6	Main findings chapter 3

4	Conceptual design of the Composite Indicator4.1 Objective and scope of the CI	30 31 31 32 33 34 34 35
5	Measurement with the Composite Indicator5.1Description of the base case5.2Data preparation and analysis5.3Design of the CI.5.3.1Phase 1. Selection of the indicators5.3.2Phase 2. Normalization of the indicators.5.3.3Phase 3. Weighting of the indicators5.3.4Phase 4. Aggregation of the indicators5.3.5Phase 5. Post-analysis evaluation of the CI.5.4Conclusions performance measurement5.5Main findings chapter 5.	42 45 46 46 46 47 47
6	Design   6.1 PESTLE analysis.   6.2 Design benchmark   6.2.1 Description of the benchmark   6.3 Benchmark testing   6.4 Conclusions performance measurement with the benchmark   6.5 Verification   6.6 Main findings chapter 6.	51 52 52 53 55
7	Evaluate7.1Design and set-up scenarios7.2Scenario testing.7.2.1Phase 1. Selection of the indicators7.2.2Phase 2. Normalization of the indicators.7.2.3Phase 3. Weighting of the indicators7.2.4Phase 4. Aggregation of the indicators7.2.5Phase 5. Post-analysis evaluation of the CI.7.3Conclusion performance measurement scenario testing7.4Validation.7.5Main findings chapter 7.	61 61 61 61 62 65 67
8	Conclusion and Discussion   8.1 Main findings	71 71 71 72
Α	Scientific research paper	91
В	Preference scale of AHP	107
С	Dimensions of a package	109
D	Context data	111

Е	Environmental data	113
$\mathbf{F}$	Cost data	115
$\mathbf{G}$	Performance gaps	117
Н	Uncertainty and sensitivity analysis	119
	H.1 Rescaled pairwise comparison matrix.	.119
	H.2 Data on the sensitivity analysis	.119
	H.2.1 Raw data after increasing and decreasing the variables	.119
	H.2.2 Aggregated values CI	.120
Ι	Scenario analysis	123

## Acronyms

- 3PL Third Party Logistics Provider
- AHP Analytical Hierarchy Process
- NCA Non-compensatory multi-criteria analysis
- LDPE Low-density polyethylene
- CBS Centraal Bureau voor de Statistiek
- DEA Data Envelopment Analysis
- MRQ Main Research Question
- GWP Global Warming Potential
- CI Composite Indicator
- **CR** Consistency Ratio
- **UA** Uncertainty Analysis
- SA Sensitivity Analysis
- GHG Greenhouse Gas
- EC European Commission
- EF Emisison factor
- EW Equal Weights
- EU European Union
- LCA Life Cycle Assessment
- LNG Liquid Natural Gas
- LIN Linear aggregation
- GME Geometric mean

## Definitions

#### Aggregation

Combining multiple variables or indicators into one measure.

**AHP** This is a weighting method where weights are assigned based on trade-offs made by experts [133].

#### Benchmark

A score to which performances are compared, where the benchmark reflects a target or an ideal situation.

#### **Brick-and-mortar store**

The physical store of a fashion retailer where customers can buy or return items.

#### Compensability

If there is compensability, it is allowed that the value of one variable may compensate for the value of another variable. There are certain levels of compensability in which the level of this trade-off between the values is determined [128].

#### **Customer return**

A return after the product has been delivered to the customer [45], also referred to as product returns in this research.

CO2e This represents CO2 equivalent emissions, commonly including all GHG emissions [84].

#### **Composite indicator**

A Composite Indicator consists of multiple indicators aggregated into one measure in a structured way, where multiple dimensions (i.e., environmental, economic) can be analyzed and compared [206].

#### **Distribution center**

A building where packages are redistributed to be transported to retailers, wholesalers, or customers.

#### **E-commerce**

Online shopping channel.

**EF** An emission factor is a value for a specific activity, material, process, or product representing the amount of GHG emissions emitted by this activity, material, process, or product [84].

**EW** Equal weighting means that the same weight is assigned for all the considered variables [133].

#### **Forward logistics**

The logistics network is designated from raw materials to delivering the product to the customer or retailer [45].

#### Fraudulent returns

Returns that do not conform to legal rules or policies, for example, after wearing an item [91].

#### **GHG emissions**

The Greenhouse Gas emissions negatively affect climate change and consists of carbon dioxide  $(CO_2)$ , nitrous oxide  $(N_2O)$ , methane  $(CH_4)$ , sulfur hexafluoride  $(SF_6)$ , hydrofluorocarbons (HFCs), perfluorocarbons (PFCs), and nitrogen trifluoride  $(NF_3)$  [85].

#### Indicator

A variable that can measure the performance of a process, system, or action [206].

**LCA** A life cycle analysis assesses the emissions created over the whole life cycle of a product, including the moment of cradle until it goes into the grave [130].

#### Legitimate return

Returns made by a customer who intends to keep a product when they purchase it but eventually decides to bring it back [146].

#### Normalization

Converting data to a comparable range or scale by transforming the unit of measurement [150].

#### **Opportunistic returns**

When the decision to return is already made before purchasing the product, for example, when different sizes or colors are bought of an identical product [123].

#### **PESTLE framework**

This is short for Political (P), Economic (E), Social (S), Technological (T), Legal (L), and Environmental (E). This framework is used to get an overview of contextual factors [148, 198].

#### Performance gap

Quantification of the difference between two values, commonly between target and actual performance.

#### **Product return**

See consumer return.

#### **Radar chart**

A chart with multiple axes, represented by spokes. Each spoke represents a different variable, making it possible to combine and analyze multiple data points in one figure.

#### **Reverse logistics**

The reverse flow of goods in a supply chain, where a product is transported from the customer back to the final destination (point of recovery or a different user) [45].

#### Rescaling

Normalization technique based on minimum and maximum values in a data set [134].

#### **Sensitivity Analysis**

Increasing and decreasing input values for the model to examine the effect on the final value of the model.

#### **Third Party Logistics**

The external party that provides transport of packages, including the logistic services. Examples of these are PostNL, UPS, and DHL.

#### **Uncertainty Analysis**

The process of analyzing the uncertainty sources in the model and examining their effect on the output [133].

#### Validation

Validation investigates whether the model is right by looking at the requirements and objectives [143].

#### Verification

Verification investigates whether the constructed model is the right model based on the conceptual model [104].

## List of Tables

		33 34
5.1 5.2	Raw indicator values for each return method in the base case, source: (Author)	45
		45
5.3		46
5.4		46
5.5	0	47
5.6		47
5.7	Environmental performance score and economic performance score of the return methods, source: (Author)	48
6.1	Raw indicator values base case with the benchmark, source: (Author)	53
6.2		53
6.3		53
6.4		54
7.1	Raw indicator values for each scenario, source: (Author)	61
7.2		62
7.3		62
7.4	8	63
7.5		63
7.6		64
7.7		65
7.8	Outcomes correlation coefficient for uncertainty trigger $X_3$ , source: (Author)	66
B.1	Preference/importance scale AHP, from [167]	07
C.1	Dimensions for the most common sizes of packaging, adapted from [34, 185, 188] 19	09
D.1	Context data	11
E.1	Data for the calculation and measurement of emissions	13
F.1	Cost data	15
G.1	Overview of relative performance and performance gap, source: (Author)	17
H.1	Pairwise comparison matrix rescaled weights, source: (Author)	19
H.2	Raw data after increasing the equipment costs by 25%, source: (Author)	20
	Raw data after decreasing the equipment costs by 25%, source: (Author)	
	Raw data after increasing the transport cost of electric vans by 25%, source: (Author) 12	
	Raw data after decreasing the transport cost of electric vans by 25%, source: (Author) 1	
	Raw data after increasing the warehouse emissions value by 25%, source: (Author) 1	
	Raw data after decreasing the warehouse emissions value by 25%, source: (Author) 1	
	Raw data after increasing the landfill cost by 25%, source: (Author)	
	Raw data after decreasing the landfill cost by 25%, source: (Author)	
	0 Raw data after increasing the incineration cost by 25%, source: (Author)	
11.1	1 Raw data after decreasing the incineration cost by 25%, source: (Author)	22

	H.12Aggregated scores CI after p	performing the sensitivity	y analysis, source: (Author)	
--	----------------------------------	----------------------------	------------------------------	--

## List of Figures

1.1 1.2 1.3	Research scope, source: (Author)	4 5 7
2.1 2.2	Return processes and sub-activities, adapted from [42]	10
	ods, source: (Author)	11
2.3	Transport structure general overview, adapted from [77]	13
2.4	Product handling and processing process, adapted from [45, 42]	16
3.1	Construction of a CI, adapted from [203, 10]	22
3.2	Flow chart to select the best methods for constructing Composite Indicators, from [120]	26
4.1	Conceptual design CI measuring the impact of product returns on companies' environmental and economic performance, source: (Author)	31
5.1	Radar chart of the different return methods in the base case, source: (Author) $\ldots$	49
6.1	Radar chart base case with the benchmark, source: (Author)	54
7.1	Overview of scenario design, source: (Author)	58
7.2	CI scores obtained with original and rescaled weights, source: (Author)	64
7.3		
	source: (Author)	65
7.4	Performance gaps scenarios relative to the benchmark, source: (Author)	66
I.1	Radar chart benchmark and base case performance, source: (Author)	123
I.2	Radar chart benchmark and scenario 1, source: (Author)	124
I.3	Radar chart benchmark and scenario 2, source: (Author)	
I.4	Radar chart benchmark and scenario 3, source: (Author)	
I.5	Radar chart benchmark and scenario 4, source: (Author)	
I.6	Radar chart benchmark and scenario 5, source: (Author)	126

# 1

## Introduction

This chapter introduces the research focusing on improving and understanding the environmental impact of product returns in the fashion industry. The structure of this chapter is as follows: first, the background of the research is elaborated on, followed by the scientific and practical problems. Next, the research scope, objectives, questions, and design are described.

#### 1.1. Research background

The fashion industry is a worldwide driver for economic growth and has transformed significantly over the past decades [81, 63]. With a growth rate of 9.93% per year (CAGR 2023-2027), further growth is expected in the years to come, with a market volume of 1.382 billion euros predicted by 2027 [172]. A driver for this growth is the notable shift towards online shopping in the fashion industry [36]. With COVID-19 acting as a catalyst and further digitalization developments, it is estimated that the total sales of e-commerce will primarily increase in fashion [81, 22]. E-commerce offers benefits to customers, such as a more comprehensive selection of products and prices, access to reviews, multiple delivery options, and an easy return of unwanted products [163]. Fashion retailers benefit from this new sale channel as well because it can boost their sales, and they have access to a broader customer audience [175]. However, with the increase in e-commerce sales, the number of returned customer products in the fashion industry is growing as well [50, 145]. [175] mentions return rates of 50% in fashion e-commerce, [36, 4] find that 25% of all fashion products ordered online result in return, and for some product categories, returns rates of 60% are observed [37]. In comparison, less than 10% of products bought in offline retailing environments are returned [36].

In parallel to the industry's considerable market size, the fashion industry's impact ranks the fourth largest on climate change within the European Union (EU) [63]. Nearly 10% of worldwide Greenhouse Gas (GHG) emissions are attributable to the fashion industry, which is twice as much as all maritime and flight transport combined [112, 72]. Every second, a truckload of clothes is either incinerated or sent to landfills [52, 63]. The fashion industry must lower its environmental footprint to meet EU targets and respond to market pressures of customers that demand improved transparency on sustainability practices, while remaining profitable in a competitive environment [63, 182]. This makes, with the expected growth in demand and volume, sustainability a challenge and opportunity for the fashion industry [63]. The European Commission (EC) has implemented regulations and initiatives to lower the industry's environmental impact as part of their broader efforts to reach net zero by 2050 [48, 72]. An important initiative that supports this objective is the Green Deal. This initiative states that the GHG emissions of the fashion industry should be reduced by 50% by 2030 and eventually become climate neutral by 2050 [72, 63]. To support the Green Deal, the EC plans on introducing mandatory sustainability reporting for companies, in which companies need to provide transparency on their performances. The mandatory sustainability reporting will be a basis to monitor and control companies' environmental impact, identify improvement areas, and prioritize reduction efforts [62].

To lower its footprint, the fashion industry continuously adopts and integrates sustainability practices throughout the value chain [63, 182]. Most of these efforts are focused on the forward supply chain [15, 11], where most emissions along the value chain are created [12]. The reverse logistics, notably the effects of product returns, have received little attention [36, 183, 15]. However, product returns have become an increased problem in this industry, particularly [36, 183, 175]. Increased return activities have exacerbated its

environmental footprint, which is expected to rise further due to the industries' growth [50, 145]. Besides the economic losses for companies due to returns [175, 191], the EC also recognizes the environmental impact of returned products as a widespread problem within the EU [66]. So, actions should be taken to reduce this impact [63, 36].

Returns must be repackaged, transported, warehoused and stored, recovered, or disposed to landfill or incineration. This increases energy consumption, GHG emissions, and other forms of environmental waste [36, 22, 183]. A logistic firm estimates the GHG emissions of transporting returned items in the USA in 2022 to be bigger than those of 5.1 million cars driven for one year, which equals over 24 million metric tons of GHG emissions [178, 141]. 4.3 billion kilograms of returns end up in landfills in the USA alone, equivalent to 10500 fully loaded Boeing 747s [141]. Looking at the worldwide scale, it is evident that returns create a negative impact on the environment that only grows with the rise of e-commerce [50, 145]. Although returns are not the largest proportion of the industries' GHG emissions along the value chain [12], they create undesirable consequences for society and companies [33, 66].

Yet, the impact of returns is unjustly, often not taken seriously by companies. Effective management of these often lacks, even though returns create high costs and companies are accountable for the environmental impact they create [157, 50]. A report of a UK retailer, ASOS, analyzing their business in 2018-2019 estimates that 12% of their total operational emissions are accountable to returns, which increases as the volume of returns in their business grows [9]. With the current developments and e-commerce growth in the industry [98, 171], it is likely to assume that returns have become a larger share of the total operation emissions at the company level. In addition, returns are, in essence, costly for companies, as expenses are made while the product loses value due to its short fashion life cycle [22, 6, 183, 66]. The returns costs can sometimes go up to three times compared to their original price [183]. A study by the National Retail Federation has found that for returns in the USA, merchandise companies pay 166 million dollars for every 1 billion dollars in sales [3, 135], emphasizing the economic losses and, thus, the relevance for companies to lower their returns from an economic viewpoint as well.

Customer returns should not be overlooked in the fashion industry's transition to meet their GHG emission targets, which is the focus area of this research. Companies should take action to, on the one hand, lowering the number of returns and, on the other, improve the current process for unavoidable returns [36, 66]. Besides reducing negative externalities on society, improving returns can lead to economic gains for companies.

#### 1.2. Research problem definition

This section defines the scientific- and practical research problems.

#### 1.2.1. Scientific research problems

The scientific problem discussed in this research is that there is no understanding of the environmental impact of product returns. It is acknowledged that knowledge and transparency on the environmental impacts of product returns could improve sustainability [202, 23]. However, a method to measure this impact lacks [183, 36]. In the scarce attempts to assess product returns, not all relevant logistic activities involved in product returns are included. Research does not include the GHG emissions, and not all return methods available nowadays are considered, although this is decisive for the environmental impact created [51]. Also, the impact of returns on a company's performance is not known yet. There is no overview to compare the return methods based on their environmental performance in the literature, which would be needed to encourage more conscious return behavior of customers [22]. Therefore, it can be concluded that research into the environmental impact of product returns in this sector still shows shortcomings and further research into this topic is needed.

This research aims to add knowledge to the scientific literature on reverse logistics in the fashion industry, focusing on the environmental understanding and assessment of customer returns. Several contributions will be made in this study to scientific research. The first contribution is adding scientific research to the currently limited field of research on consequences and potential solutions to the environmental impact of product returns in the fashion industry. The second contribution is a new model for measuring the GHG emissions of product returns in the fashion industry that include all relevant logistic activities and return methods. The third contribution is the development of a new Composite Indicator (CI) for product returns that fashion companies can use to measure, analyze, and evaluate the impact on their environmental and economic performance. This study also contributes by expanding the field of research on company perfor-

mance measurement with CIs by assessing both environmental and economic performance [203]. Also, a new benchmark expressing target performance is developed based on reduction targets of the EC to combat climate change and integrated into the CI to benchmark the measured performance. The final contribution is that a new overview and measurement method is created where insight into the impact of different return methods is gathered. This makes the return methods comparable based on their environmental and economic performance. Although it is acknowledged in the literature that choosing a return method based on its performance might lead to improved environmental impact, this insight is currently lacking in scientific research [51, 22].

#### **1.2.2. Practical research problems**

Besides the economic losses of returns, fashion companies are responsible for the environmental impact they create [50], and with the increase in returns, this will be exacerbated. With the prospected mandatory environmental reporting and targets to combat climate change, companies must measure, improve, and give transparency on their performances to comply with EU regulations [62, 59, 72]. [12] notes that although returns are not the biggest cause of emissions along the value chain of a fashion company, there is abatement potential for companies to both lowering returns and improve the processes. Reducing returns by 20% saves 12 million tonnes GHG emissions, which an improved process of returns can further improve[12]. So, the problem of returns should be tackled by both lowering the environmental impact of the process of returns and the number of returns [36, 63], leading to environmental gains for society and cost savings for a company [194].

Environmental improvements can be made by adopting practices to lower the impact of each activity within the return process [36]. To do so, there is a need to understand the environmental and economic impact of product returns in the fashion industry and how this affects a company's performance. Companies lack the tools to measure the impact of product returns on their environmental and economic performance and thereby face issues in identifying areas of improvement. A return method can perform best based on environmental performance but can also be the highest-cost alternative for a company. This knowledge is essential for companies to design effective strategies to improve their environmental impact while remaining economically viable [23, 36, 182]. However, companies do not yet have the tool to analyze, assess, and compare the environmental impact of product returns on their performances [36]. Companies cannot identify areas of improve their reduction efforts, and measure and monitor their performance, making it difficult for them to improve the environmental impact of product returns.

Additionally, fashion customers increasingly demand transparency and information to make more environmentally friendly decisions. Informed customers are more likely to choose environmentally friendly return methods, leading to environmental gains for society as a whole [23, 12]. Also, creating awareness of this impact to customers might lead to fewer products being returned due to more sustainable shopping behavior in which customers are likelier to keep the products they buy [194]. Still, companies lack understanding and cannot inform or provide transparency about the environmental impact of return methods and their behavior [22, 202].

This research is used to fill the following practical knowledge gap:

There is a gap for fashion companies in understanding how product returns and different return methods impact their environmental and economic performance. It is unclear which actions they should take to lower the environmental impact of product returns while considering their economic feasibility. As a result of this shortcoming, companies cannot provide transparency about their environmental impact or inform customers to make more conscious choices for their returns to improve sustainability.

The practical objective is to develop a transparent tool for companies to understand and measure the impact of product returns on their performances. With this study, companies can see how implementing different actions or strategies would improve their environmental performance, gain detailed insights, and compare them with each other, while remaining profitable. Further, companies are provided with knowledge of how far they are in reaching the reduction targets of the EC to combat climate change and how different actions could contribute to this. Last, companies are provided the knowledge to inform their customers and encourage conscious return behavior.

#### 1.3. Research scope

This section discusses the research scope, which consists of the system boundaries of the process product returns and some practical limitations, which environmental and economic impact are measured. Lastly, a

conceptual overview is presented to improve understanding of the research.

#### 1.3.1. System boundaries

Returns happen throughout the supply chain and are categorized by [45] as manufacturing, distribution, and customer returns. This research only focuses on customer returns; this type of return happens when the product has been delivered to the final customer [45]. This research intends to focus on the whole process of a return and consider all logistic activities that are crucial in analyzing the environmental and economic impact. The boundary defined for this process of product returns in this research starts from the customer requesting a return until the disposition of the product is executed [45, 15]. This is visualized in Figure 1.1. With this boundary, the forward delivery of the product is excluded from this research.



Figure 1.1: Research scope, source: (Author)

The recovery and disposition of products are included in these process boundaries because it substantially contributes to the environmental impact of a product return and thereby affect a company's performance [15, 93]. Product disposal consists of recycling, landfill, incineration, and reselling activities [36]. These activities generate additional costs and emissions for a company that would not have been made if the product would not return [107]. So, including the disposition activities will provide a complete assessment of the environmental and economic impact of a product return on a company's performance.

#### **1.3.2. Practical limitations**

The returned products in this research examined are ordered at a fashion retailer, delivered to the customer, and returned to the retailer for one of the acceptable reasons to return. This research excludes used products that are returned for recycling purposes from the research scope, as these are commonly not returned to the same retailer when handed in for recycling, according to an expert in clothing recycling. Also, insights into this stream of used products are lacking for companies, according to an industry expert, and it is often not considered by fashion companies as part of their reverse streams.

The environmental and economic impact is assessed per product item returned. For practical reasons, it is chosen to employ an average-sized clothing item that fits in a medium size e-commerce parcel box. This box size is commonly used for transporting clothing items [34]. Further, the geographical scope considered in this research is customers based in the Netherlands. The distribution and accessibility to collection points, physical stores, and other facilities are within this geographical area if not mentioned otherwise.

Further, this research uses legislation and policies set by the EC, focusing on companies operating under European laws.

#### 1.3.3. What environmental impact is assessed

The EU has marked the reduction of GHG emissions in Europe as the top urgent and most critical priority to ensuring climate neutrality by 2050 [67]. This research aims to contribute to this goal and therefore focuses on measuring the environmental impact of GHG emissions. Measuring the GHG emissions rather than solely carbon emissions provides a complete overview and assessment of the environmental impact. The different activities that create and contribute to GHG emissions are identified and calculated. There are six GHG according to the Kyoto Protocol: carbon dioxide ( $CO_2$ ), nitrous oxide ( $N_2O$ ), methane ( $CH_4$ ), sulfur hexafluoride ( $SF_6$ ), hydrofluorocarbons (HFCs), perfluorocarbons (PFCs), and nitrogen trifluoride ( $NF_3$ ) [85, 84]. All these are harmful to the environment and contribute to climate change, so an accurate assessment is made by concentrating on the GHG emissions. Additionally, by focusing on the GHG emissions of logistic activities in their value chain, the measurements can be easily compared to the objectives and targets of the EC as they are in the same unit of measurement [60]. For example, by measuring transport GHG emissions, the targets can be easily translated, and progress can be tracked, providing relevant results for companies and customers [58]. Finally, a lot of reliable information, data, and tools are available for measuring GHG emissions, which helps improve the assessment's quality. Therefore, the environmental impact of product returns is measured by the GHG emissions. The GHG emissions are expressed in CO2 equivalent emissions (kgCO2e), also referred to as CO2e. This measure incorporates all main greenhouse gasses ( $CO_2$ ,  $N_2O$ ,  $CH_4$ ,  $SF_6$ , HFCs, PFCs, and  $NF_3$ ) into a single measurement unit [84]. These emissions are converted to CO2-equivalents according to the Global Warming Potential (GWP) [26, 84].

#### 1.3.4. What economic impact is assessed

This research only includes the company's costs to perform a product return, considering the scope of this process. According to industry experts, return costing by fashion companies is preferably activity-based, so only activities in the process are considered in the economic impact assessment. This is also consistent with how the environmental impact is assessed, which is also on an activity basis. The lost revenues due to returns are not included in this analysis, as this is difficult to estimate for the level of detail of this research, negatively affecting the assessments' quality. The costs are determined in euros per returned item. The base case data originates from 2020, defining the (if needed) conversion rates.

#### 1.3.5. Conceptual understanding of the research

To get a better idea of the process of returns, what is evaluated within this research, and how this is quantified, Figure 1.2 is created. This is a conceptual model explaining the return phases, the activities that can happen within each phase, and how the environmental and economic impact is measured in this research. All elements will be elaborated in more detail in chapter 2.



Figure 1.2: Conceptual model of the scope of this project, source: (Author)

#### 1.4. Research objective

The objective of this research follows from the reviewed literature and defined problem statements. This study aims to assess the GHG emission impact and economic costs of product returns in fashion and to create an understanding of this impact on a company's performance. With these insights, a measurement method to analyze the environmental impact of product returns is created, closing this scientific research gap. Also, companies are provided insights on improving the environmental footprint of returns, as both environmental and economic dimensions are included to ensure feasible decision-making. The outcome is a CI that assesses the impact of the process of returns using indicators that quantify the activities within the process, considering the return method chosen.

The following research objective is defined:

To develop a measurement approach fashion companies can use to assess and evaluate the environmental and economic impact of product returns on their performance, that includes all relevant activities within the process of returns and return methods, and apply this to practice by developing a Composite Indicator (CI) for fashion companies to improve sustainability.

#### **1.5. Research questions**

The Main Research Question (MRQ) is formulated following the research gaps and objectives:

"How can a Composite Indicator be constructed to provide insight into the environmental and economic performance of product returns for companies within the fashion industry and help improve their sustainability?"

The following sub-questions are set up to answer the MRQ:

- 1. What does the fashion industry's product return flow look like?
  - (a) What return methods can be used to return a fashion product?
  - (b) How do the activities of the return process differ depending on the chosen return method?
- 2. How can the environmental impact of the various logistic activities involved in product returns be assessed?
  - (a) What are the key factors that should be considered for assessing the environmental impact of these logistic activities?
  - (b) How can the environmental impact of these factors be measured?
- 3. How can the economic impact of the various logistic activities involved in product returns be assessed?
  - (a) What economic factors should be considered for assessing the economic impact of these logistic activities?
  - (b) How can the costs of these factors be measured?
- 4. How can a CI be created to assess the environmental and economic impact of product returns?
  - (a) How can companies use a CI to assess their performance on multiple dimensions?
  - (b) Which steps need to be performed to construct a CI for a multi-dimensional assessment?
  - (c) What methods should be selected to construct the CI for product returns assessing environmental and economic performance?
  - (d) How can the current state of a company's performance be measured using the CI?
  - (e) How can the constructed CI be verified?
  - (f) How can fashion companies use the constructed CI in practice?
  - (g) How can the constructed CI be validated?
- 5. What is a company's most preferred return method considering its environmental and economic performance?

#### 1.6. Research design

The research uses a structured set-up, which is elaborated on first. Based on this set-up, the research outline is presented. A summary of the expected research output is given, followed by an overview of the data sources used in this research.

#### 1.6.1. Research set-up

This research involves a combination of qualitative and quantitative data analyses. The research structure is based on the DMADV process (Define, Measure, Analyze, Design, Verify) [184], which has its roots in the Six Sigma theory. This is a general framework for structuring research development using an organized approach. DMADV is used to design or improve new products, processes, or services [165]. As this research intends to develop a new model instead of improving a product or process, adaptions are made to this framework in consultation with W.W.A. Beelaerts van Blokland (2023).

For this research, the Measure phase (M) is replaced by a Conceptual phase (C). This research designs a new measurement method, so a conceptual model must be developed first. Further, the last step of Verification (V) is replaced by Evaluation (E). Verification is already built-in during the design phase, as the designed model must be verified before being evaluated. In the evaluation phase, the designed model is tested to understand how the model performs, and validation is done. An additional phase is added to this framework,

which is the concluding phase. It is essential to reflect on the findings and limitations of the new model, which is incorporated in this final phase.

First, the context, objectives, and scope of the problem are defined. Additionally, a literature review and expert consultation will identify the critical variables for measuring the performance of product returns and develop the methodology for constructing the CI. In the second phase, a conceptual model is developed with the findings from the literature on product returns will be applied to the literature found to construct a CI. The specific methods to construct the CI for this research will be chosen, the conceptual model will be developed, and an overview of the required data will be made. The third step is to analyze the company's performance by building a base case using the created CI. This step involves preparing the data, measuring the base case, and concluding the findings. Next, the benchmark will be developed in the design phase. The benchmark will be tested, evaluated, and verified. After successfully verifying the model in the design phase, insights are created into how companies can use the model in the evaluation phase. This will be done by developing scenarios tested with the constructed CI, which will validate the model. In the concluding phase, the main findings are presented together with the limitations and recommendations of this research.

#### 1.6.2. Research outline

In Figure 1.3, the outline of this research is visualized. The blue boxes represent the methods used for each step. This research has eight chapters, structured in the Define, Conceptual, Analyse, Design, Evaluate, and Concluding phases.



Figure 1.3: Research outline, source: (Author)

#### 1.6.3. Expected research output

An overview of the expected research outcome is as follows:

- 1. An overview of the return process in the fashion industry, including the activities that take place and how these differ for each return method.
- 2. Insight into the activities in the return process that create environmental impact and how they can be measured.
- 3. Insight into the activities in the return process that create costs and how they can be measured.
- 4. Construction of a CI for measuring and assessing the environmental and economic performance of returns for a fashion company.
- 5. Insight in how fashion companies can make a trade-off between their environmental and economic performance of product returns using a CI.

- 6. Design of a benchmark that represents target performance for fashion companies.
- 7. Prove of usage of the CI by performing a case study using data from a fashion company.
- 8. Discussion on how the performance of return methods differs and which method is preferred or should be avoided.
- 9. Discussion on improvement areas and risks for companies by evaluating different scenarios.
- 10. Discussion on the academic and practical implications of this study.
- 11. Discussion on the research limitations.
- 12. Recommendations for further research.

#### 1.6.4. Data sources

This research uses both quantitative and qualitative data. Literature was used to create an understanding of the return process, identification of environmental and economic factors, and how these factors can be measured. Relevant articles were found using the following electronic search engines: SCOPUS, Scholar Google, Web of Science, and TU Delft Repository. Expert consultation and interviews supplemented the reviewed literature and verified data using industry knowledge. These interviews were conducted with internal experts from Metyis, a data-driven strategy consulting firm specializing in the premium fashion industry and digital commerce. Metyis understands and has expert knowledge of global fashion retailers, supply chains, and businesses. Additionally, Metvis has experience, provision, and understanding of many business data on fashion retailers. Also, external experts were interviewed. These external experts were questioned based on their expertise; an expert specializing in recycling post-usage clothes was contacted better to understand the recovery of returned products in this research. An expert within finance and banking was interviewed to gain further insights into the costs and substantiate scenario developments. An expert with industry knowledge of GHG accounting for businesses was interviewed to understand better how this works and what methods companies can use. Additionally, an expert working at a large warehouse of fashion cooperation is interviewed, and further insights are obtained on the actual return process and how costs are acquired in practice. Further, literature was used to select an appropriate methodology for this research, clarify the choices made for the method used, and develop the construction of the model. The model is implemented using Python and Excel. The data for the base case originates from a client of Metyis. The considered data was measured for one whole business year.

## 2

## Product returns and their impact

This chapter will study the determinants for assessing product returns' environmental and economic performance using a literature review supplemented with expert consultation. This chapter is structured as follows: First, it is explained why products are returned in section 2.1. Next, the return process is elaborated on in section 2.2, followed by a concise overview of the return methods. In section 2.4, the environmental impact of the return process is analyzed, identifying factors and measurement methods for assessing the environmental impact. Lastly, the cost factors of the return process and the measurement methods for these factors are mapped in section 2.5. In this chapter, sub-research questions 1,2, and 3 will be answered, respectively: *What does the fashion industry's product return flow look like?, How can the environmental impact of the various logistic activities involved in product returns be assessed?*, and *How can the economic impact of the various logistic activities involved in product returns be assessed?* 

#### 2.1. Why fashion products are returned

In both online and offline channels, customer returns are offered. Customer returns refer to products returned after delivery (or purchase in offline environments) to the customer [45]. According to European laws, customers who purchase a product online hold the right to return their purchase within fourteen days. Companies are thus legally obliged to offer this service even though the product does not show any flaws. In offline environments, companies are not obliged by law to directly provide an exchange or refund when a product is returned, except if the product shows defects [71]. However, returns can still be offered to boost sales or when companies feel like they have to, as their competitors are offering it [45].

Customers return products for a variety of reasons. Repairs, warranty of returns, end-of-use, end-of-life, wrongly delivered product, damage, and customer fraud are mentioned as general motives [45]. Returns can be categorized as legitimate, opportunistic, and fraudulent. Legitimate returns refer to returns made by a customer who intends to keep a product when they purchase it but eventually decides to bring it back. Opportunistic returns happen when the customer knows they are returning a product before the purchase, such as when they buy identical products in multiple colors and sizes [123]. Fraudulent returns refer to customers who return products even though it is not by legal rules or regulations, for example, after wearing an item [91].

An explanation for the high return rates in particularly fashion is the uncertainty customers face when they purchase an item online. Customers who shop online experience a higher chance of a mismatch between what they expect from a product and what they receive [175, 127]. Due to incorrect sizing information or undetailed and inaccurate product descriptions, customers receive unwanted products which they send in return. Misfit and sizing issues account for 42% of the reasons for a return within e-commerce [126]. Therefore, customers tend to purchase multiple variants of the same product to increase their likelihood of receiving the right product and return unwanted items [175]. For this reason, customers are more likely to purchase products from fashion companies that provide convenient and free return policies, and fashion companies use lenient return policies to boost their sales nowadays [93, 21, 127, 158]. However, these policies normalize over-ordering behavior and drive opportunistic returns, as customers might 'just order' a product to see it at home [200, 89]. These factors contribute to the higher numbers of returns in the fashion industry. Fashion companies do not prefer adopting a restrictive return policy as it might impact their businesses. Therefore,

changing policies' leniency is not a preferred method of reducing the environmental impact of returns in fashion [194, 175, 93, 101].

#### 2.2. How the process of product returns works

A high-over definition of the reverse process is given by [45], consisting of collection, inspection/selection/ sorting, recovery, and redistribution. [15] defines this process as (p.490-491): "customer return request, return logistics, processing and sortation, inventory control, repair and refurbishment, and final disposition." From this perspective, the returns process is linear and can be seen as a singular phenomenon. However, several return methods are available, so this simplicity no longer holds [14]. Therefore, four phases are distinguished in the product return process: return preparation, collection, processing and handling of returns, and product recovery and disposition, which is visualized in Figure 2.1. Within each stage, consideration is given to how a product can be returned. Each process phase has various logistic activities that must be considered to evaluate the impact created in the return process [145, 51].

In short, the return is first prepared by the customer. The customer must request the return, choose a return method, and ensure the product's packaging is for safe transport. The return method determines how the collection transports the product from the customer to the final recovery point. This final recovery point is the brick-and-mortar store or the local warehouse. Transport networks for forward deliveries are commonly used for return transportation [40, 78]. A vehicle journey needs to be designed for each return method, and consideration must be given to factors including the integration of deliveries and pick-ups, the vehicle fleet, the combination level, and whether the split collection is acceptable. A general vehicle routing model can be used to model the reverse network [18].



Figure 2.1: Return processes and sub-activities, adapted from [42]

According to experts, when a product is collected and has arrived at the brick-and-mortar store, it can be processed and handled in-store to be resold or transported to the warehouse. Warehousing activities are receipt, inspection/processing/sorting, storage, and internal transport [45, 174, 15]. First, it is checked whether the product meets the conditions for a return. If this is completed, product inspection occurs, and the product is registered. Following, the product is moved to the temporary storage [103].

E-commerce has shifted warehouse requirements, as they need to resist higher and more fluctuating volumes of returns [164, 103, 42]. Also, orders are single-picked and handled rather than in batches [103]. The area in a warehouse dedicated to returns differs per (type of) retailer [42]. The number of product returns in the fashion industry is substantial compared to other industries. Therefore, it is common for fashion retailers to have a separate area for return receipts and storage facilities at the warehouses [42, 45]. It is also possible to have a separate warehouse for returns- and on orders when the retailers' warehouse capacity is insufficient [96]. Those separate centers are mostly managed by Third Party Logistics Provider (3PL), who specialize in return handling and improving efficiency in the process. In a case study with different e-commerce companies of [96], it was found that less than 20% of the surveyed companies used separate centers for their return handling 2016.

After the product is inspected, the product recovery strategy is chosen. High-quality items can be resold quickly through redistribution, resale, and reuse. Reprocessing is required for defective products, and procedures include recycling, mending, dumping in a landfill, and burning [45, 107, 5]. The recovery decision of a company is based on several criteria, including the cost, the product market value, added value of recovery, and the type of product [107]. Reselling does not require fixing or replacing activities and often only

requires repackaging the products before storing them in storage [107, 5]. Recovery, repair, and recycling disposition strategies of the products aim to get as much return of the product as possible. Typically for fashion e-commerce, high returns are experienced although these products are not damaged or show defects [124]. Most of these products are resold at (secondary or outlet) markets or sent to landfills or incineration [36, 5]. Currently, the share of recycled clothes worldwide is very low and below 1% [52]. An expert with industry knowledge mentions that this is also true for the recycling rates of clothes in the Netherlands.

#### 2.3. How a customer can return a product

Generally, fashion companies use three return methods: in-store returns, drop-off at collection points, and home pick-up. The availability of a return method depends on the type of retailer where a product is purchased. Purely brick-and-mortar retailers solely offer in-store options for handing in a return. Pure online retailers have no physical store and only offer returns services by dropping off packages at collection points or home pick-up. Omnichannel retailers provide all of the above, as they are active in physical stores and online channels. A short elaboration is given for each return method:

- In-store returns: Customers must return the product to the physical store. Returns will be processed instore and put on the shelves in-store, or a vehicle will pick up the return to transport it to the warehouse.
- Home pick-up returns: The company offers a home pick-up service, and a pick-up moment is scheduled with the customer. The customer must repack the product for transport, which is picked up at home by a 3PL. The 3PL service will transport the package to the final destination.
- Collection point returns: Customers must repack the product and bring it to a collection point. The company offers a set of alternatives for collection points with different 3PL companies (i.e., UPS, PostNL, and DHL in the Netherlands). The collection points can be locker systems, kiosks, office stores, or other stores that offer this additional mail service. The 3PL picks up the package from the collection point and transports it to the final destination. The number of collection points available to customers depends on the density of residents where they want to drop off the package [152, 30].

#### 2.4. How a product return creates environmental impact

This section gives insight into the environmental impact created by product returns and how this is measured. This is done by analyzing the four phases of product returns: return preparation, collection, processing and handling, and product recovery and disposition. Within each stage, consideration is given to how a product can be returned, which factors should be considered for the environmental impact, and how the environmental impact can be measured. A high-over visualization of this is presented in Figure 2.2, where the boxes in red represent the activities that create environmental impact in each phase. First, it is elaborated on how the environmental impact is measured in subsection 2.4.1, followed by a comprehensive description of the activities within each phase and how they can be measured.



Figure 2.2: Flow chart of the process of product returns, considering sources of emissions and return methods, source: (Author)

#### 2.4.1. How the environmental impact is measured

This research intends to assess the environmental impact of product returns by measuring the Greenhouse Gas (GHG) emissions. The literature shows several valuable tools to evaluate environmental impact, including the GHG Protocol and the Life Cycle Assessment (LCA). The focus of this research is to measure the GHG emissions that arise in the different activities involved in the process of product returns for a company. Activities before the return request are excluded from this research scope. Therefore, a measurement method should focus on assessing the impact of specific activities and processes rather than all the stages in a product's life. Additionally, this environmental assessment focuses solely on the GHG emissions and not other environmental factors such as water usage or resource depletion.

The LCA is a well-applicable method when measuring a more comprehensive set of environmental factors and evaluating the effect of a product throughout its entire life cycle [130]. The GHG Protocol can be more useful when specific activities and GHG emissions need to be measured. Also, the GHG Protocol is commonly used as an accounting tool at the company level for activities in its value chain [83].

Taking these aspects into account, the GHG Protocol will be used to measure the GHG emissions of the activities involved in the process of product returns. Examples of the usage of this method to assess GHG emissions can be found in the research of [76] and [162]. The GHG Protocol is a recognized tool by the European Union (EU) for measuring and reporting GHG emissions of different activities. It provides a standardized methodology for measuring scope 1, 2, and 3 emissions. In Scopes 1 and 2, emissions are measured from sources the company owns [58]. Scope 3 emissions produced by downstream and upstream activities in the company's value stream, such as outsourced transport, distribution, retail, storage, and end-of-life treatments of products [84, 83].

The GHG Protocol uses Emisison factor (EF)s to calculate the GHG emissions. These EFs should be retrieved from peer-reviewed, widely accepted databases consistent with GHG Protocol standards, such as 3EID, Defra, Ecoinvent, or IDEA. These are mostly life cycle databases, where the EF are calculated based on the whole life cycle or cradle-to-gate emissions [88]. Other sources that can be used to obtain EF data are from governments or academic studies.

#### 2.4.2. Phase I. Return preparation

In the fashion industry, packaging significantly contributes to GHG emissions [191, 118, 102, 16]. The reason for this is twofold. First, products are packaged individually to ensure that they are transported safely. Second, only a tiny percentage of customers use the original package when returning a product. Nearly 80% of the customers do not use the original box of the product when they return it [32, 191]. This implies that for this fraction of customers, product packaging doubles when they return [191], generating additional waste.

The packaging is needed if the customer selects an online return method (i.e., return through a collection point or home pick-up) or when the product gets transported from the brick-and-mortar store to the warehouse. This additional packaging waste needs to be treated by the company at the warehouse [102, 118, 188]. In the apparel industry, transit packaging is used. This can be mainly classified into paper and plastic packaging materials for e-commerce [195, 54]. Of these packaging types, cardboard boxes are the most widely used option in e-commerce in particular [33, 32, 102]. An expert with industry knowledge of fashion warehouse operations also confirms this. Cardboard packaging ensures the safe transportation of the product and is also relatively low-cost and lightweight. However, cardboard boxes are primarily designed for single-use only, and the emissions created by cardboard packaging are 60% higher than plastic packaging bags [191, 16]. This is mainly because a larger volume of cardboard must be used compared to plastic for one package (i.e., 100-gram cardboard against 8-gram plastic per package [188]) and the re-usability of plastic bags compared to single-use cardboard [16].

To calculate the GHG emissions of packaging, insight into the type of packaging material used and the package size is needed. A material-specific EF can be multiplied by the mass of a parcel to determine the kgCO2eq-emissions of packaging, as seen in Equation 2.1 [84, 176].

Packaging usage emissions = 
$$m * EF_{packaging material}$$
 (2.1)

The additional product packaging waste needs to be considered for the environmental assessment of a company [84] and is relevant as Europe's total packaging waste is growing steeper than GDP [68]. Waste treatments are recycling, incineration, or disposal to landfill [176]. The emissions of packaging waste can be determined by the total mass of packaging material multiplied by the waste-treatment-specific EF, see Equation 2.2 [84, 176].

Packaging waste emissions =  $m * EF_{waste treatement}$  (2.2)

Depending on whether the EF selected is based on the entire life cycle, it is sufficient only to include the waste method for the specific type of packaging to determine the GHG emissions, as double counting of emissions must be avoided [84].

#### Summary

Packaging emissions are created to ensure safe transportation, which is needed when a customer returns by home pick-up or collection point or when the product gets transported from the brick-and-mortar store to the warehouse. Cardboard boxes and plastic polybags are the most common packaging material. Packaging emissions consist of usage and waste emissions. The magnitude of the usage emissions depends on the type of packaging material, the mass of the packaging, and whether or not a customer reuses the packaging for their return [84, 176]. The packaging waste emissions depend on the waste treatment chosen, the type of packaging material, and the mass [84, 176].

#### 2.4.3. Phase II. Collection

There can be two transport activities considered in the transport network for returns. First is the short-haul movement between the customer and the (if present) initial recovery point (i.e., store, collection point). Second, is the product's transportation from the initial recovery point to the final destination. A general overview of the transport activities and recovery points involved in the return process is visualized in Figure 2.3. The transport activity and measurement approach are further elaborated for every return method.



Figure 2.3: Transport structure general overview, adapted from [77]

Environmental assessment models for reverse transport streams are currently designed concerning the condition that the same method is used to deliver and pick up a product. By doing so, it is assumed that the GHG emissions of the delivery are roughly the same as the pick-up transport order [40]. However, this condition does not always hold as customers can choose another return method than their delivery method, increasing the complexity for determination of the transport emissions. The transport emissions should be calculated separately for each return method to accurately assess the emissions [51, 22, 118]. According to [78], returns can, in many cases, be implemented in the forward distribution networks. This means the same logistic network designs can be used for return logistics. The methods for calculating transport emissions for deliveries can be adapted and used to determine the emissions of the different return methods [40].

Further, this research does not consider integrations of deliveries and return requests in a vehicle journey [77]. Although integration of these streams is sometimes employed, combining these pick-up and delivery tasks can be complex because of vehicle loading constraints. This may lead to situations where a vehicle first makes a delivery round, and after all the deliveries are made, it collects the returned products that are returned [77, 45]. Also, it might be that a different warehouse for collection and delivery and another vehicle fleet is used, which means that integration is not likely [78]. The collection and delivery tasks are therefore treated separately in this research. The trip purpose of the vehicles is to pick up the returned items and transport them back to the final destination. A general vehicle routing problem with a single warehouse is assumed to be the underlying transport model, in which several conditions are respected. In this transport model, the return requests are represented by demand nodes *n*, and the arcs represent the actual distance  $d_{ij}$ 

from node *i* to node *j*. The vehicle departs at the warehouse and ends its journey at the same location. The origin and destination are represented by nodes 0 and n+1. The capacity of the vehicle or time limitations limits the vehicle's journey. Additionally, it is assumed that a customer's request must be served in one visit [35, 192, 45].

Two methods for determining transport emissions on a company's vehicle journey are based on the fuel used or the distance driven [84, 85]. The former method calculates a vehicle's fuel consumption and multiplies this with fuel-specific EFs to measure the GHG emissions. The latter method multiplies the total travel distance by a specific type of modality, the transported mass, and an EF [84]. The suitability of an approach depends on several factors, including the required detail, the context of the analysis, and the availability of data to determine the emissions. The transport emissions are measured for each return method and elaborated further below.

#### Return alternative 1. Brick-and-mortar store returns

The customer must take the product back to the brick-and-mortar store. The GHG emissions created depend on the type of modality chosen by the customer and the distance traveled. After the returned products have arrived at the store, returns can either be handled and resold there, or they will be transported to the retailer's warehouse. If the decision is made to move the product to the warehouse, additional transport is needed [42].

To determine the GHG emissions from transport for a product returned in a physical store, the whole journey of a customer is considered. It is assumed that returning is the purpose of the trip; therefore, the transport emissions from the customer to the store and returning home are included in this assessment [199, 118, 51, 24]. Gathering accurate data on the amount of fuel a customer uses can be challenging and not feasible within the limited scope of this research. Therefore, the distance approach is likely more suitable than the fuel-based to determine transport emissions. The total distance is determined by the kilometers traveled from the customer's location to the closest store location and back [84]. The following equation gives the best approximation of the transportation emissions of this return method, which is adapted from [40].

Transport emissions = 
$$\sum d * m * EF_{modality}$$
 (2.3)

Here, *d* represents the distance of the customers' location to the brick-and-mortar store and the distance from the store back home in kilometers. A specific EF *EF* is assigned based on the modality. It can be difficult to obtain accurate data on a customer's specific modality, and it is common to make assumptions to this end [199, 118, 51].

For the optional transport activity from the store to the warehouse, the store is treated as a demand node in the transport model [40]. The transport emissions for this journey are added to the transport emissions of the customer trip.

#### Summary

Transport emissions depend on the customer's trip to the store and back home [199, 118, 51, 24]. If the decision is made to transport the product to the warehouse, additional transport emissions are created, depending on the distance traveled, product mass, and vehicle energy consumption. Considering the scope and availability of data in this research, the best approximation for measuring transport emissions is the distance method.

#### **Return alternative 2. Home pick-up**

The transport network structure of the home pick-up service depends on the specific locations where a customer requests a return. Transportation services must include this pick-up point in their vehicle journey to meet demand. The number of requests that can be served in one vehicle journey depends on the capacity of the vehicle and time constraints. The transport emissions created when a product is picked up at the customer's door depend on the modality used to fulfill this service, the type of vehicle journey (consolidated pick-up and delivery or separated journey), the number of stops, and the distance between the customers' location and the final destination [84, 40]. According to expert consultation, the decision can be made at the distribution center to transfer the product to a brick-and-mortar store or a warehouse.

It is considered that each demand node corresponds to one return request for a home-pick up. So, the number of nodes equals the number of return requests. The requests are represented by the set *P* demand nodes with 1,...,n requests. The distance  $d_{ij}$  represents the distance from pick-up point *i* to pick-up point *j*. Nodes 0 and n + 1 represent the recovery point [46, 35, 192, 40]. The Vehicle Routing Problem with a single

depot as formulated by [46, 35, 192] and the calculation of transport emissions on a pick-up route as proposed by [40] is adapted to the following equation:

Transport emissions = 
$$(\sum_{(i,j)\in P} d_{ij} + d_{0,i} + d_{j,n+1}) * m * EF_{modality}$$
 (2.4)

Here, a constant vehicle fuel consumption is assumed [40]. A customer might return multiple items within one return request, so the number of nodes equals the number of requests. [118] proposes a simplified calculation when there is a lack of data on the subsequent coordinates of the nodes, where the average distance of 50 km per parcel is assumed for a pick-up route, which can be multiplied by the mass of the parcel and the modality-specific EF to determine the transport emissions of this pick-up journey.

Next, Equation 2.3 is used to calculate the emissions between the initial point of recovery and the final recovery point, as this long-haul transportation is performed by another modality [84]. These vehicle characteristics need to be taken into account for this trip. The total emissions per product return using pick-up methods can be measured by a summation of the emissions generated by both trips.

#### Summary

Based on the locations of the pick-up requests, a vehicle journey is constructed from the pickup point to the initial recovery point. The total distance traveled, and the mass should be measured and multiplied by an EF for the modality used on this vehicle journey [40]. For the long-haul trip between the initial point of recovery and the product's final destination, the distance method should be applied, considering the vehicle characteristics performing that trip [84].

#### **Return alternative 3. Collection point returns**

Three transport activities are distinguished to determine the transport emissions when using collection points for returns. These are the customer's transport to the collection point, from the collection point to the initial recovery point, and the warehouse or brick-and-mortar store. In the Netherlands, collection points are relatively well accessible [152, 30]. Therefore, it is assumed that customers only need to travel short distances when they bring their product to a collection point. The most preferred modalities for these short distances are walking, PT, or cycling [28]. The transport emissions for this activity can be determined similarly to the in-store transport emissions using the Equation 2.3. Thus, for this part of the transport assignment, the customer's location, the collection point, the mass of the product, and their choice of modality need to be considered [84].

Second, the product is transported from the collection point to the initial recovery point by a vehicle of a 3PL. The vehicle journey starts at the departure of the distribution center, visiting several collection points within a specified distance and maximum vehicle payload range and returning to the recovery point. Vehicle journey constraints are the vehicle's maximum payload and the operating time [45, 40].

The emissions for this journey can be calculated by adapting Equation 2.4, where a constant fuel consumption per traveled kilometer is assumed [40]. The distance  $d_{ij}$  represents the distance from collection point *i* to *j*. Nodes 0 and k + 1 represent the distribution center. *k* represents the number of collection points. The number of requests served per collection point should be summed up to determine the total number of requests handled per journey.

Transport emissions = 
$$(\sum_{(i,j)\in C} d_{ij} + d_{0,i} + d_{j,k+1}) * m * EF_{modality}$$
 (2.5)

The emissions generated by transport from the initial recovery point to the final destination are calculated using the distance method, considering the specific vehicle characteristics for that trip [84]. The total emissions per product return can be measured by a summation of the emissions generated by the three transport activities.

#### Summary

Transport emissions are measured by distinguishing three transport activities. First, the transport of the customer to the collection point and emissions are created depending on the modality, product mass, and distance traveled. Second, the product is collected at the collection point by a 3PL as part of a vehicle journey. The transport emissions of this journey depending on the distance, modality used, the number of collection points served in that journey, and vehicle characteristics. Third, the created emissions from transporting the initial recovery point (i..e, distribution center) to the final destination should be considered and measured by the distance method [84].

#### 2.4.4. Phase III. Product handling and processing

Return handling at warehouses is critical in the return process and has become essential to warehousing operations [103]. Four sub-activities can be specified: receipt, inspection/processing/sorting, storage, and internal transport [45, 174, 15]. The activities are visualized in Figure 2.4.



Figure 2.4: Product handling and processing process, adapted from [45, 42]

First, the product arrives at the designated location in the warehouse and is checked to determine whether the product meets the conditions for a return. If this is the case, the product inspection takes place, and the product is registered. Following, the product is moved to the storage area in the warehouse dedicated to returns [103]. According to an expert with knowledge of fashion warehouses, the storage of a product takes up approximately 3-5 days before it will be reallocated (redistributed or sent for processing recovery). Different factors must be considered when looking at the environmental performance of the operation processes in a warehouse. First of all, the energy consumption of the warehouse must be considered. An environmental impact study of warehousing of [154] links the size of the warehouse to energy consumption and the geographical location of the warehouse. The main contributors to warehouse energy consumption are the lighting, heating, and cooling of the building [118, 154]. The literature proposes detailed calculation methods for determining GHG emissions per fuel and usage type. However, this level of detail is unknown as warehouse managers mainly have information on the total energy consumed at the warehouse by fuel type [75]. Therefore, this research uses a more aggregated measurement approach to determine the GHG emissions generated in the warehouse, which is derived from [84, 83]. Here, the total electricity consumed at the warehouse is multiplied by a specific EF. The fuel consumed in the warehouse is also multiplied by an EF. The electricity and fuel consumption are measured over a particular period, for example, a year or quarter [75]. These values must be in the same unit of measurement. The total warehouse emissions are divided by the number of products handled during the considered period, represented by n in Equation 2.6.

Warehouse emissions = 
$$\frac{(\sum \text{ fuel consumed } * EF_{fuel}) + (\sum \text{ electricity consumed } * EF_{electricity})}{n}$$
(2.6)

Additionally, material handling equipment is needed to process and move the products between areas in the warehouse. The equipment in a warehouse is also a contributor that should be considered in the environmental assessment of product handling and processing emissions [154]. There is fixed equipment like conveyor belts and mobile equipment like trucks and forklifts [154]. Based on this nature, different factors should be considered for the environmental assessment. Mobile equipment's generated emissions and consumed energy can be determined based on the average distance traveled and the equipment characteristics [90, 75]. For fixed-handling equipment like conveyor belts, energy consumption is primarily associated with the size of the warehouse, the hours of operation, and the equipment characteristics [154]. The input data for these emissions are likely not known on a product level, according to a warehouse expert, so therefore the equipment emissions should be determined for a whole year, considering the fuel and electricity consumed. This number should be divided by the number of products served with this equipment to derive the GHG emissions on the product level. The GHG emissions can be determined by Equation 2.7, where *n* represents the number of products handled over the considered year, adapted from [84].
Equipment emissions = 
$$\frac{(\sum \text{ fuel consumed } * EF_{fuel}) + (\sum \text{ electricity consumed } * EF_{electricity})}{n}$$
(2.7)

When a product is returned through an in-store return alternative or redistributed to the store, the product is handled in-store. The environmental impact should be considered for receiving, processing, handling, and restocking the product return at the store. The energy consumption of the brick-and-mortar store should be analyzed and converted to GHG emissions by an appropriate EF. The energy consumption can be generally approximated based on the invoices and further specified by the store size, location, and the energy rating [118]. Based on the average time a product spends at the store, and the number of products handled in the store over time, the in-store emissions allocated to a product return can be determined [84].

Summary

The total warehouse emissions can be determined by the fuel and electricity consumed, which are converted to GHG emissions by a specific EF. In addition, internal transport and the energy used by warehouse equipment for handling facilities must be considered. There are two types of equipment, which are fixed and mobile [154]. The energy consumption of the former depends on the kind of equipment, size of the warehouse, and operation time. For the latter, energy consumption can be measured as a function of the distance traveled, equipment type, and characteristics. The total emissions must be divided by the total number of products served. Further, products can also be handled, received, and processed at a brick-and-mortar store. Here, the energy consumption of the store needs to be considered.

#### 2.4.5. Phase IV. Recovery and disposition of products

The final phase in product returns is the recovery and disposition of products. High-quality items can be resold quickly through resale and reuse. Reprocessing is needed for imperfect products, and procedures include recycling, repair, remanufacturing, landfill disposal, and incineration [45, 107, 5, 77]. Recovery can be categorized as direct and process recovery, where the latter means reprocessing activities are needed [45]. The type of recovery and disposition strategy chosen affects the created GHG emissions in this stage [15, 84, 68].

The calculation of GHG emissions for this phase of the return process is defined for the chosen recovery method. In Equation 2.8, a formulation is given where the waste treatment EF refers to the chosen recovery strategy for the product [84].

Recovery emissions = 
$$m * EF_{waste treatment}$$
 (2.8)

#### Summary

The magnitude of the environmental impact of a product recovery depends on the chosen strategy. Products can be resold, repaired, recycled, remanufactured, or sent to incineration or landfills [45, 107, 5, 77]. The environmental impact can be measured by the mass of a product and the specific EF for the chosen waste strategy [84].

#### 2.5. How a product return creates costs

A breakdown of the significant cost factors involved in a company's product returns process is looked into, as well as how these cost factors can be measured. According to an industry expert, this should be done according to the theory of activity-based costing, as fashion companies prefer this method in practice. This means that each activity in the return process needs to be considered for the cost estimation. Also, the different return methods affect the costs [117]. This research only includes the costs made in the process of a product return for a company by performing the return activities. Costing of return-related activities includes transportation, labor, handling equipment, disposition, and storage of the products and packaging [89, 114]. Two independent experts confirm these cost activities for returns.

#### Transport costs

This metric captures the costs for transporting the product from the origin to its final destination (i.e., warehouse or brick-and-mortar store). Different factors must be considered for each return method, such as the transport modality, distance traveled, weight, and product size [89]. Also, the transport is usually outsourced to a 3PL but can be self-executed, affecting the cost structure [42]. Brick-and-mortar returns only generate transport costs for a company when the decision is made not to resell the product in-store and transport it to the warehouse. This cost metric can be measured using the distance-driven method or the fuel consumed [45]. Factors such as leased vehicles, transport planning, and service costs should also be considered. A calculation can be made based on the distance driven, the vehicle load, the mass of the product, and fixedand variable modality costs, as presented in Equation 2.9 [187]. Here the  $c_{charge}$  includes modality-specific factors expressed per ton-km distance driven.

$$C_{\text{transport}} = \sum d * c_{charge} * m \tag{2.9}$$

[89] proposes another measurement approach for the transport costs per item at collection points, independent from the distance driven, where a transport fee based on the return method is proposed using Equation 2.10. Here, *r* is the return rate,  $c_{tra}(q_k)$  the transport fee for a parcel with a certain weight,  $c_{cus}$  is the optional fee the company charges the customer for a return, and *k* the number of items returned in one parcel [89].

$$C_{\text{transport}} = \sum_{k=1}^{k-1} \frac{r \delta_n(q_k)}{k} * (c_{\text{tra}}(q_k) - c_{\text{cus}})$$
(2.10)

#### Warehouse space costs

A major source of costs in product returns is related to warehousing activities [42]. The activities that should be considered are product receipt, handling, inspection, holding, and storage, which all require space at the warehouse. These costs can be calculated by estimating the proportion of the used space dedicated to returns from the total warehouse area. Based on this estimated proportion, the costs can be determined based on the total warehouse space costs [42]. This can be determined based on the area for return handling and the warehouse rental cost [42]. The calculated amount should be divided by the total number of returns to derive the warehouse space cost per return item.

#### Warehouse equipment costs

These factors refer to costs for the equipment needed to process product returns (e.g., forklifts, sorting systems, conveyors, and packaging machines). In this metric, equipment depreciation is captured when equipment is not leased. Equipment costs per return can be measured by dividing the total equipment costs by the number of returns handled over a time period.

#### Warehouse labor costs

Labor costs are expenses associated with the human resources needed in product returns handling, which is quite intensive for returns. Activities include receiving, inspecting, and restocking products [89]. The product return labor cost on an annual basis is the sum of these fees [89]. Additionally, this metric should include any overhead expenses and investments like training or other employee benefits in this metric [42]. The calculation of [89] is adapted to Equation 2.11, which expresses the total labor costs per returned item:

$$C_{\text{labor}} = c_{\text{res}} + c_{\text{rec}} + c_{\text{overh}} \tag{2.11}$$

Here,  $c_{res}$  is the costs for inspection and restocking per product,  $c_{rec}$  is the cost of receiving per item, and  $c_{overh}$  is the overhead expenses per product [89, 42].

#### Product recovery and disposal costs

This metric refers to the expenses of products that cannot be resold anymore due to a return and sent to a landfill. The recovery costs per return are calculated by the mass of the returned product, the recovery charge per kg, and optional transport charges. The charge depends on the type of recovery treatment, the mass of the product, and the country of recovery [69, 53].

$$C_{\text{recovery}} = m * (c_{\text{charge}} + c_{\text{tra}})$$
(2.12)

Where *m* is the mass of the product,  $c_{\text{charge}}$  is the charge for landfill or disposition, and  $c_{\text{tra}}$  is the optional transport cost to the recovery facility.

#### Packaging recovery and disposal costs

Each product arrives at the warehouse with packaging material to ensure safe transportation. The company must handle this waste by choosing a waste treatment. The waste treatment chosen is either recycling, incineration, or landfill. Governmental charges are associated with landfill and incineration, which come at the cost of a company [94]. The created costs depend on the type of waste treatment selected, the packaging mass, the packaging material, and additional recovery or disposal charges [39].

#### Brick-and-mortar store space costs

Operating costs of the physical store should be considered for in-store returns. According to an expert, the most significant cost factor is the leasing costs of the store. Usually, these leasing costs can be derived from the annual financial statement. The costs per unit product can be estimated by obtaining the total number of products handled at the store over a certain period.

#### Brick-and-mortar store labor costs

Labor costs are considered in this metric, including receiving, inspecting, and restocking the returned products in-store. Additionally, the overhead employee expenses should be considered in this cost metric. The same formulation as Equation 2.11 can be used to derive a number for this metric, adapted from [89]. Another approach for measuring this cost metric can be determined by the hourly labor cost multiplied by the duration of a task [189], where the employee overhead expenses also should be considered.

#### Summary

Costs involved in the product return process for a company can be broken down into several factors using the theory of activity costing. The following cost factors need to be included in the economic evaluation: transportation costs, warehouse space utilization costs, material handling equipment costs, labor costs, product recovery and disposal costs, brick-and-mortar store space costs, and brick-and-mortar store labor costs.

#### 2.6. Main findings chapter 2

In this chapter, SQ1, SQ2, and SQ3 are answered. First, the process of product returns in the fashion industry is investigated, which consists of a return request, collection and transportation, handling and processing at the warehouse, and product recovery and disposal. An answer is formed to SQ1: *What does the fashion industry's product return flow look like?* 

This overarching SQ is addressed by SQ1a) *Which returns methods are available in the fashion industry?* and SQ1b) *What are the logistic activities involved in the product return process, and how do they differ depending on the chosen return method?*. The return process starts when the customer requests a return and selects a method to process this request, which can be: by brick-and-mortar store return, drop-off at a collection point, or by home pick-up. The availability of the return methods for a customer depends on whether the retailer offers this service. The customer must re-pack the product for safe transport if a home pick-up or collection point is chosen as the return method. Next, the product is collected and transported from the customer to the final recovery destination, either the warehouse or the brick-and-mortar store. How this transport is arranged depends on the return method chosen. Products with the brick-and-mortar store as the final destination are restocked and resold. If the product is sent to the warehouse, it will be received, handled, inspected, and temporarily stored until a decision is made on how to recover it. The recovery choice determines if the product gets resold, recycled, repaired, remanufactured, or disposed to landfills or incineration [45]. After the recovery activity is performed, the process of returns is finished. This process is visualized in Figure 2.1.

This chapter has also mapped the factors that should be considered for an environmental assessment of product returns and how these can be measured.

SQ2. How can the environmental impact of the various logistic activities involved in product returns be assessed?

The environmental impact is measured using EF specified for each activity to measure the GHG emissions created per returned item. The following elements were essential for assessing the environmental impact of product returns: packaging material, packaging waste treatment, transportation, warehousing, storage,

equipment usage, product recovery, and brick-and-mortar activities. To assess the environmental impact of packaging and transportation, it is necessary to consider the chosen return method [51]. Figure 2.2 highlights in red the activities where environmental impact is created and how the return method affects this assessment. These insights are obtained by answering SQ2a) *What are the key factors that should be considered when assessing the environmental impact of these logistic activities*? and SQ2b) *How can the environmental impact of these factors be measured*?

Next, this chapter has provided a breakdown of the significant cost factors of product returns that should be included in the economic assessment for a fashion company. With this information, the third sub-research question is answered.

#### SQ3: How can the economic impact of the various logistic activities involved in product returns be assessed?

This SQ is addressed by examining SQ3a) *What economic factors should be considered for assessing the economic impact of these logistic activities*? and SQ3b) *How can the cost of these factors be measured*?. Through the theory of activity-based costing, supplemented with literature and expert consultation, the following costs are identified: transport, warehouse space utilization, equipment, labor, product- and packaging recovery and disposal, storage, in-store space utilization, and in-store product processing. With exceptions made for the transport and recovery costs, measured per kg return item, these costs are measured annually and expressed in euros per returned item.

# 3

### Methodology

This chapter presents the methodology used in this research. First, an introduction will be given to performance measurement, which is the theoretical foundation for the method. The performance will be measured using indicators aggregated into a Composite Indicator (CI), elaborated on in section 3.2. Once understanding is created of the concept of a CI, the steps to construct a CI are explained in section 3.3 followed by construction guidelines in section 3.4. The measured performance will be compared relatively to a benchmark, as explained in section 3.5. In this chapter, SQ4a) *How can companies use a CI to assess their performance on multiple dimensions*? and SQ4b) *Which steps need to be performed to construct a CI for an assessment on multiple dimensions*? will be answered.

#### 3.1. Why performance should be measured

This study aims to measure, compare, and reflect on the company's performance of product returns on both environmental and economic dimensions. The chosen way to fulfill this objective is by company performance measurement. A definition of *performance measurement* is "the process of quantifying the efficiency and effectiveness of an action" [149, 136]. The choice of performance measurement for achieving the objectives of this research is threefold. First, performance measurement assists companies with the identification of potential areas for improvement [116, 92]. Fashion companies want to reduce the Greenhouse Gas (GHG) impact caused by product returns. Still, it is essential first to determine where the problem lies to specify any actions to correct it. Second, company performance measurement assists a company in tracking its progress in achieving particular goals [203, 125]. In the context of this research, fashion companies aim to lower their GHG footprint, which is a continuous process, and efforts and progress made by the companies should be measured and monitored relative to the goal of meeting their targets. Third, performance measurement is a suitable tool to communicate, motivate, and create awareness with involved actors (i.e., employees, customers, stakeholders) on specific targets or topics [113, 125]. Fashion customers increasingly demand more transparency, information, and understanding of the environmental impact of their choices to make more sustainable decisions. Fashion companies can use performance measurement to create knowledge and information to encourage customers to make better and more conscious decisions. These transparent insights can also be communicated with, for example, governments, as governments are likely to require environmental performance reporting of companies [62].

The added value of company performance measurement is greatest when it reflects reality. Previously, company measurement frameworks mainly assessed performance from a financial perspective as this was deemed the primary indicator of success [203, 2]. As the importance of sustainability and environmental issues has grown, companies need to measure, consider, report, and manage their environmental impact as well [149, 179, 136]. To accomplish environmental and performance improvements, knowledge of the factors that contribute to this environmental impact should be created for companies [205]. A shift towards more integrated measurement frameworks that include, besides economic performance also, environmental performances of companies is expected and introduced by [203].

#### 3.2. How performance is measured by indicators

The performance of a process, system, or activity can be measured using indicators [206]. An indicator is a measure that assesses, analyses, and monitors quantitative or qualitative data on specified topics and breaks it down into small pieces of information [153], which provides information for decision makers [134, 206, 169]. A combination of indicators with multiple dimensions (i.e., environmental and economic) should be considered in performance measurement to provide an overall picture of the situation. This cannot be done with individual indicators because they only cover a single aspect or dimension of the measured topic. When comparing a more extensive set of individual indicators, complexity issues arise that complicate interpretation and comparison [138]. Therefore, a broader set of indicators should be considered that covers the considered dimensions and presents this in a well-structured way. This can be done by a CI.

A CI can give a more comprehensive view of the performance while considering multiple dimensions. Developing a CI is an essential aspect of performance measurement [203, 208, 8]. CI aggregate multiple indicators into a single measure, thereby presenting large amounts of complex information in a simplified way [206, 196]. They can mathematically combine individual indicators with different units of measurement, and no standard way to weigh [179, 134]. The created CI can be used to quantify, measure, evaluate, and compare the performance of a set of indicators on multiple dimensions (see [138, 207, 203, 206]). CIs can be used to track progress, identify shortcomings, provide information, and raise awareness on the topic being measured [133].

The use of CIs has gained popularity and is commonly used in a variety of sectors as a way to provide a comprehensive picture of the performance. However, CIs are also subject to criticism because of some limitations, including complexity, subjectivity, interpretation issues, and missing data. Some steps are exposed to subjectivity during the construction of the CI because decisions and assumptions need to be made [133, 206]. This may lead to potential biases in the results. This may be amplified by misinterpretation issues that lead to misleading information [134]. This can be accounted for by systematically evaluating, reflecting, and reporting the choices that are made during the construction of the CI, thereby increasing transparency and improving the reliability of the CI [134, 133]. In conclusion, there are pros and cons to using the CIs, but if appropriately used, CIs are valuable tools for performance measurement. It is essential to pay awareness to the previously mentioned risks and limitations of the CI by systematically reflecting on the choices to improve transparency and reliability.

#### 3.3. How a Composite Indicator can be constructed

This section discusses the construction of a CI. This construction will be done according to the guidelines of [134]. Typically, five subsequent phases in constructing the CI are visualized in figure 3.1. First, the objectives and scope of the CI are defined. Next, the indicators are selected, normalized, weighted, aggregated, and evaluated [203, 10, 87]. For each phase, a short checklist is given on what should be finished before starting the next phase.



Figure 3.1: Construction of a CI, adapted from [203, 10]

#### 3.3.1. Phase 1. Identification and selection of indicators

The first step in constructing a CI is identifying and selecting all relevant indicators to measure the system's performance or action. To ensure that all relevant indicators are included [134] recommends designing a theoretical framework. This framework should be based on the objectives and scope of the CI, so the indicators represent the various sub-activities that should be measured [134]. The theoretical framework can be used as a final check for the set of indicators, selecting the underlying criteria, and improving understanding of this problem on multiple dimensions [87, 134]. Most of the indicators can be found through a literature review. Further, expert consultation, interviews, surveys, the Delphi technique, and Content analysis can be used to extract additional information [203].

By the end of this phase, a clear overview of the indicators included, their unit of measurement, and the dimension on which this variable will be measured should be provided, as well as what data should be collected [134]. Reflecting on the choices made for the final set of indicators is necessary to improve transparency.

#### 3.3.2. Phase 2. Normalization of indicators

The different indicators have different units of measurement. These indicators must be normalized to aggregate in a CI. Therefore, the indicators will be expressed in homogeneous units of the identical range [10, 87, 133, 206]. This research considers the most commonly used normalization techniques: re-scaling (min-max), standardization (z-value), ranking, and distance to a reference point [134, 133, 10, 150]. The choice of a technique depends on the requirements of the CI and the appropriateness of a technique based on its advantages and limitations. Here a short description is given on these techniques.

- Rescaling: the normalized value is determined by converting the ranges from 0 to 1, irrespective of the original range of values. It is a simple and commonly used technique but shows some weaknesses in its sensitivity to outliers and extreme values and experiences difficulties scaling negative or zero values [134, 206, 180].
- Standardization (z-value): this normalization method standardizes the mean and standard deviation of the indicators to 0 and 1, respectively [203, 206]. It is a commonly used method to make the data dimensionless [150]. Still, it has its disadvantages in that standardization is sensitive to extreme values in the final CI and cannot be used in combination with multiplication aggregation [134, 180].
- Ranking: the values are scaled according to the assigned ranks from 1 to N, in which N represents the highest rank. This method is relatively simple to use. However, scores cannot be compared on value level because the ranking scale loses absolute-level information. Therefore, it is impossible to measure the value difference between scores [180].
- Distance to a reference point: the scales of indicators are converted to a reference point to calculate the normalized value. This is done by dividing the value by the reference value [180], which is an external benchmark [206]. Although the straightforwardness of this method, this method uses the outlier values of the data set for normalization. This means that when the outliers in this data set are very large, this may affect the reliability after using this method [134, 180].

The outcome of this phase should be selecting a normalization approach and adopting it on the indicators derived from the theoretical framework. Justification of the choices made for the normalization method and the procedure should be given.

#### 3.3.3. Phase 3. Weighting of indicators

Each indicator is assigned a weight expressing relative importance [87, 134]. Weighting techniques can be objective or subjective. Objective weighting refers to constructing weights based on statistical or mathematical methods. Subjective weighting involves any expert judgment. The method of Equal Weights (EW) is also available. This weighting technique assigns the same weight to every indicator, indicating that they all have equal importance. EW can also be applied when there is not (sufficient) information available for reliable assignments of weights [87, 153]. The most used weighting techniques are listed below. See [134] for a more detailed explanation.

• Conjoint Analysis (CA): the weight is derived from how much value is attached to a certain indicator by a respondent and how much they would be 'willing to pay' for an option. The operation of this method and computation is relatively complex compared to other weight methods, as many respondents should be sampled, and the utility function should be specified [134].

- Principal Component Analysis (PCA): the (co)variation is analyzed to create weights and is commonly used when correlation exists. A relatively large data set, of which the definition may differ, but a common rule is to have more than 10 data points per variable, is needed. Therefore, this method is less applicable for small data sets as much data is needed. Also, PCA is sensitive to extreme values and changes in data, which are disadvantages of this method [134].
- Data Envelopment Analysis (DEA): weights are assigned based on the performance relative to an estimated benchmark found by linear programming. DEA can lead to difficulties in interpreting the weights of the individual indicators [134].
- Benefit of the Doubt (BOD): the CI is defined as the performance of a system relative to its benchmark performance. This method is commonly used in policy making, and weights express the priorities of policies. A downside of this technique is that categories are difficult to compare, and extreme values weigh heavier. Also, the effect of another normalization method may lead to considerable differences in the final values [134].
- Analytical Hierarchy Process (AHP): weights are constructed by pairwise comparisons where experts make trade-offs [160]. Despite being time-consuming, this subjective technique is often used for assigning weights, particularly for sustainability assessments [87], because it allows decision-makers to express their relative importance but has a built-in check for ensuring statistical consistency [203, 134].

The most suitable weighting technique for a CI should be determined based on its specific objective and requirements. As mentioned as one of the risks of developing a CI, assigning weights is subjective, which may affect the outcome and reliability of the CI and can therefore be a cause of disagreement [87, 206]. For this reason, additional attention should be paid to the choices made, and transparency should be provided in this process.

#### 3.3.4. Phase 4. Aggregation of indicators

The weighted sub-indicators are aggregated and combined into the CI in this phase, which several techniques can do. The selection of aggregation method and related flaws can also lead to different results for the CI. Therefore, this decision should be well-reasoned to prevent criticism [87].

- Linear aggregation (LIN): the aggregated value of the CI is calculated by a summation of the subindicators and commonly used with complete compensability [133, 206]. Compensability refers to the allowance of trade-offs between the variables in a CI [128].
- Multiplicative aggregation (or Geometric mean (GME)): multiplication of the sub-indices results in the aggregated value for the CI. The indicators are partially substituted with each other and are preferred when full compensability is not desired [134].
- Non-compensatory multi-criteria analysis (NCA): NCA methods aggregate the sub-indicators based on designed criteria. This method is mostly used when it is not preferred to have any substitution between the indicators. NCA aggregates the indicators by achieving a 'compromise' between the objectives [134].

The aggregation method's selection should be reflected to complete this construction phase. Further, the potential correlation between indicators should be viewed, and whether the indicators may be substituted by each other [134].

#### 3.3.5. Phase 5. Post-analysis evaluation of the CI

The final phase assesses the robustness of the CI. Several (subjective) judgments must be made while making the CI, and thus it is essential to evaluate upon choices, inputs, and structures of the CI [134]. [203] emphasizes that CIs should have more comprehensive and detailed evaluations to improve its robustness. The evaluation is done by executing an Uncertainty Analysis (UA) and examining the sensitivity to data or choices for methodologies with a Sensitivity Analysis (SA) [133, 208]. Often the evaluation of CIs is limited by solely performing an UA. By using both UA and SA for the evaluation of the CI, transparency on the construction process and underlying choices will be increased more than by just one analysis [134, 133]. The structure of the CI may be adopted and improved using the findings made during the evaluation [133]. For a complete evaluation of the CI, reliable conclusions should be drawn, together with identifying limitations and uncertainties [134].

The following elements are recognized as causes of uncertainty in CI building [134, 133]:

- 1. Indicator selection
- 2. Input data
- 3. Editing schemes for missing data
- 4. Normalization method
- 5. Weighting method
- 6. Aggregation method
- 7. Assigned values of the weights

The UA reflects on the potential sources that create uncertainty of the CI. It is investigated using the UA how these components impact the composition and outcome of the CI. This can be examined by varying the construction methods or using other data as input. CIs can be seen as models, and therefore statistical techniques can be selected to perform the analysis [133]. With the SA, the sensitivity of the CI is analyzed for changes in the sources of uncertainty [134]. While changing the input sources, an understanding of the impact of the indicators on one another should be gained. Because construction methods are also sources of uncertainty, the SA can shed more light on their significance [134, 132].

A measure should be constructed to analyze how different input values affect the final output of the CI. An answer should be formed to the following question: How- and to what extent is the final value of the CI affected by a source of uncertainty for building the CI? A statistical test can address this and provide that the CI is robust and does not extremely depend on the chosen methods [180, 79]. Variance-based approaches are appropriate analysis tools for conducting a SA [132, 203, 147]. Quantitative tests that can be executed to perform a SA are a correlation analysis, regression analysis, and partitioning of the output data [147]. Considering the objective of this SA, which is to determine how- and to what extent the final value is affected by uncertainty sources, a correlation analysis is suggested [97].

Commonly used correlation tests are the Pearson or Spearman correlation coefficient. The choice for one of these measures depends on whether the data is normally distributed and the relationship between the variables is linear [147]. Spearman's coefficient can be used when the variables are presented on an ordinal scale as it utilizes the ranks of each variable relative to another [147, 99].

#### 3.4. How a construction method can be selected

The choice for a technique relates to the degree above subjectivity constructing a CI. This is why the decisions made in every phase need to be well-grounded, transparent, and answerable. Guidelines are proposed by [120] that help to decide which technique is most suitable and appropriate for each phase. These guidelines are visualized in Figure 3.2, which can be used as a structure to select techniques. The analyst makes the choices before constructing the CI.

The flowchart (Figure 3.2) identifies four different levels at which a decision must be made (i.e., blue squares). These correspond to the following factors: indicator type, aggregation type, comparison type, and weighting type. The choices made at the higher levels determine the availabilities of techniques at the lower levels, which are visualized in the yellow boxes. Different paths can be followed based on the decisions made at the tiers.

- **Type of indicators:** The choice corresponding to the type of indicator is to decide whether the components of a CI are substitutable. Substitutable (or compensability) refers to elements that a surplus of another part can offset [128]. When compensation is not allowed, indicators are of the latter (i.e., non-substitutable) category [120]. The nature of the indicator type affects the aggregation approach in the level below.
- **Type of aggregation:** Aggregation methods are simple or complex; a choice should be made depending on the target group and purpose of this CI. Simple aggregation methods are usually applied for a more general audience and vice versa [120].



Figure 3.2: Flow chart to select the best methods for constructing Composite Indicators, from [120]

- **Type of comparisons:** Next, the objective of the CI determines if the dimensions are compared relative to one another or absolute.
- **Type of weights:** The method for assigning weights must be decided at the bottom of the flow chart, depending on the use purpose and objectives of the CI. Weighting can be objective or subjective, or equal weights can be assigned to the indicators.

#### 3.5. Why benchmarking should be performed

The value of the CI is compared relative to the benchmark CI. Benchmarking is used for identifying performance gaps and is, therefore, a commonly used method in addition to performance measurement [204]. The added value of sustainability benchmarking is most significant when the benchmark reflects the best practices [49], commonly a target performance. Therefore, competitive landscaping benchmarking, where the performance of different competitors is measured and compared, is less suitable for this research. The use of target performances for the benchmarks is more appropriate in this research context. Benchmarking also allows for easy comparison between the actual- and desired performance [49].

Target performance can be done by comparing a company's actual and target performance. This can be done in two ways; the relative performance of a company in which the actual performance is compared with a benchmark, and the magnitude of the performance gap. The performance gap between the actual company performance and the target performance should be minimized for best performance [49].

First, a company's relative performance RP is calculated (see Equation 3.1). Here,  $CI_a$  is a company's

actual performance, and  $CI_b$  is the target performance [49].

$$RP = \frac{CI_a}{CI_b} * 100\% \tag{3.1}$$

The actual and benchmark performance differences can identify the performance gap M. If the best practice is the maximization of the indicator values, Equation 3.2 should be adopted. If the best performance is achieved when the importance of the indicators is minimized, the performance gap should be calculated with Equation 3.3 [49].

$$M = CI_a - CI_b \tag{3.2}$$

$$M = CI_b - CI_a \tag{3.3}$$

#### 3.6. Main findings chapter 3

This chapter has created insight into how company performance can be measured when environmental and economic dimensions are included by outlining the construction steps of a CI, proposing guidelines on how techniques can be chosen, and how performance can be benchmarked.

Thereby, knowledge is created to answer the sub-research question 4a) *How can companies use a CI to assess their performance on multiple dimensions*?

Companies can use a CI to measure, analyze, and evaluate product returns on their performance while considering both environmental and economic impact. CI are tools to measure performance, which supports companies in measuring and tracking progress and identifying potential areas for improvement in processes [116, 125, 92]. CIs allow for easy comparisons, summarize large sets of information of individual indicators, and facilitate communication on a topic or problem [138, 207, 203, 206]. In the context of this research, CIs can provide an understanding to companies of their current performance, how the performance of return methods differ, how implementing actions affects their performance, monitor their progress, and communicate these findings with customers to increase transparency or with governments for reporting of their performances.

The added value of this CI will increase if the output from the CI can be compared relative to a benchmark, so companies know how far they are in reaching certain goals [204]. The results can be compared with a performance gap to compare the actual performance with this benchmark performance. Companies should minimize this gap to achieve the highest performance [49], which can be measured and tracked using the CI.

Next, the sub-question 4b. Which steps must be performed to construct a CI for an assessment on multiple dimensions? can be answered.

The structure of a CI typically involves five steps: selection/identification, normalization, weighting, aggregation of the variables, and evaluation [133]. During these phases, choices are made regarding the techniques and methods used, which can be a source of disagreement because of any subjectivity that may arise. Each step increases the probability of subjectivity and bias. To lower this engaged risk of subjectivity, all choices made for constructing a CI should be transparent and well-substantiated. Additionally, special care must be given to inspect the robustness and reliability of the CI, which can be done by performing an UA and SA [203].

# 4

### Conceptual design of the Composite Indicator

In this chapter, the conceptual model for the Composite Indicator (CI) is developed and provides a clear roadmap for its actual construction, including the methods to use and the data to collect. This chapter will form an answer to sub-question 4c) *What methods should be selected to construct the CI for product returns assessing environmental and economic performance?*. First, the objective and scope of the CI are defined. The methods used for the CI are selected, motivated, and discussed in section 4.2. The selected methods are explained in more detail in section 4.3. Lastly, guidelines and requirements for the data collection are presented in section 4.4.

#### 4.1. Objective and scope of the CI

The intent of the CI is to provide fashion companies with an understanding of product returns, focusing on assessing its impact on their environmental and economic performance and considering the return methods. With the knowledge gained from this assessment, companies can compare and analyze different variants of return methods on their performance. Moreover, insights are created into potential areas of environmental improvement while ensuring that their efforts are also economically feasible. This can support companies in making more environmentally sustainable choices to improve the impact of (sub-activities) on their product returns. In addition, the CI should be transparent and improve transparency on the impact of the returns and return methods as well, to be used sector-wide and communicate the insights with customers or other involved actors like governments or shareholders. Companies can encourage and inform their customers about the environmental consequences of their return behavior. This can eventually help make more environmentally conscious decisions on the customer side [194, 22].

Although the primary focus of this CI is to improve the environmental impact of returns in the fashion industry, the economic dimension is also considered in this CI. This decision is made because the fashion industry is primarily profit-driven. Therefore it is worth considering the costs of return methods assessing a company's economic viability. If a return alternative would perform best based on Greenhouse Gas (GHG) emissions but is highest in costs for a company, a trade-off is crucial to remain economically feasible. The objective of the economic dimension is to provide insight into the economic performance of the considered indicators. With this, companies are assisted in making a well-balanced trade-off between the performances of the dimensions and each other. Including these environmental and economic dimensions for the performance measurement increases reality [203, 205].

A specific CI for product returns should be developed to assess environmental and economic performance to reach these objectives. All activities that impact the returns process are relevant and should be included in the CI, which needs to be constructed in a transparent way. For interpretation purposes, it is crucial that the CI is straightforward to understand and use and that companies can easily compare the different alternatives and make trade-offs.

#### 4.2. Selection of sub-methods

This section elaborates on the choices for the methods to perform the steps for constructing a CI. A structured approach is used as outlined in section 3.4. The selection is based on the objective, scope, literature reviews, and expert consultation.

First, it is determined whether the sub-indicators of the CI can compensate for each other. According to [128], *compensability* can be defined as the existence of trade-offs between variables in the CI. This means that a high value of some variables may compensate for the lower values of other variables. Also, compensability refers to the level of correlation and dependency between the variables. Compensability can either be complete or partial. Reviewing the objectives of the CI and the theoretical framework, a partial compensatory approach is found to be most appropriate rather than complete compensability. The partial compensatory approach allows for trade-offs between indicators made by multiple decision-makers. Also, there is, to some extent, compensation for lower and higher scores of indicators instead of complete compensation [128, 129]. An example is given to illustrate the suitability of this choice for the context of this research: in a scenario that shows low costs but a considerable amount of emissions, the final CI value is likely to be not very high as the performance should depend on a combination of both. The low cost is preferred for a company and may compensate somewhat for the large emissions, but complete compensability is certainly not desired [203], as companies need to improve their environmental impact. In addition to this, the theoretical framework of Table 4.1 shows no complete independency between the variables [120]. Considering these conditions, a partial compensatory approach is most appropriate for this CI.

The CI should be functional for fashion companies, and the results should be easily interpreted and communicated with stakeholders and companies. Therefore, a more straightforward (e.g., mathematical function) aggregation method is chosen instead of a more complex method as the latter is less suitable for a general audience [120]. Straightforward methods are Linear aggregation (LIN) and Geometric mean (GME). LIN are less sensitive to outliers in the data and are commonly used for aggregation. LIN suggest complete compensability and mutual independency [133]. Although LIN can still be used, the aggregation of individual weights by a multiplicative function (GME) is better suitable for the construction of this CI [87]. The multiplicative aggregation approach makes the differences in the performances of the considered dimensions (in this research, environmental and economic) explicit [180]. As an outcome of this, this method assists in identifying the indicators that are performing less and promotes improved company performance here. Also, it controls for the effect of specific indicators on the final value of the CI [19].

Further, consideration is given to how the weights should be assigned. The variables in this CI are relatively compared amongst each other rather than absolute. This is because this CI aims for a fashion company to make a trade-off between environmental and economic performance. The company should compare the indicators of the environmental dimension relative to the economic dimension. For this reason, a relative comparison approach is chosen.

The relative weighting technique can be subjective or objective. The subjective weighting of indicators is preferred for this CI. Different fashion companies should be able to use the CI and make company-specific trade-offs. Companies have different opinions, preferences, and targets. Subjective weighting allows companies to assign weights based on their values and preferences [180]. So, a subjective, relative weighting method for the indicators is desired in this CI. The Analytical Hierarchy Process (AHP) is an appropriate method that satisfies both conditions and is often used when sustainability indicators are included in the CI [87]. A fundamental property of AHP is that expert knowledge and experience are at least as important as the data for making decisions [190]. This technique allows for subjectivity, and the weight expresses the trade-off between the indicators. Also, this subjective weighting technique has a built-in consistency standard and scores high for transparency [134]. However, AHP is a time-consuming weighting method, as careful consideration must be given to selecting experts, computing weights, and verifying the consistency of the weights [133, 87]. An alternative method that satisfies the weighting conditions of this research is the Data Envelopment Analysis (DEA). However, this method is considered less suitable for this research as it may construct less realistic weights, affecting the robustness of the CI [161]. Therefore, AHP, a relative comparison approach with subjective weighting, is selected.

Lastly, the normalization method needs to be chosen. As AHP is selected, [120] suggests using either a ranking, standardization, or re-scaling transformation method for the normalization. The ranking is not an appropriate method when the objective of a CI is to reflect on the differences in the performances of the indicators [180], making this method unsuitable for fulfilling the objectives of this CI designed in this research. The re-scaling and z-score standardization methods can be used as they can measure the relative differences

between the variables [180]. The indicators show no variances in indicator bearing, implying they can all be interpreted similarly. For this research, lower values imply better performance. The re-scaling method is simple and can widen the ranges of small variable intervals; but is less capable of maintaining proportionally [134]. Z-score standardization can better handle outliers in the data but is not usable for a geometric aggregation method as this normalization method leads to negative values. Because GME is selected for aggregation, normalization will be done by re-scaling and not by z-score.

In summary, a visualization of the chosen methods for every phase can be found in Figure 4.1. The following subsections will provide further information on the techniques.



Figure 4.1: Conceptual design CI measuring the impact of product returns on companies' environmental and economic performance, source: (Author)

#### 4.3. How the CI for product returns is designed

This section will provide detailed information on the chosen methods for constructing the CI. The following five phases will be covered subsequently: selection of variables, normalization of variables, the weighting of variables, aggregation of variables, and as the final stage, the evaluation.

#### 4.3.1. Phase 1. Selection of the indicators

First, a conceptual framework that includes all relevant variables is created, which serves as a foundation for selecting variables. A comprehensive literature review on the product returns process is performed in chapter 2 to identify what is desirable to measure. Complementary to the literature, expert consultation is done for external verification. The relevant indicators for both dimensions are briefly described and discussed below.

#### **Environmental set of indicators**

The literature study has divided the returns process into several phases to assess the environmental impact. These are the preparation phase, collection phase, product handling and processing, and recovery and disposition phase. Different logistic activities identified from literature affect the magnitude of the environmental impact, as presented in section 2.6. For the final selection of the variables for the theoretical framework, this set of factors is narrowed down to a more comprehensive selection in which double counting is avoided. The environmental variables included in this CI are  $V_1$  packaging emissions,  $V_2$  transport emissions,  $V_3$  warehouse and equipment emissions, and  $V_4$  product recovery and disposal emissions and  $V_5$  brick-and-mortar emissions.

#### Clarification choices environmental indicators

Several choices for including indicators in the theoretical framework require additional clarification. First, the return preparation phase has identified packaging and labeling emissions as impact sources. The share of labeling emissions is considered to be relatively little relative to the packaging emissions and will not likely be of significant impact on this. Therefore, labeling emissions will not be included in the final theoretical frame-

work. Packaging is an important contributor to additional waste for a company, according to a manager of a fashion warehouse. Double counting of packaging emissions is avoided by using an Emisison factor (EF) that considers the whole life cycle of the packaging material for a particular waste treatment method [84]. The environmental impact of transport can be calculated using indirect transportation emissions and direct energy consumption. The values of these variables depend on the factors identified as crucial for the environmental assessment of the transportation phase like the type of modality or distance traveled. Vehicle energy consumption can be used as input for calculating transport emissions. However, including energy consumption and transport emissions in the environmental assessment increases the risk of double counting GHG emissions [84, 83]. This should be avoided. Considering the context of this research, the most appropriate variable to include is transport emissions rather than vehicle energy consumption to understand the companies' environmental impact in terms of GHG emissions. So, the GHG emissions from transporting a product over a certain distance are included in the framework.

There is also an overlap in the emissions and energy consumed by the equipment and vehicles used for handling and processing returns at the warehouse. Therefore, the same logic for avoiding double-counting is applied as described above. Emissions created by all warehousing equipment needed (i.e., conveyor belts, forklifts, stacking cranes) and emissions that arise in warehousing operations, including heating, cooling, and other sources of GHG pollutants, are included as one indicator in the final framework.

#### Economic set of indicators

A breakdown of the high cost of product returns for a company is given in section 2.5. A reduced set of variables is included in the theoretical framework. These are  $V_6$  transport costs,  $V_7$  labor costs,  $V_8$  warehouse and equipment costs,  $V_9$  recovery costs, and  $V_{10}$  brick-and-mortar store costs.

#### Clarification choices economic indicators

Expert consultation from Metyis and other fashion industry experts has confirmed the relevant costs for returns for a fashion retailer. The transport component also captures the logistics for arranging transport from the customer to the warehouse, including fixed, variable, staff, mode characteristics, and general operating cost elements. Labor costs refer to the direct and indirect costs for human resources and labor needed to handle returns, including hourly wages, benefits, and insurance. Utilization and rental of warehouse space and equipment for product returns are combined into one metric. This is because these costs are generally connected with companies' financial statements and are most of the time leased on an annual basis. Further, disposal and recovery costs of the packages and products are combined into one metric. According to an industry expert, these actions often take place together. Product recovery costs consist of either restocking, landfill, or incineration. For packaging recovery, recycling, landfill, or incineration alternatives are included. The utilization and processing costs for return operations in a brick-and-mortar store are integrated into one variable.

#### **Theoretical framework**

The theoretical framework for this research is shown in Table 4.1. This framework provides an overview of the dimensions, the variables representing each indicator, and units of measurement. The column on the right shows the direction of the variability of a variable on the company's performance. This attribute of an indicator is called indicator bearing [150]. A positive direction means that if a variable's value increases, the company's performance improves and vice versa. On the other hand, a negative direction indicates that a lower value of that respective variable is preferred to obtain higher performance [207], so for example, a high value of transport emissions represents worse environmental performance, indicating a negative variability [19].

#### 4.3.2. Phase 2. Normalization of the indicators

The normalization of variables will be done by re-scaling. Rescaling normalizes values based on the variable's range (minimum and maximum value) and widens the interval of all variables [134, 206, 180]. The normalized indicator values can be calculated using Equation 4.1 [203]. Here,  $min_i(x_{ij})$  and  $max_i(x_{ij})$  are the minimum and maximum values for indicator j considering all return methods or return policies i [180]. The aggregation of the indicators is done by multiplication, so values with a 0 will lead to an infeasible solution. Therefore, the formulation of the rescaling equation is adapted to Equation 4.1 from [203].

$$I_{ij} = 1 + \frac{max_i(x_{ij}) - x_{ij}}{max_i(x_{ij}) - min_i(x_{ij})}$$
(4.1)

Dimension		Description	Unit of measurement	Direction of variability
Environmental	V <sub>1</sub>	Packaging emissions	[kgCO2e/returned product]	Negative
	$V_2$	Transport emissions	[kgCO2e/returned product]	Negative
	V <sub>3</sub>	Warehouse and equipment emissions	[(kgCO2e/returned product]	Negative
	$V_4$	Product recovery emissions	[(kgCO2e/returned product]	Negative
	$V_5$	Brick-and-mortar emissions	[kgCO2e/returned product]	Negative
Economic	V <sub>6</sub>	Transport costs	[euro/returned product]	Negative
	$V_7$	Labor costs	[euro/returned product]	Negative
	$V_8$	Warehouse and equipment costs	[euro/returned product]	Negative
	V <sub>9</sub>	Recovery costs	[euro/returned product]	Negative
	V10	Brick-and-mortar costs	[euro/returned product]	Negative

Table 4.1: Theoretical framework including the indicators and units of measurement, source: (Author)

Where:

*i* The return method *i*,  $i = \{1, ..., n\}$ 

*j* The value of indicator *j*,  $j = \{1, ..., m\}$ 

 $x_{ij}$  The raw value of indicator *j* for return method *i* 

 $I_{ij}$  The normalized value of  $x_{ij}$ 

#### 4.3.3. Phase 3. Weighting of the indicators

The variables are relatively weighted according to the technique AHP, allowing for some subjectivity. The variables are compared per pair. First, the decision maker decides which variable in the considered pair is more important. Second, the decision maker expresses how much more important this variable is by assigning a score on a ratio scale [161, 159, 133]. The final values of the weights express trade-offs and are not measurements of the importance of indicators relative to another [134].

Each pair of two variables is compared based on a relative preference considering a given objective. The relative preferences of the decision-makers are defined on a semantic scale that ranges from 1 to 9 [134, 159, 144]. This scale is widely validated in literature and practical applications [159]. A value of 1 (weak preference) is given when both indicators are equally important, a five corresponds to a strong preference, and a 9 expresses an extreme preference. So, a nine should be interpreted as an indicator nine times as important relative to the other [142]. These preference relationships are further elaborated in Appendix B.

A *n* x *n* matrix, denoted by *A*, is constructed in Equation 4.2. Each element  $a_{ij}$  expresses the relative preference assigned by the decision maker for alternative *i* to *j*, where the alternatives represent the variables [142, 159]. As there are 10 variables identified for the CI in this research, the size of the matrix is 10 x 10. Several conditions should be met in this matrix. The pairwise comparison between the same variables should equal one, as expressed in Equation 4.3. Also, the reciprocal property should be satisfied (Equation 4.4) [190]. This results in a reciprocal matrix as presented in Equation 4.2, where  $a_{ij} = w_i/w_j$  [142, 100].

$$A = [a_{ij}] = \begin{bmatrix} 1 & a_{ij} & \dots & a_{1n} \\ 1/a_{ij} & 1 & \dots & a_{2n} \\ \vdots & \vdots & \vdots & \vdots \\ 1/a_{1n} & 1/a_{2n} & \dots & 1 \end{bmatrix}$$
(4.2)

Where:

$$a_{ii} = 1 \tag{4.3}$$

$$a_{ji} = 1/a_{ij} \tag{4.4}$$

After the expressions of the judgments in the matrix, the relative weights will be calculated by the principal right eigenvector approach of matrix *A* [159, 142]. This approach calculates a priority vector that represents the position of each variable relative to the others [142]. First, all elements in a column of matrix *A* are summed. Next, the columns are normalized by dividing the comparison scores of each cell in the reciprocal matrix by the sum of its column in (Equation 4.5), where  $ew_{ij}$  is the element weight. Third, the weights  $w_{ij}$  are obtained by taking the average of each row (Equation 4.6) [100].

$$\operatorname{ew}_{ij} = \frac{a_{ij}}{\sum_{i=1}^{n} a_{ij}}$$
(4.5)

$$w_{ij} = \frac{\sum_{j=1}^{n} \operatorname{ew}_{ij}}{n}$$
(4.6)

#### **Consistency check**

The comparison scores are based on the judgments of decision-makers. Therefore, as part of the AHP, it is important to perform a consistency check on the reasonability of the scores and the new priority vector values. The Consistency Ratio (CR) is a statistical test, which can be calculated with equation 4.7. In this formulation, *n* represents the size of the matrix,  $\lambda_{max}$  the largest eigenvalue of the *n* x *n* pairwise comparison matrix, and *RI* is the average random consistency index [134]. In Table 4.2 the critical *RI* values for the number of variables in matrix *A* can be found.

The calculated value of the CR indicates the level of consistency, of which the acceptable level depends on the size of the matrix. A significantly low value indicates an acceptable estimate of the relative weights [159]. There are specific baselines, which are when n=3, the CR should be 0.05, 0.08 with n=4, and 0.10 with n>=5 [167, 142]. A lower or equal value than this reference represents acceptable consistency in the comparative judgments. If this value exceeds the references, the judgments should be reviewed to reach an acceptable level of consistency [167].

$$CR = \frac{\lambda_{max} - n}{(n-1) * RI} \tag{4.7}$$

Table 4.2: RI values corresponding to the number of variables *n*, from [142]

n	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15
RI	0.00	0.00	0.58	0.90	1.12	1.24	1.32	1.41	1.45	1.49	1.51	1.48	1.56	1.57	1.59

#### 4.3.4. Phase 4. Aggregation of the indicators

The multiplicative aggregation approach, also called weighted geometric aggregation (GME), provides an in-between solution between the full and non-compensability of the different variables. By multiplying the weights and normalized indicators, a value of the CI is obtained that reflects the relative importance of each indicator and how much an individual indicator contributes to the total score of the CI [207, 203].

The previously calculated functions are used as input for the formula for aggregation (Equation 4.8). In this equation,  $w_j$  expresses the weight of indicator j for product return method i.  $I_{ij}$  is the normalized value of indicator j for return method i.  $CI_i$  expresses the value of the CI for return method i [203].

$$CI_i = \prod_{j=1}^{10} I_{ij}^{w_j}$$
(4.8)

#### 4.3.5. Phase 5. Post-analysis evaluation of the CI

The choices made in each phase in constructing the CI can all be subject to criticism. In theory, using different methods to build the CI is possible, which may lead to different results [207]. To improve the robustness of the CI, it is essential to provide transparency on the construction process in the first place, and a Uncertainty Analysis (UA) and Sensitivity Analysis (SA) are performed secondly [134].

#### **Uncertainty analysis**

For the UA, different input sources embedded with uncertainty are studied. These sources are referred to as uncertainty triggers [134]. The chosen method for the UA is sustained, followed by a description for performing that method. This research inspects three uncertainty triggers in more detail, the weights, aggregation method, and the input data. It is worth mentioning that it would add value to analyze the uncertainty of the normalization method. However, in the context of this CI, the rescaling method is the only appropriate normalization method considering the alternative methods. The z-standardization can measure differences between the values but cannot be used in combination with the multiplicative aggregation approach. Other

normalization methods, like ranking or distance to a reference point, do not take the absolute differences between the values into account and are therefore unable to measure or analyze the difference in performance [133]. This is a requirement for this CI; therefore, the uncertainty of the normalization method is not analyzed further in this research.

#### Uncertainty trigger $X_1$ . Weighting method

A source of uncertainty for this CI is the assignment of weights in the AHP, particularly because subjective weighting is used. The first trigger for this uncertainty analysis is, therefore, weighing. A different weighting scheme can gain a further understanding of the robustness of this CI. The AHP already performs a consistency check reflecting on the realism of the assigned weights. For this reason, rescaling weights is sufficient to examine the robustness and is referred to as the first trigger in this UA [133]. Rescaling the weights will be done by questioning a different group of experts who must assess the weights according to a defined perspective. The same criteria apply to the number of experts and the consistency ratio.

#### Uncertainty trigger X<sub>2</sub>. Aggregation method

The second trigger  $X_2$  of uncertainty is the aggregation scheme. The nature of this CI requires that the aggregation scheme satisfy the partial compensability of the indicators of this CI. Therefore, the non-compensatory weighting method is not suitable. Additionally, using a multi-criteria aggregation is incompatible when using AHP [133]. Considering these technical feasibilities, the arithmetic mean can be used, where the normalized values are summed to an arithmetic mean. This aggregation method can be more extreme than the geometric mean as it substitutes values to a larger extent with each other [180]. The following equation can be used to determine the values according to linear additive aggregation, Equation 4.9 and Equation 4.10 [133].

$$CI_i = \sum_{j=1}^{10} I_{ij} w_j \tag{4.9}$$

$$\sum_{j} w_j = 1 \tag{4.10}$$

Uncertainty trigger X<sub>3</sub>. Input data

Assumptions are likely to be made when not all input data is provided, so the uncertainty of this source will be inspected [134]. This will be done by making explicit assumptions and performing a SA.

#### Sensitivity analysis

The correlation of the uncertainty triggers on the final value of the CI are inspected with a statistical test [203, 147]. The variables are not expressed in ranks, so the Spearman correlation is unsuitable, and Pearson is used [99]. This test is two-tailed, as the variables can change in two directions (positive or negative) [147, 99]. A trigger's impact is isolated to analyze the correlation with Equation 4.11. In this formulation, n is the

A trigger's impact is isolated to analyze the correlation with Equation 4.11. In this formulation, *n* is the number of variables, and *x* and *y* are the variable sets  $(x_1, y_1), (x_2, y_2), \dots, (x_n, y_n)$  [99].

The coefficient *r* ranges from -1 to 1. A negative correlation is indicated by -1, 0 is no, and 1 expresses a perfect positive correlation [20]. For each trigger, a set of hypotheses are formulated. The null hypothesis  $(H_0)$  states no significant relation between the examined variables, and the alternative hypothesis  $(H_1)$  states a significant relationship at a certain confidence level [147, 99, 203]. The p-value determines the significance of this relationship.  $H_0$  is accepted if the significance level is below the p-value, and  $H_0$  is rejected otherwise, meaning there is a significant relationship between the variables [20, 203, 99]. The p-value is calculated using the t-distribution, presented in Equation 4.12 [99].

$$r = \frac{n\sum xy - (\sum x)(\sum y)}{\sqrt{(n\sum x^2 - (\sum x)^2) - (n\sum y^2 - (\sum y)^2)}}$$
(4.11)

$$t = r * \sqrt{\frac{n-2}{1-r^2}}$$
(4.12)

#### 4.4. Data collection

This section provides insight into the data collection requirements. The CI will investigate several return scenarios, which can be return methods or return policies. For each method, a value  $x_{ij}$  is calculated, expressing the raw indicator value j for product return method i. In total, there are 10 variables identified

from Table 4.1 and including their unit of measurement. These are:  $V_1$ : packaging emissions,  $V_2$ : transport emissions,  $V_3$ : warehouse and equipment emissions,  $V_4$ : product recovery emissions,  $V_5$ : brick-and-mortar emissions,  $V_6$ : transport costs,  $V_7$ : warehouse and equipment costs,  $V_8$ : labor costs,  $V_9$ : recovery costs,  $V_{10}$ : brick-and-mortar costs.

#### Required data and guidance on the data collection

This subsection guides which data needs to be collected to measure the variables, what sources to use, and how to analyze the data.

#### V1. Packaging emissions

The required data to determine the packaging emissions are the type of packaging material and the weight of the packaging material, which should be collected through external client data. A material-specific EF can be retrieved from the literature, CO2emissiefactoren, DEFRA emission sources, and EPA. This EF is multiplied by the mass of a package according to the approach suggested by [84]. Further, data on the waste treatment strategy should be collected to select the correct EF for the waste treatment. If the client cannot supply this data, assumptions are made about the most frequently used materials for packaging and corresponding waste treatments.

#### V2. Transport emissions

The transport emissions can be calculated by the fuel- or distance-based method. The appropriateness of the methodology for transport emissions depends on the analysis objectives and the availability of data [84]. The fuel method requires detailed data on the volume and fuel types used [84]. This level of accuracy in data may not be feasible for the time scope of this study. The distance-based approach requires less detail in data [84]. The following data needs to be gathered to calculate the transport emissions with this method: the transport modes used, the distance traveled per returned product, the return method, the transported weight or volume of the product, and the vehicle capacity. Different data sources can be used and combined to retrieve the necessary information. The client should provide data on the input sources. This provider should also be contacted if the transport process is outsourced to a Third Party Logistics Provider (3PL). Knowledge of the type of modality is needed to select the correct EF, which can be retrieved from external sources like CO2emissiefactoren, DEFRA, EPA, or TNO. Different transportation networks for each return method should be considered to calculate the emissions to obtain accurate values. The emissions are determined according to in-store returns, home pick-up, and collection points network structures. A more detailed description of the data and additional assumptions is provided below.

#### **Return method - Brick-and-mortar returns**

The transport emissions are calculated using the distance method as collecting fuel data of customers' shopping trips is complex and not assumed to be feasible within the time scope of this research. The input data necessary are the modality, the mass of a product, the customer's location, and the store's location. If the product is transported from the store to the warehouse, additional data should be collected on the vehicle route, vehicle type, warehouse location, and total journey distance traveled. The client should provide the location of the customer and the physical stores. This is used as input to determine the distances traveled in kilometers. If it is impossible to collect this input data, assumptions for the average distance driven for a shopping trip in the Netherlands are made based on data from Centraal Bureau voor de Statistiek (CBS) [118, 27].

The level of accessibility affects the choice of the modality of citizens [27]. Slow modes of transportation (bike, walking, bus, metro, tram) are preferred in the Netherlands under the threshold of 6 kilometers [27] if distances longer than 6 km need to be traveled, cars are mainly used for shopping trips [27, 28]. The top three most used cars per fuel type are petrol, diesel, and electric vehicles in the Netherlands [29]. Suppose it is impossible to retrieve any data on the modality used for this transport by customers. In that case, it is assumed that when a customer needs to travel less than 6 kilometers to the physical store, an average EF of the slow transportation modes is utilized. When the customer needs to travel distances larger than 6 kilometers, an average EF for the car will provide a reasonable approximation [84].

#### **Return method - Home pick-up**

To determine the transportation emissions per product return, the following data must be collected with the client and supplemented by 3PLs: the number of return requests served in one journey, the vehicle charac-

teristics and type, the location of the return requests, and the location of the final destination. Suppose it is not possible to retrieve this complete set of data; in that case, assumptions should be made on the number of stops per journey based on the vehicle's maximum payload and the package's dimensional mass. This can be done by using an average EF for the transport vehicle [84], and the average distance per pick-up round (50 km) can be used for the distance calculation [118]. In the worst-case scenario, the emissions can be calculated using the total transport costs and the modality used [84].

#### **Return method - Collection point**

Three transport activities need to be calculated. The transport emissions for the first activity can be determined similarly to the in-store transport emissions, where the distance is determined from the customers' location to the collection point, using data from the retailer and 3PL their collection points. The latter is available on the websites of the 3PL. The second transport activity to consider is vehicle journey emissions. The input data necessary for determining the transportation emissions per product by a collection point are the number of collection points served in a single route, the location of the collection points and the final destination, the vehicle characteristics, and the number of parcels picked up per collection point. The client should supply this, preferably in cooperation with their 3PL. If this data cannot be collected, assumptions should be made for this measure. The number of parcels picked up is limited by the vehicle's maximum payload capacity and time constraints. Based on the dimensions of a product package, an estimation of the number of packages transported by one vehicle can be made. The average distance driven on a collection round can be used for the distance calculation if needed [51], and an average EF for the transport vehicle can be used if data lacks [84]. This can be found in the literature. In the worst-case scenario, the emissions can be calculated using the total transport costs and the modality used [84].

#### V3. Warehouse and equipment emissions

Energy consumption data is expressed over a certain period and should be available to the client. Once data is gathered from the warehouse's energy consumption, the warehouse emissions can be determined. A proportion of the total warehouse emissions will be allocated to product returns depending on the area. The warehouse emissions given to returns should be divided by the number of returns handled at the warehouse over the same period considered to determine the warehouse emissions per product return. The client should provide the following data: fuel and electricity consumption, warehouse space, and return handling area. Supplementary data can be found through literature studies; for example, [42] studied returns in warehouses, [75, 154] made an environmental assessment of general warehouse emissions, and [118] has taken warehouse activities into account in their carbon assessment. The EFs for this calculation are most accurate when location-specific; sources are CO2emissionfactors, DEFRA, and EPA. In the worst-case scenario, the emissions can be approximated using the total space used for product return handling at the warehouse, the cost structure (leased or owned), and the warehouse location [84].

Data should be collected on the number of warehousing equipment, the type of equipment, the operating hours per day, and the power consumption of the equipment. According to an expert with industry knowledge, the client, their warehouse manager, should have insights into these numbers. It could be that this data cannot be provided entirely by the client source for this research. In that case, additional assumptions should be made based on the type and number of equipment used for handling/processing returns at warehouses and the energy consumed by equipment in a warehouse. Studies of [75, 153] have assessed the emissions of warehouse equipment in the US. If the total equipment emissions are known, a fraction of the whole equipment emissions must be allocated to product returns. This will be the final value for this indicator.

#### V4. Product recovery emissions

The emissions created during the product disposition depend on the mass of the product and the chosen waste treatment. The client should collect data on the number of products handled by a treatment method, the type of treatment method, and the mass of the product. Client data will be the most accurate and reliable source of data. If no data is given on the recovery method chosen, insights are gathered through interviews with experts in the industry. The waste treatment EFs can be derived from DEFRA, CO2emissionfactoren, or EPA.

#### V5. Brick-and-mortar store emissions

To measure brick-and-mortar store emissions, data should be collected on the consumed energy at the store and the time a product return spends there. The client should provide this data, as they receive invoices for their energy consumption and electricity used at the store and have insight into their product stock levels. If this data cannot be provided, assumptions can be made regarding the store's size, location, number of products handled, and energy consumed. In the worst case, information on the store area and the number of returns dealt with there [84] should be collected, which can be retrieved from the literature or publicly available reports.

#### V6. Transport costs

The data that should be collected to calculate the transport costs are the distance driven; the modality used, the vehicle load, and the mass of the transported package. Sources to contain this data are the external client and possibly the 3PL to have the desired level of accuracy in the transport costs. Literature or reports can be used as a secondary source if this data is insufficient; for example, [187] provides detailed insight into key economic figures of freight transport. Once the total transport cost for a vehicle route is known, this amount should be converted to the cost per product by dividing the total costs by the number of products transported. When this data cannot be retrieved, assumptions should be made based on the shipment sizes. Therefore, a vehicle's maximum payload capacity and the product package's length should be known or determined [45].

#### V7. Labor costs

Necessary data is the number of employees, the time spent per employee on inspection, restocking, band receiving of product returns, the total labor costs of employees at the warehouse, and the external overhead expenses. The client should provide this data. By dividing the total labor costs for returns by the number of product returns dealt with during the period, the labor cost per product return can be determined [45, 50]. Suppose this data cannot (entirely) be provided. In that case, additional assumptions should be made based on the workforce required for handling/processing returns at warehouses and the total labor costs of warehouse employees. This data can be found in studies of [89] and [42]. Data on total costs can be found in companies' financial statements, and based on literature, assumptions can be made about the workforce needed.

#### V8. Warehouse and equipment costs

Cost on the handling and processing of return logistics at a warehouse provided by an external client. These costs include rent, mortgage payments, utilities, insurance, and security. A proportion of these costs should be allocated to return handling. This proportion is based on the time a product returns spent at the warehouse or the number of products over a time period. It is preferred to get this data from the client. Additional assumptions should be made if this data is not entirely provided on the area dedicated to handling/processing returns at warehouses and the cost of operating, which need to be verified using expert consultation.

#### V9. Recovery costs

Recovery costs consist of the recovery cost of a product and its packaging. The number of products per recovery or disposal method should be provided to derive a product's recovery cost. Country-specific disposition fees for landfill or incineration can be derived from European statements when client inputs on this lack [69]. This data can be supplemented by an expert consult specializing in clothing disposition. If this data is insufficient, secondary literature sources can be used. The cost for storage of a product is derived from the total storage costs at the warehouse and the number of product returns stored, preferably provided by the client. Otherwise, this can be approximated using findings of the study of [89] and expert knowledge. The packaging disposal and incineration fees can also be derived from the municipal waste charges if client input is not provided [39]. This can be supplemented with packaging-material-specific recycling fees from literature [73].

#### V10. Brick-and-mortar costs

The data that should be collected are the following: rent, mortgage payments, labor, insurance, and security costs. To specify the return costs, information should be provided on the estimated time and space for return handling tasks at the store and the number of products handled in the store over a time period. The primary source of data is client data. Otherwise, assumptions should be made based on the labor needed area dedicated to returns at a store to handle product returns based on the literature or reports.

#### 4.5. Main findings chapter 4

This chapter aimed to answer SQ4c) What methods should be selected to construct the CI for product returns assessing environmental and economic performance?

The following methods should be selected considering the requirements and findings of the previous chapters: normalization by rescaling, AHP weighting, and multiplication aggregation, which is visualized in Figure 4.1. The data collection requirements and guidelines are identified for every indicator based on internal and external sources. Additional attention is paid to the availability of data sources, as this affects the requirements and measurement method. Several alternatives are defined for each indicator, and experts check whether a company can obtain the input data requirements.

## 5 Measurement with the Composite Indicator

This chapter will design the Composite Indicator (CI) to measure a company's environmental and economic performance for product returns. First, an introduction is given to the fashion retailer of the base case in section 5.1, followed by a description of the data set used. Second, the data collected for each indicator and additional assumptions will be described in section 5.2. Third, the model is designed in section 5.3, where the performance of the base case is measured. Lastly, the insights found with performance measurement are analyzed and evaluated in section 5.4. Question 4d) 'How can the current state of a company's performance be measured using the CI?' is answered.

#### 5.1. Description of the base case

Data from a fashion retailer is collected to test the CI and to analyze and compare the performance of the return methods. The retailer is an international fashion company active in online and offline retailing channels. The fashion company is part of a larger global fashion group. The company offers all types of fashion clothing, such as shirts, dresses, tops, trousers, skirts, blouses, and jeans. The company is mainly active in the downstream supply chain, which entails the production of clothes by manufacturers to the point of customer consumption. The management and operations for handling returns are also part of the downstream supply chain activities. The company offers free returns to customers within 30 days of receiving the item, and depending on the channel the item was purchased, a return method can be selected.

The customer data is limited to the geographical scope of the Netherlands. This means that only customers returning items from the Netherlands are included. The data shows all products purchased and returned at the retailer's e-commerce channel and brick-and-mortar store for the time period of one year (Jan.-Dec.). The retailer operates one brick-and-mortar store in the Netherlands where customers can hand in returns. Other return alternatives offered are through home pick-up and at collection points.

The products collected at the brick-and-mortar store are either resold in-store or returned to the warehouse. Returns handed in through collection points or home pick-up can be transported to the warehouse or relocated to the brick-and-mortar store where they are resold. The retailer owns a warehouse for their European operations that is entirely dedicated to returns, which is located in Spain. PostNL and UPS transport the returns as Third Party Logistics Provider (3PL). Both 3PLs have several collection points and distribution centers in the Netherlands. Products purchased and returned through other franchise retailing channels like Zalando or AboutYou are excluded from the data as this is less accurate than the data owned by the fashion company.

#### Data specifications of the base case

The data provided by the client retailer entails return data for a whole business year in 2020. The data set contains 1735 return orders from different customers in the Netherlands. The number of returned items per order is also known, resulting in 6388 returned items in the base year. The frequency of return methods used was 92.45% by collection point (1603 packages), 6.11% by home pick-up (106 packages), and 1.44% by brick-and-mortar returns (26 packages). As said, the number of items per return package is also known. In

total, 5912 items were transported by collection points, 407 by home pick-up, and 69 were returned at the brick-and-mortar store.

#### 5.2. Data preparation and analysis

The data must be prepared, analyzed, and structured before being used as input for the CI. The previous chapter has provided insight into the data requirements for each variable (see section 4.4). The collected data and quality are elaborated on in this section, Appendix D provides an overview of the data.

#### V1. Packaging emissions

Packaging waste is only created for the retailer when the product is shipped using the return methods: collection point returns, home pick-up, and if the products are transported from the store to the warehouse. The retailer's data gives insight into which return method is used so that packaging emissions can be allocated accordingly. The retailer from the base case offers a wide assortment of apparel, so a combination of packaging materials is likely used. From the expert consultation, it follows that all products are individually packaged in a plastic polybag (Low-density polyethylene (LDPE)), and to ensure safe transport, either a cardboard box or a plastic shipping bag is used. Both packages are used often according to an expert and [191], so it is assumed that this is 50-50 in the base case. This is randomly decided in the model in Python. According to the same industry expert, most of the packaging material is sent to disposal and destroyed, and only a meager share is recycled in fashion mainly. This is partly due to the low quality of the packaging when it gets returned. In addition, the material type of returned packaging differs as almost 80% of the customers do not return their product in their original packaging [191]. These factors make the recycling of packaging more difficult. It is therefore assumed that no packaging is recycled in the base case. Whether the package is sent to a landfill or incineration is randomly assigned by the model in Python. Municipal waste treatment facilities will likely handle all packaging waste [39]. The mass of an average package is derived from literature sources, which can be found in Appendix C. The Emisison factor (EF) used for the measurement of material-specific packaging waste can be found in Appendix E under packaging waste treatments.

#### V2. Transport emissions

The data provides insight into the chosen method of return and the 3PL, the location of origin, final destination, the mass of the product, and its packaging. With these inputs, the transport emissions for each return are calculated using the distance method, considering each trip between recovery locations separately. This will improve the reliability of the calculation because each trip has different characteristics, and other transport modalities are involved. The total transport emissions of a returned product are the sum of these trips. The retailer only uses road transportation, which PostNL and UPS perform. The return methods offered by the retailer are in-store returns, collection point returns, and home pick-up. PostNL only provides home pickups. Products returned by customers at the brick-and-mortar store can be transported to the warehouse by PostNL or UPS. Both 3PLs have distribution centers for packages spread throughout the Netherlands. The model assumes that the parcels are transported to the nearby distribution center of the specific 3PL and, from there, are transferred to another modality to be transported to the final destination. The city's firstand last-mile transport is done by distribution vans and the long-haul trip by trucks. The distances are measured using Google Maps and implemented in Python. For home pick-up trips, it is assumed that the first address severed is of the customer, whose exact location is known, with an average pick-up trip distance of 50 km [118]. This implies that the vehicle serves 100 customers with an average distance of 0.5 km between the customers [40]. For the collection point trip, it is also assumed that the first location of the stop is the collection point of the customer, with an average trip distance between the stops of 80 kilometers [51]. For the long-distance trip, distances are measured between the distribution points of the 3PL providers and the final destination using Google Maps. No data is provided on vehicle characteristics. Assumptions are made based on the vehicles of the 3PL of the executed trip. According to [41], almost all trucks are diesel-fueled in the Netherlands, which is assumed to be the same for the base case. Based on [110], freight emission values are derived for the trucks. This study expresses the load factor based on the type of cargo transported, which determine the emissions created per tonne-kilometers. The distinction is made between light, medium, and heavy loads. The freight transported for this study falls under the light bulk and piece transport category, which corresponds to less than 0.4 kg/L in the cargo area [110]. Most distribution vans in the Netherlands are diesel-fueled, followed by petrol [41]. The values of the EF used for the different modalities are obtained from multiple databases (i.e., CO2emissiefactoren, CE Delft, BEIS). They are presented in Appendix E under freight *transport.* For the base case, the EF for the truck trailer light diesel is used for long-haul transportation; for short-haul transportation, the EF for vans with an average fuel type is used.

For the trip from the customer to the store and the distribution centers, statistics of Centraal Bureau voor de Statistiek (CBS) are used to specify the modality as explained in the previous chapter. It is assumed that when a customer needs to travel less than 6 kilometers to the physical store, an average EF of the slow transportation modes (bike, walking, bus, metro, tram) is used. When the customer needs to travel distances larger than 6 kilometers, an average EF for the car will provide a reasonable approximation [27, 29, 84].

#### V3. Warehouse and equipment emissions

The warehouse's energy consumption or rating is unknown from the retailers' data. The warehouse emissions are therefore determined according to the proposed method of [84], which is based on the warehouse area. The warehouse area for returns and the number of returns handled over a time period are obtained from client data. An EF for warehouse energy is derived from the Greenhouse Gas (GHG) Protocol database specified for a leased warehouse facility, which can be found in Appendix E. The area is multiplied with this EF and divided by the number of returns to derive the warehouse emissions per returned product. Next, the retailer has given insight into the type and number of equipment used for handling product returns at the warehouse. With this input, the equipment emissions for handling the returns at the warehouse location and measured in monetary values. Prices for the equipment are derived from several large equipment manufacturers or suppliers, like Toyota Material Manufacturing, Bigrentz, and AltaEquipment. The client has shared the number of returns handled in the warehouse, which will be used to derive the warehousing and equipment emissions per item. This is presented in Appendix F.

#### V4. Product disposal emissions

The disposal emissions are determined using the method of [84] for waste, as the client has only provided information on the mass of the products and not on which waste treatment is used. It is chosen only to limit the emissions of recovery to reselling the product and disposal to landfill or incineration emissions of the products. Particularly in the fashion industry, the most significant proportion of returned products is either resold and reused or sent to incineration or landfills. Recycling rates of textile waste, including both returns and non-returned products, in Europe is below 1% [122, 52]. Considering the proportion of recycled items from returned products, it is assumed that a value close to zero is likely. Little information is known about the exact number of recycled items, which is also true for items that get repaired. Multiple industry experts confirm this lack of information and transparency in business operations. In addition, [166] studied the environmental footprint generated by recycling clothes in Japan. They concluded that the GHG emissions of recycling a clothing product highly depend on the type of recycling method chosen, the type of product that is recycled, and which sort of product it will be recycled into. So, estimating the environmental impact of recycling a clothing item is very complex and surrounded by a lot of uncertainty, which is even more difficult as there is little data available and it is highly product dependent. This implies that for including recycling as a recovery option in this base case, the additional assumptions that need to be made will likely be insignificant and inaccurate. It is therefore chosen only to focus on reselling/reusing the product (direct recovery) and disposal to landfill or incinerated (process recovery). This research assumes, therefore, that recovered, recycled, or repaired parts are stored and resold again if not mentioned differently, and the features that cannot be retrieved are eventually redirected to landfills or incineration.

Waste-treatment specific EFs for clothing are selected from the BEIS database from DEFRA, presented in Appendix E under *Waste treatment*. This gives insight into the landfill-, reuse-, and incineration of kgCO2-equivalent emissions per kg of clothing, including the collection and transport to the distribution facility [43]. There is no specific data available for determining the frequency of disposal treatments. Therefore, insights from an expert consultation with a fashion warehouse manager and an expert in the field of textile recycling are gathered to gain more understanding of this matter, combined with findings from [4]. Most items are reused and resold again, but there is also a share that gets sent to landfill or incinerated. This has led to the following allocation for the base case: it is assumed that 80% of the products are resold and reused, 10% are sent to landfill, and 10% are sent to incineration. The model assigns the waste treatment to the return orders based on a random assignment to improve reliability.

#### V5. Brick-and-mortar emissions

No data is provided on the store's energy consumption by metered data. This can be approximated using the energy label and the area of the retail store. Each label corresponds to a maximum primary fossil energy consumption in the Netherlands in kWh/m2/year, with A the best and G the worst performance [155]. Following the energy rating of the store, which is C, the value of the maximum primary fossil energy consumption in the Netherlands is obtained. This number is multiplied by the area of the store in m2, which estimates the store's energy consumption in kWh for a year. The total emissions are allocated to the total number of products handled in-store for the year, so items in inventory, sold, and returned in-store are considered. As no specific data is available on the time spent at the store for each item, it is assumed that the total emissions are equally distributed over the total number of products handled in-store in one year. These values can be found in Appendix D.

#### V6. Transport costs

The data provided by the external client for transport costs are the customer's location, the initial point of recovery, and final destination, the transported mass, and the 3PL used. From [41], it is found that for long-haul transport, diesel trucks are used for short-haul distribution vans that are mostly diesel- and sometimes petrol based. Based on these inputs, conversion factors of the Netherlands Institute for Transport Policy Analysis (KiM) are used [187]. This modality-specific conversion factor includes fixed, variable, staff, mode characteristics, and general operating cost elements. The most recent report uses fuel prices from 2018 [187]. Based on the type of vehicle used for the transport, a mode-specific conversion factor in [euro/kg-km] is multiplied by the distance driven in kilometers and the parcel mass in kilograms to derive the transport costs per kg returned product. The distance driven is already determined for the measurement of the transport emissions. This is presented in Appendix F. For the base case, the costs for a diesel truck for transporting light bulk goods are used; for short-haul transport, the costs of an average fueled van are used.

#### V7. Labor costs

The labor costs are calculated by the sum of receiving, inspecting, and restocking products and any additional overhead expenses. Data is provided on the number of warehouse employees. Further data is found from literature, governmental reports, and expert consultation. Expert consultation has given insight into the total activity time of a return handling in-store and at warehouses on average. This activity is quite labor-intensive as the product needs to be inspected to determine if it is good enough to be restocked. The warehouse's labor costs are determined based on the hourly return rate and the average time needed to take a return [89]. Further, the employee overhead expenses depend on the country where the labor is performed and is retrieved from EuroDev. The total costs per product return at the in-store are calculated by the labor cost per hour, including the overhead expenses, which must be multiplied by the time spent fulfilling the tasks. The minimum wages are derived from Rijksoverheid, and the overhead costs are found in the database of Eurodev.

#### V8. Warehouse and equipment costs

Data on the warehouse size, the warehouse location, the leasing structure, the number of equipment, and the type of equipment is provided by the retailer. The missing component for calculating the costs is the warehouse leasing costs on an annual basis. For the warehouse leasing costs, data sources from Statista provide the average warehouse rental costs per square meter for a specific location [173]. Additional consultation with a fashion warehouse manager has provided insight into the most commonly used type of equipment and loading capacity for retailer warehouse operations to specify the equipment type further. Data from secondary resources of large equipment manufacturers or suppliers, like Toyota, Bigrentz, and AltaEquipment, are used to estimate the costs of the equipment.

#### V9. Recovery costs

The recovery costs of a product depend on the disposition method chosen, the mass of the product, the disposition fees, and storage costs. There is no data on the retailer's disposition costs, so secondary sources from European reports, expert consultation, and literature are used to collect data. It is assumed that municipal waste treatment facilities handle all waste. Country-specific disposition fees per kg can be retrieved from [69] and supplemented by [53]. The mass of a product is known so that the disposition costs can be accurately determined for each item. Expert consultation and literature have been used to assess product

reuse/redistribution costs. This is assumed to be half the cost of return handling as the product needs to be inspected and stored, so resorting and repackaging activities are considered the most cost-intensive activities. The cost of return handling is derived from a study of [89]. The cost of packaging waste per item can be measured using European reports' disposition and recycling fees. These costs are measured in euros/kg, so the mass of the packaging is another input for this measurement and can be found in Appendix C.

#### V10. Brick-and-mortar store costs

The brick-and-mortar store location, store area, and leasing structure are known. To determine the rental costs, an estimation is made using the report of [38]. This report has provided a detailed insight into the rental fees for shopping areas in the store's location and is used as a source for cost estimation. Data is collected on the number of products handled in stores in the considered year and the number of returned items. Based on this data, a fraction of this cost can be allocated to an item.

#### 5.3. Design of the CI

In this section, the CI to measure the environmental and economic impact of product returns on a company's performance is created using the collected data as input. The model is designed using the data from the base case. The different return methods are also analyzed separately for the base case. This section follows the construction structure of the CI, as outlined in section 3.3.

#### 5.3.1. Phase 1. Selection of the indicators

Ten variables are aggregated into the final value of the CI. Variables 1-5 are categorized as environmental variables, and 6-10 as economic performance variables. The data is collected, prepared, and analyzed for the base case. The raw values of the CI are presented for the overall performance of the fashion retailer of the base case. Also, the performance of the different return methods for this retailer is analyzed and measured, which can be seen in Table 5.1. The red values indicate the highest value of the return methods on that indicator, which is not desired. The green numbers correspond to the lowest, and thus most desirable in terms of performance, values of a return method for an indicator.

Table 5.1: Raw indicator values for each return method in the base case, source: (Author)

	V1	V2	$V_3$	$V_4$	V5	$V_6$	V7	$V_8$	$V_9$	V10
	[kgCO2e/	[kgCO2e/	[kgCO2e/	[kgCO2e/	[kgCO2e/	[euro/	[euro/	[euro/	[euro/	[euro/
	returned product]									
Base case overall performance	0.075	32.217	1.092	0.012	0.712	0.230	12.379	5.096	0.523	1.903
Return method 1. Home pick-up	0.077	23.721	0.624	0.005	0.784	0.432	7.076	4.721	0.280	2.093
Return method 2. In-store	0.013	42.567	1.559	0.016	0.642	0.073	17.666	5.470	0.755	1.896
Return method 3. Collection point	0.078	32.592	1.118	0.010	0.710	0.217	12.673	5.117	0.539	1.896

With  $V_1$ : packaging emissions,  $V_2$ : transport emissions,  $V_3$ : warehouse and equipment emissions,  $V_4$ : product recovery emissions,  $V_5$ : brick-and-mortar emissions,  $V_6$ : transport costs,  $V_7$ : warehouse and equipment costs,  $V_8$ : labor costs,  $V_9$ : recovery costs,  $V_{10}$ : brick-and-mortar costs.

#### How the environmental and economic impact of a return method can be calculated

Table 5.1 presents for each return method the raw value per returned product for each indicator based on the data of this retailer. This is used to calculate the total environmental and economic impact per return method. For the environmental and economic dimension, the raw values of all indicators are summed, as seen in Table 5.2. It can be seen that in-store returns have both the highest environmental impact and the largest costs. Contrary, home pick-ups have lower kgCO2e per returned product as well as the cost per returned item. The collection point returns have values that are in between the home pick-up and in-store methods.

Table 5.2: Summation raw values of the indicators for the environmental and economic impact per return method, source: (Author)

	Environmental impact	Economic impact
	[kgCO2e/	[euro/
	returned product]	returned product]
Base case overall performance	34.108	20.132
Return method 1. Home pick-up	25.211	14.602
Return method 2. In-store	44.797	25.678
Return method 3. Collection point	34.509	20.443

#### 5.3.2. Phase 2. Normalization of the indicators

The data from Table 5.1 is normalized using the adapted rescaling method proposed in Equation 4.1, where all variables are normalized between 1 and 2. 1 represents the least desired value in Table 5.3, and 2 is the best in terms of performance.

	$V_1$	$V_2$	$V_3$	$V_4$	$V_5$	$V_6$	$V_7$	$V_8$	$V_9$	$V_{10}$
Base case overall performance	1.055	1.549	1.499	1.345	1.504	1.563	1.499	1.499	1.487	1.504
Return method 1. Home pick-up	1.024	2.000	2.000	2.000	1.000	1.000	2.000	2.000	2.000	1.000
Return method 2. In-store	2.000	1.000	1.000	1.000	2.000	2.000	1.000	1.000	1.000	2.000
Return method 3. Collection point	1.000	1.529	1.471	1.507	1.520	1.598	1.471	1.471	1.454	1.520

Table 5.3: Normalized indicator values for each return method in the base case, source: (Author)

With  $V_1$ : packaging emissions,  $V_2$ : transport emissions,  $V_3$ : warehouse and equipment emissions,  $V_4$ : product recovery emissions,  $V_5$ : brick-and-mortar emissions,  $V_6$ : transport costs,  $V_7$ : warehouse and equipment costs,  $V_8$ : labor costs,  $V_9$ : recovery costs,  $V_{10}$ : brick-and-mortar costs.

#### 5.3.3. Phase 3. Weighting of the indicators

In this stage, experts are requested to express relative preferences between each pair of variables to derive the weights using the Analytical Hierarchy Process (AHP) method. The experts operate in the fashion industry and have relevant knowledge and expertise in the retailer's objectives, criteria being evaluated, and the decision-making context. The experts are selected independently to avoid biases, and the perspective is explicit upfront. The pairwise comparison matrix presented in Table 5.4 is derived from these expert weighting. The scores should be interpreted as  $V_2$  (transport emissions) is believed to be four times more preferred than  $V_1$  (packaging emissions).

Table 5.4: Reciprocal comparison matrix, source: (Author)

	$V_1$	$V_2$	$V_3$	$V_4$	$V_5$	$V_6$	$V_7$	$V_8$	$V_9$	$V_{10}$
$V_1$	1	1/4	1/2	1/2	1	1/4	1/3	1/2	1/2	1
$V_2$	4	1	2	2	3	1	1/2	2	2	6
$V_3$	2	1/2	1	3	2	1/2	1/2	1	2	3
$V_4$	2	1/2	1/3	1	1/3	1/2	1/3	1/2	1/2	3
$V_5$	1	1/3	1/2	3	1	1/2	1/4	3	2	1
$V_6$	4	1	2	2	2	1	1/3	2	2	3
$V_7$	3	2	2	3	4	3	1	3	6	2
$V_8$	2	1/2	1	2	1/3	1/2	1/3	1	2	1
$V_9$	2	1/2	1/2	2	1/2	1/2	1/6	1/2	1	2
$V_{10}$	1	1/6	1/3	1/3	1	1/3	1/2	1	1/2	1

With  $V_1$ : packaging emissions,  $V_2$ : transport emissions,  $V_3$ : warehouse and equipment emissions,  $V_4$ : product recovery emissions,  $V_5$ : brick-and-mortar emissions,  $V_6$ : transport costs,  $V_7$ : warehouse and equipment costs,  $V_8$ : labor costs,  $V_9$ : recovery costs,  $V_{10}$ : brick-and-mortar costs.

Next, the weights are determined in Table 5.5. This is done using Equation 4.5 and Equation 4.6. The total sum of the weights is equal to 1.

#### Consistency check

The Consistency Ratio (CR) is calculated using Equation 4.7. A value of 1.49 is taken as the baseline for the acceptable level of consistency for the CR of this CI [142]. The consistency check with  $\lambda_{max} = 10.942$  results in a CR value of 0.070. This is below the threshold value of 0.10, implying an acceptable consistency in the comparative judgments [167].

#### 5.3.4. Phase 4. Aggregation of the indicators

The value of the different return methods in the CI are calculated using the normalized value and the weights obtained from the previous phases by applying the multiplicative aggregation approach in Table 5.6. A higher value for the CI indicates better performance. The analysis shows that the in-store return method scores lowest, visualized with the red color. The home pick-up method performs better than the other return methods and is represented in green. The base case consists of the combined value of all return methods to provide

Table 5.5: Weights derived by AHP, source: (Author)

Weight
0.044
0.158
0.104
0.060
0.085
0.134
0.228
0.075
0.063
0.048

insight into the retailer's overall performance. If the company would implement an action, measure it with this setting on the CIs, and obtain a higher value for its aggregated CI, this would mean that a better performance is obtained. If the score of the CI would be decreased after implementation, this would imply that a lower performance is gained. As can be seen, the base case overall performance and the collection-point returns are relatively close. This is as expected, as the returned products are for 92.45% performed with the collection-point method for this data set, and the overall base case performance is the average of the return methods.

Table 5.6: Aggregated CI scores base case, source: (Author)

	CI score
Base case overall performance	1.482
Return method 1. Home pick-up	1.613
Return method 2. In-store	1.241
Return method 3. Collection point	1.479

#### 5.3.5. Phase 5. Post-analysis evaluation of the CI

The post-analysis evaluates the robustness and reliability of the constructed CI. Therefore, several uncertainty triggers were identified during the construction of the model ( $X_1$  weighting method,  $X_2$  aggregation method,  $X_3$  input data), see subsection 4.3.5. The insight should be obtained if- and to what extent different construction methods or changes in the input data would lead to significant changes and differences in final values of the CI. Therefore, the uncertainty sources are analyzed, and their correlation with the model results obtained in this chapter is checked. Context scenarios should be developed to gain insight into the sensitivity of the construction methods and input data of the CI.

#### **5.4.** Conclusions performance measurement

This section concludes and gives insight into the performance scores the CI obtained. First, the environmental and economic impact of the return methods before aggregation is analyzed and compared with the outcomes of the aggregated scores, followed by an evaluation of the aggregated values of the different return methods and their performances.

#### Performance scores before and after multi-dimensional aggregation

In Table 5.7, the scores of the return methods are visible for each dimension separately. The indicators of the environmental and economic dimensions are weighted and aggregated but not combined to a multidimensional CI yet, which provides insight into how each method performs before aggregation. A higher value indicates better performance and vice versa. In-store returns perform lowest on the environmental scores; collection point returns perform somewhat better, and home pick-ups score best on environmental performance for this company. The economic scores show that home pick-up performs best, followed by the collection point returns with a relatively small difference in performance score. In-store returns have a relatively large difference in the economic performance with the other return methods, indicating that this method has worse economic performance than the other methods. As seen in Table 5.6, the same conclusions can be drawn when the dimensions are aggregated to reflect on the overall performance of a company. Home pick-up outperforms the other methods, followed by collection point returns, and in-store scores are the lowest based on this data set.

Table 5.7: Environmental performance score and economic performance score of the return methods, source: (Author)

	Environmental performance	Economic performance
	score	score
Base case overall performance	1.334	1.513
Return method 1. Home pick-up	1.545	1.588
Return method 2. In-store	1.150	1.260
Return method 3. Collection point	1.339	1.504

#### Base case return methods performance

To better understand the differences in performance of the return methods, Figure 5.1 is created. This radar chart presents the values of each return method on the indicator after the assignment of the weights. The orange line depicts the value of a return by home pick-up, the blue line represents in-store returns, and the green line displays returns through collection points. The indicators are defined on the spokes of the radar. For each indicator, better performance is obtained when the value is higher, corresponding to values more on the outside of the axis in the radar. Values closer to the center are less preferred in terms of performance. The radar shows that for home pick-up, the values are often at the outside points of the radii compared to the other return methods. The green line from the collection point returns is mainly found between the different methods' values. The in-store returns scores are often the lowest compared to the other methods on the indicators but also show some outliers indicating high performance. An outlier for in-store returns is visible for indicator  $V_6$ , transport costs. The in-store return method performs thus well on this indicator compared to the other return methods. An explanation for this is that no costs are accounted to the company when the customer makes the trip to return the item to the store. However, there still are transport costs created with this return method. The data shows that multiple products returned in-store are transported to the warehouse in Spain and not resold in-store. This partly explains why this alternative has relatively high transport emissions  $(V_2)$ . The transport emissions for in-store returns include, besides the optional trip from the store to the warehouse, the emissions created by the customer to the store and back home are accounted for in this metric. This creates relatively many transport emissions per return. Home pick-up performs much better on the transport emissions per return due to the more efficient transport networks of 3PL, where shipments are consolidated. However, this return method is more costly for the company. This is because the transport costs are determined using the distance-based method, where the costs per kilometer with distribution vans are slightly higher than with modes for long-haul distances. The home pick-up alternative has more kilometers per return driven with vans than the other return methods. The warehouse and equipment costs for a home pick-up are also less costly for a return. An explanation is that relatively many products collected with home pick-up are resold in-store, causing the highest brick-and-mortar costs and emissions in  $V_{10}$  and  $V_5$ , respectively, relative to the other methods.

Looking at the labor costs  $V_8$ ; the in-store method performs worst than the other methods. An explanation for this is that the products are first checked in the brick-and-mortar store, and multiple products are transferred to the warehouse afterward, creating double labor. The labor wages for employees in the Netherlands are relatively higher but do not outweigh the costs of labor when a product is handled at a warehouse. Products dealt with at the warehouse require relatively much time and activities, increasing the labor cost per return. So, as most products from home pick-up are transported to the brick-and-mortar store, they create relatively lower labor costs. As seen, in-store returns score best on packaging emission  $V_1$ , which is expected as packaging is not needed for transporting the item to the brick-and-mortar store. Packaging material is needed for each return using online return methods (i.e., home pick-up and collection point), creating emissions. However, a return using the in-store return method also creates packaging emissions because the data shows that some products are transported from the brick-and-mortar store to the warehouse, still requiring packaging. Warehouse and equipment emissions are created when products have a final destination in the warehouse. For home pick-up, this value is best amongst the other methods as a larger share of these returns gets transported to the store, lowering the warehouse and equipment emissions on the product return level. The other methods have higher shares of products ending at the warehouses, leading to higher values for these warehouse and equipment emissions.



Figure 5.1: Radar chart of the different return methods in the base case, source: (Author)

With  $V_1$ : packaging emissions,  $V_2$ : transport emissions,  $V_3$ : warehouse and equipment emissions,  $V_4$ : product recovery emissions,  $V_5$ : brick-and-mortar emissions,  $V_6$ : transport costs,  $V_7$ : warehouse and equipment costs,  $V_8$ : labor costs,  $V_9$ : recovery costs,  $V_{10}$ : brick-and-mortar costs.

#### 5.5. Main findings chapter 5

This chapter has answered the question *'How can the current state of a company's performance be measured using the CI?'* To answer this question, the CI is developed using data from a fashion retailer, where the environmental and economic performance of the different return methods were analyzed. A CI score is obtained that represents a company's combined environmental and economic performance for each return method, where a higher score represents a better performance. A company can only consider the environmental or economic impact of a return method on its performance by only including the set of indicators of one dimension. An in-depth evaluation of how each return method scores on the indicators shows valuable insights into how- and why each method differs in performance. This showed a company's relevance in comparing scores with a reference, which is currently lacking in the CI.

# 6

### Design

This chapter designs a benchmark for companies to compare their performances with. It followed from the previous chapter that with the Composite Indicator (CI), the performance of a company can be measured, but insights lack on how good the companies' current performance is relative to a reference. The European Union (EU) sets sustainability objectives to combat climate change [59]. As companies need to comply with legislation and regulations of the EU, they need to know how far they are in reaching these sustainability targets. Therefore, this research intends to develop a benchmark which companies can use to evaluate their performances and give them crucial insights to improve their sustainability and comply with EU targets. The benchmark will be developed using the PESTLE framework, explained in section 6.1. Next, the benchmark design is presented in section 6.2, followed by its set-up. Following, the benchmark will be integrated into the CI and tested with the base case in section 6.3. It will be verified how- and whether the model can demonstrate its intended purpose, which is reflected in section 6.5. As a result, this chapter answers sub-question 4e) *How can the constructed CI be verified*?

#### 6.1. PESTLE analysis

The benchmark is designed using the PESTLE analysis (Political (P), Economic (E), Sociological (S), Technological (T), Legal (L), and Environmental (E)) [148, 198]. This analysis is an extended version of the original PEST (Political (P), Economic (E), Sociological (S), Technological(T)) analysis. The analysis helps identify and understand external environmental factors in a dynamic environment and identifies risks [198]. The analysis serves as a tool to ensure a complete overview of all these contextual factors is created [148, 109]. Scientific research and expert consultation identify the factors, trends, and shifts that drive developments. Examples are stricter environmental policies, more conscious shopping behavior, improved efficiency of equipment and modalities, and availability of recycling alternatives [145, 23, 119, 84].

#### 6.2. Design benchmark

This benchmark is defined according to specific targets for each indicator driven by Legal (L) and Policy (P) factors from the PESTLE criteria. European companies should comply with their performances with legislation and policy-making decisions, which therefore drive company improvements. The benchmark is set for the year 2030. This will represent the shorter-term goals of a company in its net-zero journey for 2050, which are more tangible and more accessible for companies to achieve as it is less than a decade away. Also, shorter-term goals allow companies more flexibility in their strategies for achieving them. Values for these standards are found through reports, official publications, and, optionally, expert consultation. These values should be defined in the same unit of measurement as the actual indicators. The standards for each indicator are applied to the base case data presented in section 5.1. So, for example, the original value of the indicator for transport emissions in the base case will be adjusted to conform to the reduction target of 55% from the base year to the target year. Next, it needs to be mentioned that making any accurate assumptions on the benchmark costs for 2030 is impossible, as this is highly surrounded by uncertainty and external factors, which are confirmed by experts from the industry. Many factors affect the costs (i.e., inflation, innovations, potential crisis, war), making the expenses not accurately estimated for the benchmark, but can be looked into in scenarios. For this reason, it is chosen to apply the base case costs for the economic benchmarks in this research. This assumption will be reflected in the discussion.

#### 6.2.1. Description of the benchmark

The benchmark is defined according to specific targets for each indicator driven by Legal (L) and Policy (P) factors from the PESTLE criteria. European companies should comply with their performances with legislation and policy-making decisions and drive company improvements. The European Commission (EC) defines no specific targets and legislation for packaging currently. E-commerce packaging should improve its environmental impact by reducing the amount of material used, using sustainable materials that are recyclable or reusable, and being responsible for waste management. The EC proposal on packaging and waste is not accepted [64] yet. The proposal's objectives are threefold: packaging should be made out of recycled materials, minimizing waste by 15% in 2040, and increasing the recycling and reuse rate of packaging. The packaging recycling rate for cardboard should be 85% and for plastic 55% in 2030. Further, landfill should decrease to 9.6% [64, 68]. The base case assumes that a combination of packaging materials like plastic Low-density polyethylene (LDPE) and recycled cardboard is used. Further, the waste treatments for recycling and landfill are defined according to the target values of 85%, 55%, and 9.6%, respectively. The remaining packaging material is combusted.

The target value for transport emission reduction by the EC is set at 55% by 2030 compared to the base year 1990, which was accepted in 2020. The EU in the 'Sustainable and Smart Mobility Strategy' report supports the feasibility of this benchmark value [70]. Assuming a linear reduction rate, this would mean a yearly Greenhouse Gas (GHG)-emission reduction of 1.57% from 1995-2030. The benchmark target in this research is set at this value for 2030 to comply with the EC target values.

The point of departure for the improved environmental impact of warehousing and in-store building emissions is using renewables. The EC has put a benchmark of 32% of renewable energy sources in buildings by 2030, which will be the target value for the benchmark in this research [61]. Electricity produced by renewables should power warehousing equipment as well in 2030.

Several strategies are proposed to lower waste emissions in the fashion sector, including developing ecodesign measures, guidance on waste management and circular businesses, and promoting waste treatments. The EC proposes Extended Producer Responsibility (EPR) for 2025 for textiles. With this, a closed-loop system is upgraded with recycling and reusing as preferred options; disposition to landfills should be reduced to a minimum as it generates the highest negative environmental impact according to the waste hierarchy [63, 68, 65]. The disposed of and not separated products are treated as municipal solid waste [111]. The EC has set targets to reduce the GHG emissions of municipal waste by 30% in 2030 [65]. Considering this, the benchmark value for the variable disposition emissions in this research is set to a limitation of incineration and landfill to 1% by 2030 and a reduction of overall disposal emissions by 30%.

#### 6.3. Benchmark testing

The benchmark development follows the same construction steps outlined in the previous chapter section 5.3 for the normalization and aggregation of the values. The weighting is done by Equal Weights (EW) instead of Analytical Hierarchy Process (AHP). It is decided that all indicators should have equal weights, as subjectivity is not preferred here. This also ensures that the preferences of companies cannot distort the benchmark values and the importance of all standards remains equal for all users of the CI. This will increase transparency on the benchmark values and is confirmed by industry experts. EW assigns equal weights to the indicators. As the total sum of the weights should equal 1, the individual weights of the 10 variables are 0.1.

The raw data is expressed in Table 6.1. The target reductions for the benchmark are applied to the data of the overall base case. For the return methods and the base case overall performance, the same data is used as presented in the previous chapter. In this table, the lowest values are marked red for an alternative on the indicator, and the green values are perceived as the best possible value from the set of alternatives on that indicator.

Next, normalization is done using the rescaling method proposed in Equation 4.1, shown in Table 6.2. All indicators are converted to a range between 1 and 2, where 2 represents the best value and 1 the lowest [204]. Weighting is applied to the indicators, and aggregation leads to Table 6.3. The CI scores should be interpreted the same as in the previous chapter, where a higher value of the CI represents better performance and a lower value indicates worse performance. As can be seen, the benchmark performs highest in terms of performance. The best performance among the alternatives is marked in green, and the lowest performance
of the alternatives is marked in red. The results show that in-store returns score lowest among the alternatives, and the home pick-up score best. The overall performance of the base case, which is the combined value of all three return methods, scores in between. As can be seen, the benchmark performs highest in terms of performance.

In Table 6.1 and Table 7.2:  $V_1$ : packaging emissions,  $V_2$ : transport emissions,  $V_3$ : warehouse and equipment emissions,  $V_4$ : product recovery emissions,  $V_5$ : brick-and-mortar emissions,  $V_6$ : transport costs,  $V_7$ : warehouse and equipment costs,  $V_8$ : labor costs,  $V_9$ : recovery costs,  $V_{10}$ : brick-and-mortar costs.

	$V_1$	V2	$V_3$	$V_4$	V5	V6	V7	$V_8$	$V_9$	V10
	[kgCO2e/	[kgCO2e/	[kgCO2e/	[kgCO2e/	[kgCO2e/	[euro/	[euro/	[euro/	[euro/	[euro/
	returned product]									
Benchmark	0.016	14.498	0.743	0.007	0.484	0.230	12.379	5.096	0.523	1.903
Base case overall performance	0.075	32.217	1.092	0.012	0.712	0.230	12.379	5.096	0.523	1.903
Return method 1. Home pick-up	0.077	23.721	0.624	0.005	0.784	0.432	7.076	4.721	0.280	2.093
Return method 2. In-store	0.013	42.567	1.559	0.016	0.642	0.073	17.666	5.470	0.755	1.715
Return method 3. Collection point	0.078	32.592	1.118	0.010	0.710	0.217	12.673	5.117	0.539	1.896

Table 6.1: Raw indicator values base case with the benchmark, source: (Author)

Table 6.2: Normalized values base case with the benchmark, source: (Author)

	$V_1$	$V_2$	$V_3$	$V_4$	$V_5$	$V_6$	$V_7$	$V_8$	$V_9$	V <sub>10</sub>
Benchmark	1.950	2.000	1.873	1.860	2.000	1.563	1.499	1.499	1.487	1.504
Base case overall performance	1.055	1.369	1.499	1.345	1.238	1.563	1.499	1.499	1.487	1.504
Return method 1. Home pick-up	1.024	1.671	2.000	2.000	1.000	1.000	2.000	2.000	2.000	1.000
Return method 2. In-store	2.000	1.000	1.000	1.000	1.473	2.000	1.000	1.000	1.000	2.000
Return method 3. Collection point	1.000	1.355	1.471	1.507	1.246	1.598	1.471	1.471	1.454	1.520

Table 6.3: Aggregated CI scores with benchmark, source: (Author)

	CI score
Benchmark	1.710
Base case overall performance	1.430
Return method 1. Home pick-up	1.568
Return method 2. In-store	1.209
Return method 3. Collection point	1.427

# 6.4. Conclusions performance measurement with the benchmark

The performance of the different return methods and the base case are compared relative to the benchmark, which is the target performance. Literature has shown that this can be done by determining the relative performance RP or the performance gap M, as found in section 3.5. Relative performance is the actual and target performance ratio, calculated with Equation 3.1. The performance gap represents the difference in performance between a company's actual and target performance, which is the benchmark and is calculated by Equation 3.3, using the data from Table 6.3. The best performance is achieved when the values of all indicators are minimized, so a lower value for the performance gap M means a better performance score [49]. As seen in Table 6.4, the performance gap with the benchmark is the smallest for the home pick-up return method and the largest with in-store returns. The relative performance expresses the same, which shows how each method scores in terms of its ratio with the benchmark.

More in-depth insights can be found by plotting the weighted indicator values in a radar chart, visualized in Figure 6.1. The indicators are defined on the spokes of the radar. For each indicator, better performance is obtained when the value is higher, corresponding to values more on the outside of the axis in the radar. Values closer to the center are less preferred in terms of performance. The blue line represents the benchmark values. The benchmark outperforms the alternatives on the environmental indicators, as the benchmark values are mainly on the outsides of the radii compared to the others. This is expected, as the benchmark expresses target values for environmental performances.

The overall base case performance has a performance gap of 0.280. Comparing the base case values with the benchmark in Figure 6.1, it can be seen that the benchmark outperforms the base case in almost all

	Relative performance RP [%]	Performance gap M
Benchmark	100%	0.000
Base case overall performance	84%	0.280
Return method 1. Home pick-up	92%	0.142
Return method 2. In-store	71%	0.501
Return method 3. Collection point	83%	0.283

Table 6.4: Relative performance and performance gap base case with the benchmark, source: (Author)

indicators. The largest gap, and thus performance difference, between the benchmark and base case is in the environmental indicators. This is in line with the expectations, as in the base case, not much has been done to improve the company's environmental performance. The gap between the base case and environmental indicators is smaller for the economic indicators, resulting from the different weights. The weights, especially the transport cost ( $V_6$ ) and warehouse and equipment costs ( $V_7$ ), for the base case, are higher compared to the weights assigned by EW. For the benchmark, it was chosen to apply EW to reduce subjectivity in the weights, improving the transparency of the benchmark. The weights assigned with AHP lead to higher weights, resulting in better performance on these indicators. However, this effect is outweighed by the aggregated performance of the benchmark, which shows a better overall performance than the base case.

The home pick-up return method outperforms the benchmark on  $V_2$  and  $V_3$ : transport and warehouse and equipment emissions. This is a logical result of the different weights that are used. For the economic indicators, the benchmark performs moderately, which is also not an unexpected outcome as EW is applied. Another outlier is the warehouse costs in  $V_7$ . All return methods perform better on this indicator relative to the base case. With AHP weighting, the experts have assigned relatively higher weights to this indicator than the lower weight assigned to this indicator by using EW in the benchmark. This is in line with the expectations.

The low CI score of in-store returns relative to the other alternatives is observed in the chart as well. The green line, representing the in-store returns, is mainly inside the radii, indicating lower performance. In-store returns score only better on  $V_6$  on performance compared to the benchmark. This is in line with the findings from the previous chapter. Here, it was found that in-store returns have relatively low transport costs relative to the other methods, and the benchmark uses the transport cost value of the overall base case performance.

The observations found align with the observations from the performance analysis in section 5.4. Therefore, this section only gives a concise evaluation of the difference, and underlying reasons, between the return methods. It is essential to observe that the ranking of the return methods has remained the same before- and after the benchmark implementation in the model. This confirms the reliability of the model's output. The change in scores of the CI compared to Table 5.6 is as expected as an alternative is added to the CI.



Figure 6.1: Radar chart base case with the benchmark, source: (Author)

# 6.5. Verification

Verification examines the question, 'Is this the right model?', in other words: how can it be proved that it is possible to find with this model what we want to understand? This question is addressed by thoroughly reading the model structure and its implementation to see whether this accurately represents the conceptual design and the expected outcome [7, 104]. Additionally, a framework is introduced as a tool to verify the reliability of the model outputs.

Before verification of the model, it is checked whether the conceptual model represents all essential elements to serve its purpose and requirements [181]. The conceptual model (see chapter 4) is iteratively built and continuously reviewed with multiple independent experts. These reviews served as inspection moments to verify the conceptual model's completeness and the assumptions made.

A structured walk-through verifies the model with a critical evaluation of the model's equations and formulation [104]. First, it is checked whether all variables are correctly implemented and formulated in the model. Each variable is isolated, and the calculation steps and unit of measurement are inspected. Particular consideration was given to checking the transportation emissions- and costs, as this consisted of multiple transport modalities with different characteristics that affect the calculation output. After each variable was checked, the different input values of the parameters were checked with their sources to ensure they were correctly implemented in the code. Next, the building of the CI is inspected. This is done for each construction phase separately. The selection of variables was already checked. Next, the normalization is inspected, which was done by using simple dummy data. The model calculations of normalization were compared to calculations done by hand, which verified this phase. Additional attention is paid to verifying the weighting method. The weighting method, AHP, is subjective and, therefore, has a higher risk of inconsistency [160]. To verify the weighting in this model, the group of experts assigned weights was between the recommended number of 3-7 experts. Also, clear instructions were provided for these experts to express the perspective from which they assigned the weights. The latter is particularly useful in understanding and preventing potential bias from the perspective from which experts assign weights. Besides, a consistency check is carried out that examines the reasonability of the weight scores and ensures that these represent an acceptable level of consistency [167]. This consistency check contributes to the verification of the model and its accuracy. Lastly, the aggregation of the model was checked by first checking the equations; next, the same dummy data was used to compare by-hand calculation with the models' output.

Last, the reliability of the model's output was verified [104]. Therefore, the PESTLE framework is introduced as verification to ensure the model can answer what we want to know: to gain insight into the company's performance relative to a benchmark. To ensure that the conceptual description of the benchmark is accurate and is developed by independent criteria, the PESTLE (Political (P), Economic (E), Sociological (S), Technological (T), Legal (L), and Environmental (E)) analysis is used [148, 198]. PESTLE is an internal verification tool for the benchmark design used to help to verify the model. The benchmark's description, criteria, and set-up were verified using expert consultation. Special care was given to the weighting method; it was decided in cooperation with industry experts to assign weights according to equal weighting to increase transparency and reliability of the weights when comparing the benchmark sector-wide (EW). Next, it was checked if the benchmark was correctly implemented in the model. This is done by inspecting the model's output by testing and evaluating the results generated after implementation of the benchmark on the base case performance of the CI. Performance analysis tools were identified from the literature, which was used to verify these outputs and gather insights into the needed information. The analysis showed plausible results, which were in line with the expectations and set-up of the benchmark. The previous chapter has already created data on the company's measurement, which was used to verify the benchmark models' output. This benchmark testing did additional verification of the model's output, as it was found that the ranking of the different alternatives remained the same before and after the implementation of the benchmark. So, the model's output with the benchmark generates reliable results and is successfully verified.

# 6.6. Main findings chapter 6

In this chapter '*How can the constructed CI be verified*?' was answered. This is done through a structured walk-through to inspect the correctness of the models' structure, implementation, and formulation and see whether the model represents the conceptual design and expected outcomes [104]. First, the variables were inspected and checked on their calculation and dimensions. Second, the equations and formulation of each phase of the CI was reviewed. Thirdly, the reliability of the model's output and set-up of the benchmark was inspected. The benchmark's set-up was verified using the verification framework PESTLE and through expert

consultation. Next, it was checked whether the benchmark was implemented correctly to verify the reliability of the model's output. The benchmark was tested using benchmark techniques from the literature, which showed plausible results aligned with the expectations, and no surprises were found. The model's output showed that the model could find what needs to be understood, and the CI can be successfully verified.

# 7

# Evaluate

In this chapter, scenarios are designed to test and validate the model. The previous chapter has provided a verified Composite Indicator (CI) and benchmark for fashion companies to compare their performance with. This research intends to identify areas of improvement for fashion companies in the process of product returns and gain knowledge of which return improvement strategy performs best regarding the environmental and economic impact. To validate if the model is able to identify improvement areas and understand the impact of external factors and the effects of adopting strategies to improve sustainability on the performance of a company, scenarios are designed. The previous chapter has introduced the PESTLE framework, which will be used to identify external environmental factors and to design the scenarios presented in section 7.1. In section 7.2, the scenarios are analyzed using the constructed CI. As part of this, an evaluation of the model takes place where the robustness of the model is tested. This gives insight into the model's sensitivity to changes in input data or the usage of different construction methods. Next, the performance analysis takes place where the performance of the different scenarios with the base case performance and the benchmark are compared in section 7.3. Lastly, the validation of the model is done in section 7.4. Thereby, SQ4f) *How can the constructed CI be used in practice by companies*? and SQ4g) *How can the constructed CI be validated*? are answered.

# 7.1. Design and set-up scenarios

Different scenarios will be designed that include strategies to improve the sustainability of the return process and the potential effects of external factors. The external factors are identified using the PESTLE analysis, as introduced in section 6.1. In total, 5 scenarios are designed based on the PESTLE framework and MECE principle. MECE is particularly useful for designing unique scenarios to ensure they do not overlap, according to an expert of Metyis. MECE stands for Mutually Exclusive, Collectively Exhaustive [201]. The base case, as introduced in section 5.1, will be used to design the scenarios. Assumptions made for the base case are tested, and data will be adapted and analyzed with the developments identified for each scenario. Each scenario has a moderate and an extreme variant to investigate potential outcomes and effects, which are based on expert consultation, reports, and literature studies. An overview of the different scenarios is given in Figure 7.1, which are elaborated on in more detail in this section.

## Scenario 1. Shifting customer behavior

This scenario entails the PESTLE analysis's Social (S) and Environmental (E) criteria. A visible market trend is customers' increased value and awareness of the environmental impact of their fashion consumption [131]. Changed customer behavior steers this scenario as customers focus more on sustainability practices. Customer behavior is a crucial component for lowering Greenhouse Gas (GHG) emissions in the fashion industry [194, 37, 12]. In this scenario, customers are both increasingly aware of the environmental impact of returns and want to improve their sustainability footprint. On the one hand, this leads to customers showing more conscious return behavior, researching better before buying a product, limiting opportunistic returning behavior, and lowering over-ordering due to environmental reasons. This can lead to lower product returns in general [194, 22]. If customers decide to return, they are more likely to choose the return alternative with a minor impact on the environment and reuse their package box, lowering the GHG emissions [191]. Accord-



Figure 7.1: Overview of scenario design, source: (Author)

ing to an expert, retailers can recycle their packaging material better when the original packaging box is used. This can therefore lead to a higher recycling rate of packaging waste.

On the other hand, companies respond to this sustainable customer demand by improving transparency on the impact of the different return methods, which is lacking in the base case. Transparency may lead to lower returns as customers can make more conscious decisions [194, 22]. Further, companies invest in better product descriptions, online customer experiences, and assistance to lower purchase uncertainty in reaction to this conscious customer behavior. Fitting and uncertainty issues contribute significantly (i.e., 42%) to reasons for returns, indirectly affecting the environment negatively [95, 126]. Return reasons are also dependent on other factors; these are not incorporated in this scenario. Lastly, companies discourage opportunistic customer behavior by charging a return fee of 1 euro in the moderate variant and 2 euros in the extreme variant of this scenario. This fee is subtracted from the customer's refund after the returned product is received by the company. The company uses this fee to cover the warehouse's return handling and processing costs.

In the base case scenario, transparency on the environmental impact of product returns to customers is lacking, so customers are unaware of which return alternative would perform best in terms of environmental impact. Besides, there is still much potential to reduce the environmental impact due to improved conscious customer behavior [12]. The base case data shows that customers show a lot of opportunistic return behavior; sometimes, more than five products are returned in the same order. Therefore, the share of customers showing conscious return behavior is assumed to be very low in the base case scenario. Considering this, the moderate variant of this scenario assumes that 5% fewer products are returned due to more conscious customer behavior. The extreme variant assumes a more intense effort of customers and companies by incorporating more sustainability practices. The extreme circumstances take that 15% fewer products are returned due to more conscious customer behavior.

The moderate variant of this scenario supposes that 40% of the customers use the original packaging for returns, and the extreme presumes 60% of the packaging reusing. In the base case, it was assumed that the company recycled no packaging. Besides consultation with experts, the study of [191, 32], has shown that only 20% of the returned products are sent back in their original package. As larger amounts of the same packaging material are returned, companies are better able to recycle packaging, assuming 20% in the moderate variant and 40% in the extreme variant. These values are considered to be realistic, according to an industry expert.

The customers will consider the return method's environmental footprint for their selection, of which insights are lacking in the base case. The moderate variant assumes that 40% of the customers choose their return method based on ecological considerations, and the extreme scenario variant is 80%. The base case shows that the least preferred method from both environmental considerations is in-store returns, followed by the collection point (see subsection 5.3.4). This might differ depending on the data.

#### Scenario 2. Green supply chain management

Fashion companies focus more on integrating environmental practices into their supply chain [131]. This focus is driven by the strong emphasis of company shareholders on improving environmental performance. Therefore, this scenario's environmental (E) principle is a driving factor. Companies aim to improve the environmental impact of their whole supply chain, including transport, warehousing, and post-usage of products, so that environmental goals can be achieved and a company's brand image can be improved [209, 131]. A market trend that aligns with green supply chain management is greentailing. Greentailing, or eco-tailing,

refers to a business strategy emphasizing socially responsible, environmentally friendly, and financially successful retailing across all commercial endeavors. Greentailing calls for retailers to increase the environmental performance of their supply chains and business operations while maintaining profitability [131, 137].

Transport providers are selected based on environmental performance. A shift from road transport towards more environmentally friendly fuels or modalities is visible for long-haul trips [108]. The short-haul trip, including the first and last-mile delivery, can be increasingly done with electric vehicles [186, 47, 151]. Additionally, governmental policies will help companies implement these renewable energies through financial support, making it broader available [61]. Companies will charge customers a fee for their returns to compensate for their investments.

Two variants of this scenario will be examined that incorporate green supply chain management; one moderate and one more extreme. A realistic shift for this scenario is consulted by an expert, considering the current base case situation. The moderate variant will ensure that companies' short-haul operations are 40% by electric vans, for 25% of the long-haul road transport, Liquid Natural Gas (LNG) is used instead of diesel [41], and that renewable energy sources supply 25% of the warehouse and buildings energy. Companies focus on changing the packaging type to improve their environmental impact by using recycled materials [191, 54]. In the moderate variant, 25% of the packaging material is made from recycled materials, which is 50% in this scenario's extreme variant. Corrugated carton boxes are, on average, half made of recycled materials [105] and are environmentally friendly alternatives to cardboard packaging [17]. 100% Low-density polyethylene (LDPE) is used for environmentally friendly plastic alternatives [94]. The more extreme variant of this scenario uses electric vehicles for transport for 80% of their short-haul operations, assumes a 50-50 share of diesel and LNG fuels for the long-haul transport, and supplies 50% of the warehouse and building energy by renewables. The transport costs for electric and LNG-fueled vehicles are derived from a study of [41] from 2020 that compared the price per kilometer for operating different fueled trucks. It is assumed that a company's investment in renewable energy for buildings leads to a 25% higher price in the moderate variant and 40% in the extreme due to economies of scale. A return fee of 2 euros and 4 euros is charged in the moderate and extreme variants, respectively, to compensate for these investments in renewable energy in the warehouse costs.

## Scenario 3. Economic attractiveness of renewables

The fashion industry is primarily profit-driven, and economic costs are a primary factor in decision-making. In this scenario, economic developments are driving. A combination of inflation and rising prices due to shortages in fossil fuels has led to increased oil, coal, and natural gas prices in the market. Parallel to this development, governments aim to be less dependent on fossil fuels and invest more in renewable energy sources like solar, biomass, or wind power. Governments promote the adoption of renewable energy and offer, for example, a lower tax on renewable energy than fossil fuels [140]. Combining these factors will increase demand for renewable energy as it becomes a more cost-attractive alternative. As a result, Third Party Logistics Provider (3PL) can invest increasingly in greener vehicles in their fleet, leading to a broader offer of green modalities for a more competitive price. Besides, companies are starting to invest in renewables to power their stores and warehouses as it shows gains for both environmental and economic goals. Based on consultation with an industry expert in finance, developments in this scenario are formulated. A price increase of 5% is considered in the moderate variant of this scenario, together with an increase of 10% in renewables. The extreme variant assumes an overall increase of 10% of prices due to inflation and shortages, but renewables power 25% due to tax support on renewables. As fleets are increasingly becoming greener, the moderate variant assumes 10% of the kilometers are driven with electric vehicles, and in the extreme variant, 25%.

#### Scenario 4. Governmental support to a circular business model

In this scenario, governmental policies force companies to prioritize circularity in their business models. Circularity is a way to create value by lowering the GHG emissions created by waste processes in the operations [177]. An example of a governmental policy is the Extended Producer Regulation, where companies are held accountable for the waste generated by their products. There are different directions where fashion companies can include circularity in their operations.

Companies aim to lower harmful waste treatments and increase packaging recycling rates. An industry expert emphasizes the low quality of the packaging from returned products and, therefore, the lowered potential for recycling. Companies use stronger packaging materials to increase the recycling potential [131, 17]. This will lead to a recycling rate of 30% of the packaging material in the moderate variant and 60% in the extreme variant. Besides, companies focus on providing packaging made of biodegradable or recycled materials

[131]. The government has introduced European Union (EU)-Ecolabels to increase circularity and promote the environmental performance of products and consumption. An EU-Ecolabel can be granted if a product, service, or manufacturer meets specific scientifically developed and transparent criteria. These criteria are based on, for example, the created environmental impact, the energy efficiency, or the product's recyclability [57]. EU-Ecolabel for packaging requires a percentage of recycled material for at least 25% [80].

In this scenario, corrugated cartons are increasingly used to substitute cardboard, as these materials contain recycled content and are easier to recycle (50% in the moderate variant, 85% in the extreme) [54, 17]. The circularity of plastic bags can be improved when the material is made of 100% LDPE [94]. Governmental incentives in the form of financial subsidies increase the adoption of solar energy power and, thereby circularity of the warehouses and buildings. This leads to an adoption of 15% renewable energy for the warehouse and buildings in the moderate variant and 30% in the extreme variant of this scenario.

Recycling techniques are still under development, but companies donate a larger share of clothes that cannot be resold as new anymore to improve circularity. In the base case, donating clothes is not a common practice. An increased share of donated products will lead to fewer items being sent to landfills or incineration. Governmental policies will financially discourage landfill or disposal of clothes and increase charges by 20% per tonne textile in the moderate variant and 40% in the extreme variant.

The scenarios variants developed based on this increased demand for a circular business model are moderate and extreme, which are based on literature and consultation with several experts within multiple industries (finance, fashion, warehousing). In the moderate variant, 50% of the packaging material will be recycled, and 80% in the extreme variant. In the moderate scenario, 40% of the packaging is made of recycled materials; in the extreme scenario, this is 80%. The recycling techniques for clothing are still under development, so the moderate variant of this scenario will assume a reusing rate of 90% and the extreme variant of 95%. Under the reusing of clothes are also the clothes that will be donated. The share of donated items is nearly zero in the base case situation. The share of items sent to landfill and incineration is reduced to 5% each, respectively, in the moderate variant and 2.5% in the extreme. It is assumed that there is much financial and governmental support to increase circularity, so the investments made in the moderate variant lead to a price increase of 5%. In the extreme variant, this will be 7.5%.

#### Scenario 5. Technological innovations

Digitalization is a market trend that continues to grow and drives this scenario together with technological innovations (T) [25]. In a digital landscape, customers increasingly shop online and do not often use in-store alternatives, leading to higher e-commerce volumes. Companies adopt more innovative technologies in their business models and their operations. Specific technologies aim to improve the digital experience and presentation of products to customers and lower uncertainty issues when purchasing online [115, 117]. This will lead to a higher likelihood of ordering the correct product directly, reducing the need to return items for these reasons. Further, technologies will help optimize logistics and transport processes, reducing travel time and distances and adopting more innovative and efficient transport [86]. Additional development of electric and hybrid vehicles will increase the potential for adopting green transport on long-haul trips. Governments support shifting towards more renewable energies by providing adjusted infrastructure and financial incentives. Additionally, technological innovations make it possible to separate the waste streams and recycle more products. Developing mechanical and chemical sorting techniques will recover value from the end of a product's life [131, 111, 52]. In the current situation, it is assumed that no products are recycled, partly because of the low recycling rates and the uncertainty on the emissions- and cost of the recycling technologies. This scenario explores the recycling alternative with mechanical-based closed-loop recycling, which is based on fiber recycling [52]. From expert consultation, it was found that this alternative is the least cost-intensive and is likelier to be used.

Two variants of this scenario are developed. The first is the moderate variant, where the number of returns handled in-store is reduced by 50% compared to the base case. In this scenario, no consideration is given to the best environmental method based on the choice of the return method. The original in-store returns are randomly redistributed to home pick-up or collection point returns. The extreme variant of this scenario reduces in-store returns by 80% compared to the base case scenario. The transport innovations make transport more efficient, leading to an overall reduction of 5% emissions in the moderate variant of this scenario and 10% in the extreme variant. Misfit and sizing issues are the main reasons a product is returned (i.e., almost 50 percent according to [126]). Even with better-fitting technologies, the probability of returning due to these factors will still not equal zero. In the moderate variant, the total number of products returned is lowered by 10%. The extreme variant assumes that much uncertainty is taken away, and customers are likely to purchase

the correct product directly, reducing the number of product returns by 20%. In the base case, recycling is not considered an alternative for waste disposal. Adopting recycling technologies will increase the share of recycled products to 5% in the moderate scenario and 15% in the extreme scenario. Further adoption of recycling technologies also requires a fundamental change in the materials of products, which is not assumed to be a feasible scenario for this research. 80% of the returned items are resold through reuse; the remaining share is equally divided by landfill and incineration.

# 7.2. Scenario testing

In this section, the designed benchmark and scenarios are measured, analyzed, and evaluated on the environmental and economic performance. The same phases are followed as in section 5.3 for the design of the CI.

## 7.2.1. Phase 1. Selection of the indicators

For each variable in the theoretical framework, the data is measured for the variants of the scenarios. This is presented in Table 7.1. Lower values are better for each indicator, and higher are less preferred. The best- and worst values over all the alternatives are highlighted in green and red, respectively, in the table.

Here:  $V_1$ : packaging emissions,  $V_2$ : transport emissions,  $V_3$ : warehouse and equipment emissions,  $V_4$ : product recovery emissions,  $V_5$ : brick-and-mortar emissions,  $V_6$ : transport costs,  $V_7$ : warehouse and equipment costs,  $V_8$ : labor costs,  $V_9$ : recovery costs,  $V_{10}$ : brick-and-mortar costs. An overview of the scenarios and their moderate and extreme variants is given in Figure 7.1.

 $V_7$ V<sub>10</sub> [euro/  $V_8$  $V_9$ V2 [kgCO2e V<sub>3</sub> [kgCO2e/ V5 [kgCO2e/ [kgCO2e/ [kgCO2e [euro [euro/ [euro [euro/ ned product ned produ ed product returned product] ned produc ned product] retu ned product ned product returned product ned product] 0.016 0.743 0.484 0.230 12.379 5.096 0.523 Benchmark Base case overall perfo. Scenario 1. Moderate Scenario 1. Extreme Scenario 2. Moderate 32 21 1.09 0.712 0.230 12 37 5.09 0.523 0.067 32.355 31.509 30.653 1.092 1.083 1.079 0.840 0.011 0.011 0.010 0.523
0.489
0.501
0.517 0.229 11.280 5.089 1.919 0.719 1.92 5.101 0.268 Scenario 2. Extreme Scenario 3. Moderate 0.046 29.089 30.237 0.011 13.298 12.998 0.307 5.096 5.356 2.664 1.998 0.996 0.549 0.641 0.077 Scenario 3. Extreme Scenario 4. Moderate 0.074 0.840 0.534 0.289 13.617 27.269 32.217 0.011 0.526 2.093 0.028 0.945 0.010 0.609 0.230 13.062 5.101 2.008 Scenario 4. Extreme Scenario 5. Moderate 32.217 0.793 0.501 0.230 5.101 5.116 0.076 12.659 0.518 1.904 Scenario 5. Extreme 29.097 1.081 0.009 0.231 1.920

Table 7.1: Raw indicator values for each scenario, source: (Author)

# 7.2.2. Phase 2. Normalization of the indicators

Next, the raw values are normalized using rescaling. The formulation of Equation 4.1 is adopted, as instead of return methods, the different scenarios are used as alternatives. Table 7.2 shows that the values of all variables are rescaled between 1 and 2. Here, a 2 represents the best value in terms of performance and a 1 the lowest. This transforms the variables' values to a standardized scale, making them comparable independent of their unit of measurement.

Here,  $V_1$ : packaging emissions,  $V_2$ : transport emissions,  $V_3$ : warehouse and equipment emissions,  $V_4$ : product recovery emissions,  $V_5$ : brick-and-mortar emissions,  $V_6$ : transport costs,  $V_7$ : warehouse and equipment costs,  $V_8$ : labor costs,  $V_9$ : recovery costs,  $V_{10}$ : brick-and-mortar costs. An overview of the scenarios and their moderate and extreme variants is given in Figure 7.1.

## 7.2.3. Phase 3. Weighting of the indicators

The variables are weighted using the weights obtained in the previous chapter in Table 5.5. For the benchmark, Analytical Hierarchy Process (AHP) weighting is adopted with Equal Weights (EW) (see section 6.2).

## 7.2.4. Phase 4. Aggregation of the indicators

Next, the CI scores are obtained by aggregation with multiplication, presented in Table 7.3. The aggregated values of the CI should be interpreted as follows: the highest score represents the best performance and the lowest is the worst. The results show that the benchmark, defining the target performance, scores best compared to all alternatives in terms of performance. Looking at the scenarios, the extreme variant of scenario

	$V_1$	$V_2$	$V_3$	$V_4$	$V_5$	$V_6$	$V_7$	V8	$V_9$	V10
Benchmark	1.994	2.000	1.706	2.000	1.824	1.963	1.440	1.979	1.706	2.000
Base case overall performance	1.044	1.014	1.047	1.000	1.510	1.963	1.440	1.979	1.706	2.000
Scenario 1. Moderate	1.172	1.006	1.063	1.240	1.000	1.973	1.726	1.993	1.929	1.979
Scenario 1. Extreme	1.449	1.053	1.072	1.408	1.501	2.000	2.000	2.000	1.882	1.976
Scenario 2. Moderate	1.231	1.101	1.523	1.241	1.755	1.484	1.000	1.970	1.756	1.250
Scenario 2. Extreme	1.508	1.188	2.000	1.278	2.000	1.000	1.200	1.979	2.000	1.000
Scenario 3. Moderate	1.014	1.124	1.228	1.213	1.608	1.645	1.279	1.480	1.510	1.875
Scenario 3. Extreme	1.060	1.289	1.523	1.212	1.755	1.221	1.117	1.000	1.687	1.750
Scenario 4. Moderate	1.801	1.014	1.323	1.415	1.653	1.963	1.262	1.970	1.309	1.861
Scenario 4. Extreme	2.000	1.014	1.611	1.477	1.800	1.963	1.182	1.970	1.000	1.799
Scenario 5. Moderate	1.021	1.000	1.000	1.379	1.509	1.922	1.367	1.941	1.745	1.998
Scenario 5. Extreme	1.000	1.188	1.067	1.510	1.501	1.948	1.472	1.996	1.987	1.978

Table 7.2: Rescaled indicator values for each scenario, source: (Author)

1, shifted customer behavior, comes closest to this benchmark performance with a CI score of 1.589. The extreme variant of the economic competitiveness of renewables scores the lowest, indicating the worst relative performance score, and is marked red.

	CI score
Benchmark	1.852
Base case	1.399
Scenario 1. Moderate shifting customer behavior	1.447
Scenario 1. Extreme shifting customer behavior	1.589
Scenario 2. Moderate green supply chain management	1.322
Scenario 2. Extreme green supply chain management	1.388
Scenario 3. Moderate economic attractiveness of renewables	1.352
Scenario 3. Extreme economic attractiveness of renewables	1.294
Scenario 4. Moderate governmental support to a circular business model	1.437
Scenario 4. Extreme governmental support to a circular business model	1.439
Scenario 5. Moderate technological innovations	1.395
Scenario 5. Extreme technological innovations	1.491

### 7.2.5. Phase 5. Post-analysis evaluation of the CI

The post-analysis evaluation provides an understanding of the robustness and reliability of the constructed CI. Several uncertainty triggers were identified during the construction of the model ( $X_1$  weighting method,  $X_2$  aggregation method,  $X_3$  input data), see subsection 4.3.5. Insight must be obtained if- and to what extent different construction methods- or data would lead to different final values of the CI. First, the uncertainty sources  $X_1$  and  $X_2$  are analyzed, and their impact is discussed. Second, a statistical test is used to inspect the correlation of the uncertainty triggers. The uncertainty source  $X_3$ , changes in input data, is inspected with a Sensitivity Analysis (SA) by increasing and decreasing the values by 25%. The Pearson correlation coefficient is used to determine the statistical relation of this effect [147].

#### Uncertainty trigger $X_1$ . Weights

The weights are rescaled to examine the impact of the first uncertainty trigger  $X_1$ . Three new experts are requested to assign weights from the perspective of a CEO, and this effect is analyzed on the final value of the CI. The pairwise-comparison matrix is constructed, resulting in the weights presented in Table 7.4. These weights are statistically checked to ensure sufficient consistency (Consistency Ratio (CR) < 0.10) [142]. The pairwise comparison matrix *A* can be found in section H.1.

With  $V_1$ : packaging emissions,  $V_2$ : transport emissions,  $V_3$ : warehouse and equipment emissions,  $V_4$ : product recovery emissions,  $V_5$ : brick-and-mortar emissions,  $V_6$ : transport costs,  $V_7$ : warehouse and equipment costs,  $V_8$ : labor costs,  $V_9$ : recovery costs,  $V_{10}$ : brick-and-mortar costs.

Table 7.4: Rescaled weights with AHP, source: (Author)

	Weight
$V_1$	0.031
$V_2$	0.167
$V_3$	0.054
$V_4$	0.069
$V_5$	0.032
$V_6$	0.231
$V_7$	0.128
$V_8$	0.118
$V_9$	0.106
$V_{10}$	0.082

Next, the normalized data from Table 7.2 was weighted using the rescaled weights and aggregated using multiplication. The scores of the CI are shown in Table 7.5. The scores should be interpreted as in Table 7.3, where higher scores correspond to better performance and lower scores to a less desired performance. As can be seen, the extreme variant of scenario 1 performs best, and the extreme variant of scenario three the worst.

Table 7.5: Aggregated CI scores obtained after rescaling weights, source: (Author)

	CI score
Benchmark	1.852
Base case	1.497
Scenario 1. Moderate shifting customer behavior	1.562
Scenario 1. Extreme shifting customer behavior	1.652
Scenario 2. Moderate green supply chain management	1.377
Scenario 2. Extreme green supply chain management	1.343
Scenario 3. Moderate economic attractiveness of renewables	1.403
Scenario 3. Extreme economic attractiveness of renewables	1.292
Scenario 4. Moderate governmental support to a circular business model	1.505
Scenario 4. Extreme governmental support to a circular business model	1.475
Scenario 5. Moderate technological innovations	1.506
Scenario 5. Extreme technological innovations	1.609

The values obtained with the rescaled weights are compared to the original values of Table 7.3. This is visualized in Figure 7.2, where the blue values indicate the original values of the CI and the green values the CI scores obtained by rescaling the weights. The height of each column represents the score of the CI, where a higher value represents better performance. The benchmark value remains unchanged; in both cases, EW is used. Both cases show that the same scenario variants perform best- and worst (S1E and S3E, respectively). Further, the overall performance scores increase after using the rescaled weights observed compared to the original weights. With the benchmark value remaining the same, all scenarios have a smaller performance gap by the benchmark. An explanation for this increase is that the rescaled weights of Table 7.4 indicate higher importance of the decision makers to economic indicators relative to the environmental indicators. This means that the economic performance will be slightly more influential in the overall performance. It is expected that scenarios that perform relatively better on the economic indicators than the environmental indicators show the highest increase relative to the original weights in overall performance. This hypothesis is confirmed when looking at the raw data values of Table 7.1 and comparing these with the outcomes in Figure 7.2.

Further, it is essential to see if the ranking of the scenarios has been changed to see the effect of rescaling. Most ranks have remained equal. However, S4M and S5M have switched positions; looking at the raw data, it is discovered that S5M performs a little better on the economic indicators, which are assigned higher weights now than SM4. Also, S2E has lowered by two positions ending at the second last, and S3M and S2M are increased by 1. This is a result of the relative importance of the decision-maker to the different indicators. As seen in Table 7.1, S2E performs relatively worse on the economic dimensions. As the economic indicators are perceived as more important than the environmental indicators after rescaling, it is a logical consequence



Figure 7.2: CI scores obtained with original and rescaled weights, source: (Author)

that this variant performs worse compared to the previously assigned weights. The results show thus no surprises.

#### Uncertainty trigger $X_2$ . Aggregation method

The second uncertainty trigger is the aggregation method  $X_2$ , which is analyzed using the additive method (see Equation 4.9). This aggregation method is applied to the normalized data presented in Table 7.2 and weighted using AHP with the original weights from Table 5.5. For the benchmark, EW is used. The scores of the CI are shown in Table 7.6. Better performance is represented by higher values for the CI score and vice versa. The same variants score best- and worst compared to the initial results shown in Table 7.3.

	CI score
Benchmark	1.861
Base case	1.442
Scenario 1. Moderate shifting customer behavior	1.500
Scenario 1. Extreme shifting customer behavior	1.634
Scenario 2. Moderate green supply chain management	1.340
Scenario 2. Extreme green supply chain management	1.409
Scenario 3. Moderate economic attractiveness of renewables	1.364
Scenario 3. Extreme economic attractiveness of renewables	1.304
Scenario 4. Moderate governmental support to a circular business model	1.469
Scenario 4. Extreme governmental support to a circular business model	1.485
Scenario 5. Moderate technological innovations	1.433
Scenario 5. Extreme technological innovations	1.522

Table 7.6: Aggregated CI scores using additive aggregation, source: (Author)

As seen in Figure 7.3, the results after linear additive aggregation show a little increase in all the final scores of the CI compared to the original aggregation method, which was multiplication. As all scenarios are increased slightly, the ranking of all scenarios remains unchanged. However, the performance gaps with the benchmark for each scenario have improved slightly as the increase in CI score with the additive aggregation method is lower compared to the scenarios. This is the result of EW applied to the benchmark.

Testing the correlation of the uncertainty triggers  $X_1$  and  $X_2$ 

The correlation between the methods is investigated with Equation 4.11, where r ranges from -1 to 1, and values between 0.5 and 1 represent a strong and positive relationship [20]. So, from Table 7.7, it can be concluded that both uncertainty triggers  $X_1$  and  $X_2$  positively correlate with the original method of the CI, which



Figure 7.3: Overview of difference between the CI scores using GME and additive aggregation method, source: (Author)

has also proven significant at 0.05 and 0.01 levels. Thus, the null hypothesis  $H_0$ , stating that no correlation exists between the chosen and alternative construction methods, can be rejected at both levels for both uncertainty triggers [20, 203, 99]. So, the different construction techniques produce similar results.

Table 7.7: Outcomes correlation coefficient for uncertainty trigger X1 and X2, source: (Author)

	Correlation coefficient r	t-statistics	Significant
X <sub>1</sub> . Alternative weighting	0.941**	0.101	yes
X <sub>2</sub> . Aggregation method	0.994**	0.034	yes
* alpha < 0.05 (2-tailed)			
** alpha < 0.01 (2-tailed)			

#### Uncertainty trigger $X_3$ . Input data

The third source of uncertainty examined is the input data  $X_3$ . Following from section 5.2, several data points are worth investigating based on the assumptions made in the base case and scenarios. The following uncertainty sources are selected based on the assumptions made and their expected effect on the model's output: equipment cost, transport costs of the electric distribution van, emission value for the warehouse, landfill costs, and incineration cost. The contribution of the uncertainty of the data values on the final value of the CI is examined by performing a SA [134]. Each input data value increases and decreases by 25%, while the rest remains unchanged. In Appendix H.2, the results on the scores of the CI are presented for each scenario together with the data after changing the variable.

The effect and contribution of the input data on the final CI is analyzed with the correlation coefficient. The results are presented in Table 7.8. From these outcomes, it can be concluded that there is a strong positive and significant relationship. All investigated inputs have an r close to 1, and p-values statistically significant at 0.05 and 0.01 levels. This indicates limited sensitivity to changes in the examined input data between the pairs of values, and  $H_0$  can be rejected at 0.01 and 0.05 [20, 99].

# 7.3. Conclusion performance measurement scenario testing

This section evaluates the performance of the different scenario designs, which were presented in Table 7.3. The performance gap M will be determined to understand better the actual performance of the scenarios relative to the target performance. The performance gaps are calculated by Equation 3.3, using the data from Table 7.3. The performance gap represents the difference in performance between a scenario's actual and target values, which is the benchmark. The best performance is achieved when the values of all indicators are minimized, so a lower value for the performance gap M means a better performance score [49]. The results of the performance gaps with the benchmark are visualized in Figure 7.4. In Appendix G, the relative performance and the performance gap can be found for each scenario.

The base case and the scenarios are plotted on the horizontal axis in Figure 7.4. The height of the column

	Correlation coefficient r	t-statistic	Significant
Equipment cost (+25%)	0.992**	0.039	yes
Equipment cost (-25%)	0.981**	0.059	yes
Electric van transport cost (+25%)	0.988**	0.049	yes
Electric van transport cost (-25%)	0.976**	0.067	yes
Warehouse emission value (+25%)	0.991**	0.044	yes
Warehouse emission value (-25%)	0.993**	0.036	yes
Incineration cost (+25%)	0.961**	0.030	yes
Incineration cost (-25%)	0.992**	0.039	yes
Landfill cost (+25%)	0.990**	0.042	yes
Landfill cost (-25%)	0.992**	0.040	yes

Table 7.8: Outcomes correlation coefficient for uncertainty trigger  $X_3$ , source: (Author)

\* alpha < 0.05 (2-tailed)

\*\* alpha < 0.05 (2-tailed)



Figure 7.4: Performance gaps scenarios relative to the benchmark, source: (Author)

represents the performance gap relative to the benchmark. The largest performance gap is found in the extreme variant of scenario 3; the economic attractiveness of renewables. The extreme variant of scenario 1 has the smallest performance gap with the benchmark relative to the other scenarios, indicating better performance. In Appendix I, radar charts are presented for each scenario with the benchmark and base case. This gives an in-depth understanding of the developments in the scenarios. The performance of the base case with the benchmark was already analyzed in section 6.4, and the radar chart is presented in Figure I.1.

Both variants of scenario 1 have a smaller performance gap than the base case with the benchmark, meaning better performance. Although recycling is more costly than landfill or incineration alternatives for packaging, the overall performance is improved in both variants of this scenario compared to the base case. The higher costs of packaging are compensated for by better environmental performance and the usage of financial incentives by companies to discourage returns, which also helps lower the return cost for a company. The return fee is used to lower the warehouse and equipment cost, leading to a relatively good performance on this indicator in both cases of this scenario, as seen in Figure I.2. The extreme variant of this scenario shows a relatively sizeable improved performance compared to the moderate variant and the base case, partly due to the higher return fee to discourage customers from returning and better environmental performance. Also, customers consciously choose their return method based on environmental considerations, lowering the number of returns in-store. This leads to an improvement in the brick-and-mortar emissions and transport emissions accordingly in both variants of this scenario compared to the base case.

Both variants of scenario 2 have worse performance than the base case relative to the benchmark. Both variants show high costs, as seen in Table 7.1. So, the increased environmental improvement efforts in both scenarios do not outweigh the costs, leading to a bigger performance gap with the benchmark. Figure I.3 shows that modality shift, more environmentally friendly packaging material, and investing in renewable energy sources for buildings lead to better environmental performances and, in the extreme variant, may even

outperform the benchmark. Although the company charges a return fee in both variants, no governmental financial support is given to lower the costs. Therefore, these economic costs are not compensated by environmental gains. In the extreme variant, the efforts are more compatible with the costs made, improving the overall performance compared to the moderate variant of this scenario. However, the overall performance is not improved compared to the base case, as the costs in the extreme variant are highest for both transportant brick-and-mortar stores, which were as expected as large investments were made.

Both variants of scenario 3 perform worse compared to the base case relative to the benchmark due to the higher prices of inflation and fossil fuel. Figure I.4 shows that the performance is better relative to the base case only for the transportation, warehouse and equipment, and brick-and-mortar store emissions. Although renewable energies and green transportation modes are increasingly attractive, the costs are still too high. The extreme scenario, where more environmental improvements are made but a higher price increase is assumed, shows even worse performance compared to the moderate variant of this scenario. Mainly, the labor cost is high in this scenario's extreme variant, resulting in a price increase of 10%.

The performance gaps of both moderate and extreme variants of scenario 4 are smaller compared to the gap of the base case with the target performance, meaning that improvements are made. This results from a higher packaging recycling rate, more environmentally packaging material, a higher reuse rate of the clothes, more products donated, increased adoption of renewable energy sources, and fewer products disposed of or sent to landfills (see Figure I.5). Yet, adopting these efforts leads to high recovery costs ( $V_9$ ). It can be seen from the extreme variant of this scenario that the efforts focusing on adopting renewable energy sources in buildings ( $V_3$  and  $V_5$ ) come close to the benchmark values. This scenario illustrates that the government's financial support has ensured that the overall price increase due to company investments does not outweigh the environmental improvements. This makes both variants of this scenario function better in terms of overall performance than the base case. No improvements are made for the transport costs and emissions in this scenario.

The moderate variant of scenario five, technological innovations, shows no improvements compared to the base case relative to the benchmark. The more detailed radar chart in Figure I.6 shows that the moderate variant of this scenario outperforms the base case on product disposal emissions. However, the higher recycling and transport costs lead to an overall decrease in the performance of this variant relative to the base case. In the extreme variant, improvements are visible due to more efficient transport, fewer in-store returns, higher clothing recycling rates, and lower costs due to economies of scale. Under these conditions, advancements are observable compared to the base case, and the performance gap with the benchmark is tightened. One outlier in the extreme variant compared to the moderate variant of this scenario is the transportation improvements, leading to relatively large improvements in transport emissions in the extreme variant.

# 7.4. Validation

Validation examines the question '*Is the model right*?'. This question relates to the defined objectives, requirements, and purpose. Validation consists of conceptual and operational validation [143]. This research will only validate the model by conceptual validation, as no real-life observed results are available to compare the model's output. Conceptual validation investigates the correctness of the assumptions made in the model and whether the model performs plausible results [104]. It is determined whether its behavior aligns with the expectations by testing the effect of different scenarios. Scenarios were designed based on the verification framework of PESTLE and expert consultation, which improves the reliability of the scenarios with real-life developments. After testing the scenarios with the model and conducting a performance analysis, it was concluded that the scenarios generated plausible results. This proves the correctness and reliability of the model and creates a better understanding of the model's behavior and how it performs. The model can provide desired outputs, serve its purposes, and create valuable insights.

The model is successfully validated by inspecting the plausibility of the behavior of the scenario outputs.

# 7.5. Main findings chapter 7

First, the sub-question *How can companies use the constructed CI in practice?* is answered. In the previous chapter, it was already found that a company can measure and compare the performance of its different return methods relative to a verified benchmark. This creates the knowledge for companies to encourage customers to make more environmentally friendly choices for returns based on their current performance. It also creates an understanding of how their current performance scores on the environmental targets the EU set on their performances. In this chapter, the verified benchmark was used to give companies insight

and understanding into potential improvement areas of their returns. The impact of external factors and the effects of adopting strategies to improve sustainability on a company's performance was analyzed using scenarios based on PESTLE, literature, reports, and expert consultation. The outcomes were translated to performance gaps, identifying a company's performance relative to their benchmark performance. The results showed companies insight into the effects of implementing different strategies or actions and external factors on their performance. This helps identify potential areas of improvement and risk and how they can reach the targets set by the European Commission (EC), demonstrating how companies can use this CI and the insights created by its usage.

The model is conceptually validated by investigating the model's behavior by scenario analysis. Multiple scenarios were developed based on the PESTLE framework, which was tested with the model, and their effect on the model's output was evaluated. The model is successfully validated by inspecting the plausibility of the model's behavior and the scenario outputs. This proves the correctness and reliability of the model and contributes to a better understanding of the model's performance, answering the question *How can the constructed CI be validated?* 

# 8

# **Conclusion and Discussion**

This research will conclude with the findings of this research. Each research chapter has ended by answering one of the sub-research questions; these insights will be combined to formulate an answer to the Main Research Question (MRQ). Next, this research's scientific and practical contributions are discussed, together with the limitations. Finally, recommendations for future research are made.

# 8.1. Main findings

To achieve the goals the European Union (EU) set to combat climate change and comply with increased market pressure, the fashion industry must change and lower its Greenhouse Gas (GHG) emissions [72, 182]. The ongoing rise of product returns has become an increased problem for this industry as it exacerbates its GHG footprint when it should be lowered. This leads to undesirable consequences for companies and society, and improvement of returns is necessary [63, 36, 50]. The problem of returns should be tackled by both lowering the environmental impact of the process of unavoidable returns and the number of returns in general [36, 63], leading to environmental gains for society and cost savings for a company [194, 63]. Although transparency and understanding of the impact of returns on the performance of a company are needed to effectuate change [182], there is a knowledge gap as the environmental impact of product returns, and how this impact differs per return method is poorly understood, and accurate methods to assess this impact lacks [183, 36]. Therefore, this study has assessed the impact of product returns on the performance of fashion companies and created an understanding of this to improve the environmental footprint. Besides the scientific relevance of developing a method for assessing returns from an environmental viewpoint, the fashion industry is primarily profit-driven, and this study considers both dimensions relevant to ensure effective and realistic decision-making. Therefore, the economic impact of product returns on a company's performance is also measured, creating a company performance measurement with multiple dimensions. The outcome is a Composite Indicator (CI) for companies to measure, analyze, and evaluate their performance and see how different return methods or improvement strategies affect this. A new benchmark is designed based on GHG reduction targets of the European Commission (EC) for 2030 which is developed and integrated into this CI to help companies improve sustainability. The MRQ of this study is:

"How can a Composite Indicator be constructed to provide insight into the environmental and economic performance of product returns for companies within the fashion industry and help improve their sustainability?"

This CI was constructed in five steps, where the performance is measured using indicators. First, it was determined which environmental and economic indicators needed to be included in this CI to measure the impact of product returns on a company's performance and how these could be quantified. This was done using a literature study and interviews with experts, and the findings were combined into a theoretical framework. Second, the indicators were normalized by rescaling. Third, the Analytical Hierarchy Process (AHP) technique was applied to weigh each indicator. The indicators were aggregated into one measure in the fourth step using the multiplication technique, followed by evaluation of the robustness of the CI using a Uncertainty Analysis (UA) and a Sensitivity Analysis (SA). Lastly, a benchmark was designed using literature, reports, and expert consultation and developed and integrated into the constructed CI, where AHP weighting

was replaced by Equal Weights (EW).

In more detail, five sub-questions were defined that were answered throughout the research to pose an answer to the MRQ. The main findings of these questions are elaborated on below.

SQ1. What does the fashion industry's product return flow look like?

The returns process consists of a return request, collection, handling, processing at the warehouse, and product recovery and disposal. The return process starts when the customer requests a return and selects a method to process this, which can be: by brick-and-mortar store return, drop-off at a collection point, or by home pick-up. The availability of the return methods for a customer depends on whether the retailer offers this service. The customer must re-pack the product for safe transport if home pick-up or collection points are chosen. Next, the product is collected and transported from the customer to the final recovery destination, either the warehouse or the brick-and-mortar store. How this transport is arranged depends on the return method chosen. Products with the brick-and-mortar store as the final destination are restocked and resold. If the product is sent to the warehouse, it will be received, handled, inspected, and temporarily stored until a decision is made on how to recover it. The recovery choice determines if the product gets resold, recycled, repaired, remanufactured, or disposed to landfills or incineration [45]. After the recovery activity is performed, the process of returns is finished.

# SQ2. How can the environmental impact of the various logistic activities involved in product returns be assessed?

The following elements were essential for assessing the environmental impact of product returns: packaging material, packaging waste treatment, transportation, warehousing, storage, equipment usage, product recovery, and brick-and-mortar activities. An activity-specific Emisison factor (EF) was used to measure the GHG emissions of each activity, following the guidelines of the GHG Protocol. The choice of a return method is relevant to the environmental impact of packaging, transportation, and brick-and-mortar activities.

*SQ3. How can the economic impact of the various logistic activities involved in product returns be assessed?* According to several experts, activity-based costing is a commonly used method to assess the return costs for a company. The following costs were considered to create a complete overview of the process of returns from an economic perspective: transport costs, warehouse space utilization costs, equipment costs, labor costs, product recovery and disposal costs, product storage costs, in-store space utilization costs, and instore product processing costs. Measurement of these costs was done on an annual basis and specified per return.

#### SQ4. How can a CI be created to assess product returns' environmental and economic impact?

The structure of a CI involved five steps and was done according to the guidelines proposed by [134]. First, ten indicators to measure were selected using the insights created in SQ1, SQ2, and SQ3. This provided insight into the raw indicator values and the overall environmental and economic impact of a return method before aggregation. Second, the indicators were normalized by rescaling and weighted by AHP. AHP allows companies to make a trade-off between the importance of the indicators amongst each other but ensures consistency by using statistical methods to determine the weights [134]. Next, the indicators were aggregated using multiplication, a commonly used technique that allows straightforward interpretation and partial compensability of the indicators [120, 133]. The same construction techniques were used to design the benchmark in the CI, but the weights were assigned using EW to improve the transparency of the benchmark. Lastly, evaluation of the CI was done by performing a UA and SA that verified and validated the CI using data from a fashion retailer. This confirmed the robustness of the model, and evaluation has proven that the CI can meet its objectives and requirements.

# Q5. What is the most preferred return method for a company considering its environmental and economic performance?

The environmental and economic impact of each return method in the base case were calculated in the first phase of the CI. Combining the environmental set of indicators shows that from the return methods, home pick-up creates the lowest GHG emissions per return, followed by collection point returns, and in-store returns score worst. The costs per return are highest for in-store returns, followed by collection point returns, and home pick-up has the lowest cost per returned item. The differences between the performances of the return methods are larger for the economic set of indicators. So, considering the environmental and economic

performances separately, home pick-up performs best on both dimensions and is preferred over the other return methods. This outcome aligns with the results from the aggregated CI scores, where home pick-up has the best overall performance, collection point in between, and in-store returns the worst.

# 8.2. Discussion

This section discusses the scientific and practical research contributions. Additionally, the limitations of this research are discussed to give a better understanding and interpretation of the results.

### 8.2.1. Scientific contributions

Multiple contributions were made in this study to scientific research. The first contribution was adding research to the scarce scientific literature on reverse logistics in the fashion industry, focusing on the effects and potential solutions for improving the environmental impact of customer product returns. The need to create this knowledge and increase transparency on the environmental impact of product returns was emphasized by [202, 23]. The second contribution was developing a measurement method for analyzing the GHG emissions of a return that include all relevant logistic activities and return methods. The few studies in this field were incomplete as these did not include all relevant logistic activities in this process, were limited to only analyzing carbon emissions, or did not consider the differences between the return methods [36, 118, 51]. The third contribution was developing a new CI that fashion companies can use to measure, evaluate, and improve the impact of product returns on their environmental and economic performance, which was lacking in research. Companies can thereby provide transparency on the impact of returns and can encourage more conscious return behavior [194, 23], which was a gap in the literature [22]. In addition, this research expanded the field of research on company performance measurement with CIs by assessing both environmental and economic dimensions [203]. The final contribution of this research was providing a new overview and measurement method where insight into the performance of different return methods was obtained and how they differ on their performance for a company. Although it was acknowledged that the impact of these return methods differs and improvements could be gained by choosing methods based on their performance [23, 22, 51], this insight was lacking in the literature.

### 8.2.2. Practical contributions

The practical contributions made with this research are formulated from a fashion company's and society's perspectives.

#### Perspective fashion companies

This research contributes to improving the ongoing problem for companies that may account for approximately 12% of their GHG emissions [9] and where they may lose yearly approximately 17% of 1 billion dollars in sales to [3, 135]. Contributions are made to improve companies' performances by, on the one hand, lowering the number of returns and, on the other hand, improving the current process for unavoidable returns.

Fashion companies can improve their performances by measuring the process of returns: lowering their environmental footprint- and costs. This study provides insight into the sources of GHG emissions and the costs of product returns, and how companies can measure them. Also, differences between the return methods and their performance are made explicit for companies. Companies can make well-informed decisions to improve the environmental impact of returns as they can define, measure, compare, and prioritize improvement efforts based on environmental impact while ensuring economic viability [182]. This relevance was proven by conducting a case study of a fashion retailer. It showed valuable insights into how its performance could be improved by avoiding or selecting a return method based on its impact and which trade-offs could be made between environmental and economic performance for implementing new strategies to improve the return impact.

Companies can mitigate returns by encouraging customers' conscious return behavior, leading to environmental and economic gains for a company [194, 23]. This study has provided a tool for companies to gain insights into the impact of returns and the different methods so that they can measure, analyze, and compare the performance between return methods. Awareness can be created about the impact of returns, and opportunistic return behavior can be limited, lowering returns. Also, with the increased transparency, companies can answer the increased customer pressure for more well-informed and conscious return choices [22]. Conscious return choices- and behavior improve company performance on both environmental and economic facets as less environmentally damaging return methods are chosen [23], and fewer products are expected to be returned [194]. This research showed that encouraging conscious behavior and thereby reducing the number of returns is a very effective way to improve the overall performance of a company.

This research has proven the added value for companies to compare their performances with a benchmark. Companies know how far they are in reaching the goals of the EC to combat climate change. The EC targets for 2030 to lower the GHG emissions were translated into indicator-specific goals in this study. This research has defined a verified benchmark representing these target values. Companies can compare their performance relative to this benchmark. With this, companies can see how far they reach these targets and how- and to what extent (potential) changes in their operations might lead to solutions to close this gap. Additionally, companies can use the results of this CI to respond to the prospected environmental reporting of the EC on the environmental impact of operations.

#### Societal perspective

This research contributes to halting climate change by focusing on one of the most environmentally polluting industries worldwide. Attention is given to the increased problem of product returns, which received too little attention previously. Although returns are not the largest share of emissions along the value chain, the problem of returns continuously grows and generates negative externalities for society [63, 36]. This research pays attention to this problem by providing insights into how returns can be mitigated and how the process of returns can be improved.

The role of fashion companies was already mentioned. Further, customers have an essential role in establishing change as well. Increased customer awareness on the impact of returns and limiting opportunistic return behavior is needed to achieve the potential savings of 12 million tonnes of GHG emissions when the return rates would decrease by 20% [12]. This research is dedicated to increasing knowledge on the impact of returns and contributes to creating awareness about the environmental costs of returns. Customers should also be held responsible for the negative externalities of these returns created, as their behavior drives high return rates. This is an essential contribution of this research, as besides the companies, customers also have to take responsibility and have an important role in lowering the environmental impact of product returns and creating a positive impact on society.

Next, a contribution is made as a transparent tool is provided, where customers and governments know which- and how the environmental impact of a product return is measured for a company's performance. The insights created give customers clear and transparent information about the GHG emissions of a return, where the complete return process and return activities are considered. For governments, this tool contributes as it can be used in their prospected mandatory environmental reporting sector-wide. Governments can use this transparent tool to evaluate, monitor, and track the progress of a company's performance and evaluate the efforts made by companies in achieving the EC goals to become net-zero. Companies' performances within the sector can be compared with each other as they are measured with the same measures, their progress can be evaluated, and governments can use this benchmark to set more concrete targets to combat climate change.

Further, governments are drivers to effectuate change in this industry and should have a more leading and supporting role in tackling the growing problem of returns. Currently, no specific government regulations target the return process of fashion companies, although this is a growing phenomenon in the industry that exacerbates the GHG emissions. A more leading and supporting role is desired. Regulations incentivizing companies to integrate sustainability practices in their returns are essential, as well as government financial support for companies to remain profitable.

#### 8.2.3. Limitations

This section gives a discussion on the limitations of the research and the results to understand and interpret them better.

#### Research scope and approach

As reducing GHG is the top priority of the EC to combat climate change, this research is demarcated to only measuring the GHG emissions of the process of a return [67]. The relevant EC targets set for reducing GHG emissions are translated into a target benchmark in this research. As the assessment of this research is consistent with the unit of measurement of these targets, this tool is highly suitable for companies to measure their progress relative to these EC objectives. Yet, the environmental impact of product returns is broader than only the GHG emissions, as negative externalities like noise and water pollution are also created [106, 170, 72]. These effects are not considered in this assessment. A more comprehensive overview of the environmental impact of product returns can be made by including these factors, which is a limitation due to this study's scope.

Further, only the environmental impact created in the return process was considered in this study. However, companies increasingly use carbon offsetting alternatives, where financial compensations are paid to compensate for the emissions the buyer creates, lowering their emissions [82]. This is not considered in this research. If a company uses carbon offsetting while using this research's CI, it should be specified for which activity the offsetting is used, and double-counting should be avoided. It is essential to provide transparency on this to enhance the quality of the outcomes.

#### Research methodology, results, and data

This research aimed to create a CI to measure company performance in the process of product returns. Constructing a CI involves some level of subjectivity as choices must be made for the construction methods. To lower potential subjectivity, the choices were communicated, motivated, and made transparent in this research. In addition to this, the effect of these choices on the models' output was investigated by performing a combination of UA and SA. The results showed that the constructed CI is robust, as using a different construction method leads to similar results, increasing the reliability of the selection of construction methods. Special care was given to test the reliability of the weighting method. Although the consistency of the weights was verified using a statistical test, the expert group was within the recommended range of 3-7 experts, and a different group of experts was used for rescaling, it might be argued that the reliability of the weighting method still can be improved by increasing the number of experts to assign weights. Nevertheless, the choice of construction methods was successfully verified, lowering the risk of subjectivity.

The first objective of the CI was looking at whether the performance of an average return item with the different return methods for a company can be measured and how- and if these methods can be compared. The CI shows that home pick-up returns perform best based on the data set used of a fashion retailer, followed by collection point returns and the in-store scoring worst for this retailer. Considerable differences are seen in transport emissions- and costs, the warehouse- and equipment costs, and the warehouse- and equipment emissions between the methods. Identification of the underlying causes of these outliers shows plausible and logical results. For example, home pick-up performs much better on the average transport emissions per return due to the more efficient transport networks. However, home pick-up is more costly as longer distances are driven with relatively expensive distribution vans. In-store returns have lower transport costs as customers bring the product to the brick-and-mortar store but score worst on transport emissions as no consolidated shipments are used. From this in-depth analysis of the results of the return methods with this data set, it can be concluded that the first objective can be served with this CI. Nevertheless, it should be noted that the findings from the base case in this research cannot be directly generalized to other fashion retailers as they are highly company-specific. For example, transport emissions depend on the distance driven, which depends on the location of the warehouse, customers, distribution centers, return strategies, and modalities used. So, it can be that different return methods perform better than the findings from this research pose based on the data set used. This research is limited to the data set of only one fashion retailer, so to create more generalizable results, data sets from different companies should be measured, analyzed, and evaluated.

The second objective of this CI was to enable companies to identify areas of improvement, track progress, and compare their performance relative to targets. These objectives can be fulfilled with the CI developed in this research. A benchmark was constructed according to specific targets driven by Legal (L) and Policy (P) factors from the PESTLE framework, which serves as a verification tool. Companies can compare their actual performance with the target performance by measuring the performance gap. The magnitude of this gap expresses how far a company is in terms of performance in reaching the targets. Scenarios were designed using the PESTLE framework, literature, reports, and expert consultation, which showed the practical usage of the CI to identify improvement areas and economic risks on their performance. It was found that, based on this company's data set, the most considerable improvements in overall performance can be achieved when fewer products are returned due to more conscious behavior and, on the other hand, a higher recycling rate of packaging materials is achieved. Additionally, a return fee to discourage customers from returning items also leads to performance improvements relative to the base case, closing the gap with the benchmark. This is a crucial component to lower the costs of returns when no governmental support is given. The scenario analysis shows the importance of the financial support of governments for companies to ensure feasibility in company efforts to improve their environmental impact. Adopting less harmful waste treatments that enhance circularity shows improvements in environmental areas but are slightly higher in costs, affecting the overall performance. The gains achieved by technological innovations depend on the availability of technology and economies of scale that might help to lower costs, shrinking the performance gap with the benchmark. The insight gathered from evaluating the results of the CI can be used to make more informed decisions, prioritize improvement efforts, and design actions to improve sustainability while ensuring economic feasibility. These analyses validated the models' behavior and results, showing that the model can meet its objectives, requirements, and purpose.

When interpreting the results, emphasis should be given to the benchmark. The benchmark represents a company's shorter-term goals to comply with EU targets by 2030 in its net-zero journey for 2050. As companies can make trade-offs between environmental and economic indicators to measure their performance, EW is applied for the benchmark values in this CI. This ensures that the values of the benchmark are generalizable and transparent sector-wide. For the definition of the environmental indicators, the targets for 2030 are derived from EU targets and reports and stated in section 6.2. Companies need to reduce their emissions according to reduction percentages set by the EU. This is incorporated in this benchmark but has as a result that the raw values of the benchmark are company-specific. The reduction targets are translated into values for the benchmark using the data set of the base case for the specific base year. So, when a company uses this CI, it should define the benchmark values accordingly to its own data. Further, the target values are determined according to EU legislation, so this should be considered when a company outside the EU wants to use the benchmark targets. The set-up and structure of the benchmark can be used. Still, the target reductions should be defined accordingly to the policy and legal objectives and requirements that the company needs to obey. It is recommended to verify the target values with experts to ensure an accurate benchmark. As mentioned, defining any accurate target values for the economic indicators for 2030 is not possible. The economic indicator values are subject to uncertainty and external influences, so deriving numbers for these indicators for 2030 is not likely to be representative or accurate. Therefore, the base case values for the economic indicators are used in the benchmark. This is certainly a limitation of this research and should be considered when interpreting the results. By exploring the effect of different scenarios on a company's performance, the effect of the economic indicators was demonstrated. This analysis has shown that variance of the economic indicators can lead to large improvements in the overall performance of the CI, which needs to be considered for the CI interpretation.

According to expert consultation, the input data of this CI to measure the environmental and economic impact of the returns is well designed and suitable for companies. Companies should have all the inputs required by themselves to measure their performance. Data from the client retailer was obtained, analyzed, and prepared with a data expert from Metyis. Unreliable data points were removed to ensure consistency in the data set, and data from a whole year was used, so demand fluctuations were considered. However, the data set used in this research was not fully complete, partially due to confidential reasons, so some data was not shared. Assumptions needed to be made, and input data from secondary sources were used for supplementation. Special care was given to inspect the reliability and accuracy of the input data and the assumptions made. The input data and assumptions were validated by looking into the reliability of the data source, its year, and geographical dependency of the data, and experts checked assumptions with their industry knowledge. Additionally, the impact of some of these uncertainties in input data on the final value of the model was analyzed and evaluated using a SA. This statistical test revealed that the model's outcome is robust to changes in data, reflecting the resilience of the CI and increasing its reliability. However, the data's reliability would be improved when this is not obtained through secondary sources, all data originates from the same year, and fewer assumptions were made. Nonetheless, special care is given to this, and the reliability of the missing data and assumptions is increased through validation.

From the literature, several waste treatments were identified [45, 107, 5, 77]; however, expert consultation indicated that the treatments used are pretty limited for return operations by fashion companies in practice. Recycling is increasingly promoted to improve circularity, but in reality, the adoption of this technique is currently shallow due to technical and economic feasibility [52]. Additionally, assessing this impact requires a high level of detail, as the emissions created due to recycling largely depend on the type of clothing recycled, the technique chosen, and the efficiency of that technique [166]. As there is very little data available on this treatment in the literature, additional assumptions would need to be made that would negatively affect the quality and reliability of this research. Therefore, it was decided to exclude recycling as an alternative in the base case. For the scenario analysis, the impact of mechanical recycling is explored and included in one of the alternatives as optional waste treatment. The costs- and emissions are based on reports but are still highly embedded with uncertainty, which should be considered with interpreting the results. If a company uses recycling techniques, they can adopt this in this CI by including the economic cost of this treatment. The environmental impact should be determined as accurately as possible, where preferably the type of clothing

item and its material is considered.

Another limitation of this research is that it assumes the activities of the Third Party Logistics Provider (3PL)s at the distribution centers activities of the 3PL to be neglectable. However, these activities may generate additional emissions, depending on the 3PL their performance, and should therefore be discussed. This research aimed to measure the environmental and economic impact of returns on the performance of fashion companies. Consultation with experts from the industry gave insight that fashion companies lack information and data on the GHG emissions of their 3PL at their transfer locations or distribution centers. Additionally, the specific activities at these distribution and sorting centers of the 3PLs are not within the control of fashion companies, making it difficult or even impossible for them to effectuate change here. So, assumptions are necessary to make. From reports published by the 3PLs, it follows that for PostNL, the 3PL transporting more than 90 percent of the returns in the base case, their buildings are entirely free of emissions and optimizing their operations [151]. However, for other providers, like UPS, insights are lacking. This lack of data makes it difficult to make reliable assumptions about the emissions generated by these activities on the product return level. Therefore, the decision to neglect these emissions in this research is twofold; first, the information provided by PostNL on their performances and the total share of this 3PL as a transport provider for the returns in the base case, second the lack of data on other 3PL. This assumption is further investigated by performing a SA for the warehouse emissions, which found that the results generated by the model in the base case were not sensitive to changes in this variable. However, this is still a limitation in this research, and further investigation into the activities and emissions generated at these distribution centers would improve the accuracy.

# 8.3. Recommendations

This section gives recommendations and outlines future research possibilities. Some of these recommendations come from the research findings and limitations; others are based on insights gained during the research, which require further investigation.

The main recommendation of this research is for companies to offer return methods to customers based on how they perform to improve the environmental impact. The findings of this research have shown that the impact of the return methods differs, considering both environmental and economic dimensions. Improvements can be achieved when companies avoid return methods that perform worse as an alternative to return a product.

Next, as a transparent tool is provided on which- and how the environmental impact of a company is measured, it is recommended for governments to use this tool in their environmental sustainability reporting for companies. Additionally, the integrated benchmark can be used by the government to compare the performances of companies sector-wide, measure their progress, and set further targets to combat climate change.

This study has created the information needed to provide transparency and inform customers but does not analyze how- and where companies in the purchasing- and return process can use this information the best to encourage customers' conscious return decisions- and behavior most effectively. Therefore, it is recommended to develop a dashboard or digital twin where insights are obtained where- and how customers can best be encouraged using the insights obtained from this study.

Further, it is recommended that companies mitigate the number of returns by adjusting the leniency of return policies. Lenient return policies of companies are causes for high return rates. The scenario analysis has shown that lowering the number of returns and charging customers a return fee significantly improves overall performance. However, companies are less willing to change these policies and make them less lenient as it is a tool to boost sales and attract customers and, therefore, impacts their profitability, they say [194, 175, 93, 101]. However, this research showed the potential to improve the company's performance by changing the return policies. The potential effect on customer sales was left unconsidered, forming the basis for the following possibilities for future research, which have both practical and scientific implications. First, the findings of this study should be incorporated to design new return policies, targeting lowering returns using financial discouragements based on the environmental impact created when returning a product. Research should be conducted into optimal financial discouragement while considering its effect on customer demand and a company's profitability. Also, exploring the efficiency of targeting specific product categories with their return policies is recommended, as some categories cause higher returns than others. The effect of the policies can be analyzed and tracked using the CI developed in this research. These insights may help

companies adjust their return policies, contributing to lower returns and improved environmental impact.

Investigating the decision-making behind the choice of a company on the final destination of a product is recommended, as this affects performance. It should be explored why a product will be sent back to the store or returned to the warehouse to be handled. This research shows that the largest share of returned items is redirected to stores for particular home pick-ups in this case study. For returns through collection points, this percentage is much lower. The research has shown that whether a product is directed to a brick-and-mortar store or a warehouse significantly impacts the performance of a return method. Gathering insight into how these decisions are made can lead to improved performance for the company and is therefore recommended as a potential area of future research.

Future research should be done to increase the adoption potential of textile recycling and overcome its current limitations, where both the technical and practical feasibility should be considered for fashion companies. The recycling rates are currently meager within the fashion industry, although this creates a better environmental impact than landfill or disposal [45, 52]. A technical recommendation is, therefore, to investigate textile recycling alternatives for fashion companies and improve this adoption potential for both used and unused products. Additionally, barriers and drivers should be explicit, and insights need to be created into how they can be overcome, contributing to increased adoption of the technique.

The final recommendation for future research is to investigate governmental policies' role in lowering the number of product returns in fashion and identify the most effective approach. Experts in the field have noted that to incentive change within the operations of fashion companies, governments either need to have a strict policy or there should be pressure from shareholders, primarily because of the high costs. This study has shown that financial support is essential for companies to improve the sustainability of their operations. It is recommended to investigate governmental policies' role in lowering the number of product returns in fashion and which approach would be most effective. This includes examining, for example, the most effective targets for recycling-, landfill-, and incineration rates to improve the environmental footprint of this industry while considering the economic feasibility for a company in terms of financial subsidies or investments. This will contribute to the adoption potential of textile recycling in the fashion industry and improve sustainability. Additionally, it is recommended to incorporate the designed CI and benchmark to investigate the effectiveness of implementing these policies and monitor the effects on companies' performances.

# Bibliography

- A-Lined Handling Systems. Belt and incline conveyors. URL https://a-lined.com/products/ belt-incline-conveyors/.
- [2] A. Abdel-Maksoud, D. Dugdale, and R. Luther. Non-financial performance measurement in manufacturing companies. *The British Accounting Review*, 37(3):261–297, 9 2005. ISSN 08908389. doi: 10.1016/ j.bar.2005.03.003. URL https://linkinghub.elsevier.com/retrieve/pii/S0890838905000235.
- [3] D.-A. Adegeest. Cost of returns soars to 761 billion dollars, 1 2022. URL https://fashionunited.uk/ news/retail/cost-of-returns-soars-to-761-billion-dollars/2022012660919.
- [4] J. Ader, J. Chai, M. Singer, P. Adhi, and H. Yankelevich. Returning to order: Improving returns management for apparel companies. *McKinsey & Company*, 4 2021. URL https://www.mckinsey.com/industries/retail/our-insights/returning-to-order-improving-returns-management-for-apparel-companies.
- S. Agrawal, R. K. Singh, and Q. Murtaza. Disposition decisions in reverse logistics: Graph theory and matrix approach. *Journal of Cleaner Production*, 137:93–104, 11 2016. ISSN 09596526. doi: 10.1016/j.jclepro.2016.07.045. URL https://linkinghub.elsevier.com/retrieve/pii/ S0959652616309337.
- [6] P. Ambilkar, V. Dohale, A. Gunasekaran, and V. Bilolikar. Product returns management: a comprehensive review and future research agenda. *International Journal of Production Research*, 60(12): 3920–3944, 6 2022. ISSN 0020-7543. doi: 10.1080/00207543.2021.1933645. URL https://www.tandfonline.com/doi/full/10.1080/00207543.2021.1933645.
- [7] A. E. Anderson, B. J. Ellis, and J. A. Weiss. Verification, validation and sensitivity studies in computational biomechanics. *Computer Methods in Biomechanics and Biomedical Engineering*, 10(3):171–184, 6 2007. ISSN 1025-5842. doi: 10.1080/10255840601160484. URL http://www.tandfonline.com/doi/abs/10.1080/10255840601160484.
- [8] M. Asif and C. Searcy. A composite index for measuring performance in higher education institutions. *International Journal of Quality and Reliability Management*, 31(9):983–1001, 9 2014. ISSN 0265671X. doi: 10.1108/IJQRM-02-2013-0023. URL https://www.emerald.com/insight/content/doi/10. 1108/IJQRM-02-2013-0023/full/html.
- [9] ASOS. Carbon 2020 Progress report 2018-2019. Technical report, 2020. URL https://asos-12954-s3.s3.eu-west-2.amazonaws.com/files/8816/3232/1509/ carbon-report-18-19-final.pdf.
- [10] S. Azevedo, R. Godina, and J. Matias. Proposal of a Sustainable Circular Index for Manufacturing Companies. *Resources*, 6(4):63, 11 2017. ISSN 2079-9276. doi: 10.3390/resources6040063. URL http://www.mdpi.com/2079-9276/6/4/63.
- T. A. Banihashemi, J. Fei, and P. S.-L. Chen. Exploring the relationship between reverse logistics and sustainability performance. *Modern Supply Chain Research and Applications*, 1(1):2–27, 2 2019. ISSN 2631-3871. doi: 10.1108/MSCRA-03-2019-0009. URL https://www.emerald.com/insight/content/doi/10.1108/MSCRA-03-2019-0009/full/html.
- [12] A. Berg, A. Granskog, L. Lee, and K.-H. Magnus. Fashion on climate: How the fashion industry can urgently act to reduce its greenhouse-gas emissions. Technical report, McKinsey & Company and Global Fashion Agenda, 2020. URL https://www.mckinsey.com/industries/retail/our-insights/ fashion-on-climate.

- [13] N. Bernando. Choosing the Right Order Picker for your Operations. URL https://www. totalwarehouse.com/blogs/choosing-the-right-picker-for-your-operations/.
- [14] Bernon and J. Gorst. Online retail returns management: Integration within an omni-channel distribution context. *International Journal of Physical Distribution & Logistics Management*, 46(6/7), 2016. URL http://shura.shu.ac.uk/11656/.
- [15] M. Bernon, S. Rossi, and J. Cullen. Retail reverse logistics: A call and grounding framework for research. *International Journal of Physical Distribution and Logistics Management*, 41(5):484–510, 6 2011. ISSN 09600035. doi: 10.1108/09600031111138835. URL https://www.emerald.com/insight/content/ doi/10.1108/09600031111138835/full/html.
- [16] R. F. Bertram and T. Chi. A study of companies' business responses to fashion e-commerce's environmental impact. *International Journal of Fashion Design, Technology and Education*, 11(2):254–264, 5 2018. ISSN 1754-3266. doi: 10.1080/17543266.2017.1406541. URL https://www.tandfonline.com/doi/full/10.1080/17543266.2017.1406541.
- [17] Better Packages. What's the Difference: Cardboard vs. Corrugated Cartons , 1 2023. URL https://inbound.betterpackages.com/blog/whats-the-difference-cardboard-vs. -corrugated-cartons.
- [18] P. Beullens, D. Van Oudheusden, and L. N. Van Wassenhove. Collection and Vehicle Routing Issues in Reverse Logistics. In M. Dekker Rommert
  and Fleischmann, I. Karl, and V. W. L. N, editors, *Reverse Logistics*, pages 95–134. Springer Berlin Heidelberg, Berlin, Heidelberg, 2004. ISBN 978-3-540-24803-3. doi: 10.1007/978-3-540-24803-3{\\_}5. URL http://link.springer.com/10.1007/978-3-540-24803-3\_5.
- [19] F. J. Blancas and M. Lozano-Oyola. Sustainable tourism evaluation using a composite indicator with different compensatory levels. *Environmental Impact Assessment Review*, 93:106733, 3 2022. ISSN 01959255. doi: 10.1016/j.eiar.2021.106733. URL https://linkinghub.elsevier.com/retrieve/ pii/S0195925521001839.
- [20] R. Bolado-Lavin and A. C. Badea. Review of Sensitivity Analysis Methods and Experience for Geological Disposal of Radioactive waste and Spent Nuclear Fuel. Technical report, European Commission Joint Research Centre Institute for Energy, 2008. URL https://publications.jrc.ec.europa.eu/ repository/handle/JRC49536.
- [21] A. B. Bower and J. G. Maxham. Return Shipping Policies of Online Retailers: Normative Assumptions and the Long-Term Consequences of Fee and Free Returns. *Journal of Marketing*, 76(5):110–124, 9 2012. ISSN 0022-2429. doi: 10.1509/jm.10.0419. URL http://journals.sagepub.com/doi/10.1509/jm. 10.0419.
- [22] C. Bozzi, M. Neves, and C. Mont'alvão. Fashion E-Tail and the Impact of Returns: Mapping Processes and the Consumer Journey towards More Sustainable Practices. *Sustainability (Switzerland)*, 14(9), 2022. ISSN 20711050. doi: 10.3390/su14095328. URL https://doi.org/10.3390/su14095328.
- [23] H. Buldeo Rai. The net environmental impact of online shopping, beyond the substitution bias. *Journal of Transport Geography*, 93(October 2020):103058, 5 2021. ISSN 09666923. doi: 10.1016/j.jtrangeo.2021. 103058. URL https://linkinghub.elsevier.com/retrieve/pii/S0966692321001113.
- [24] I. D. Cardenas, W. Dewulf, T. Vanelslander, C. Smet, and J. Beckers. The e-commerce parcel delivery market and the implications of home B2C deliveries vs pick-up points. *International Journal of Transport Economics*, 44(2):235–256, 6 2017. ISSN 03035247. doi: 10.19272/201706702004.
- [25] D. Casciani, O. Chkanikova, and R. Pal. Exploring the nature of digital transformation in the fashion industry: opportunities for supply chains, business models, and sustainability-oriented innovations. Sustainability: Science, Practice and Policy, 18(1):773–795, 12 2022. ISSN 1548-7733. doi: 10.1080/15487733.2022.2125640. URL https://www.tandfonline.com/doi/full/10.1080/ 15487733.2022.2125640.

- [26] CBS. CO2-equivalent, 2020. URL https://www.cbs.nl/nl-nl/nieuws/2020/19/ uitstoot-broeikasgassen-3-procent-lager-in-2019/co2-equivalent.
- [27] CBS. Mobiliteit; per persoon, verplaatsingskenmerken, vervoerwijzen en regio's, 2022. URL https: //opendata.cbs.nl/statline/#/CBS/nl/dataset/84708NED/table.
- [28] CBS. Mobiliteitstrend; per rit, vervoerwijzen, reismotief, leeftijd en geslacht, 11 2022.
- [29] CBS. Verkeersprestaties personenauto's, brandstof uitgebreid, leeftijd 2015-2020, 2 2023. URL https: //www.cbs.nl/nl-nl/cijfers/detail/83703NED.
- [30] Centraal Bureau voor de Statistiek. Inwoners per gemeente, 2022. URL https://www.cbs.nl/nl-nl/ visualisaties/dashboard-bevolking/regionaal/inwoners.
- [31] CO2-emissiefactoren. Lijst emissiefactoren. URL https://www.co2emissiefactoren.nl/ lijst-emissiefactoren/.
- [32] P. M. Coelho, B. Corona, R. ten Klooster, and E. Worrell. Sustainability of reusable packaging–Current situation and trends, 5 2020. ISSN 2590289X.
- [33] H. Constable. Your brand new returns end up in landfill, 6 2022.
- [34] Consumer Ecology. Carbon footprint of a cardboard box, 2022. URL https://consumerecology. com/carbon-footprint-of-a-cardboard-box/.
- [35] J.-F. Cordeau, G. Laporte, J.-Y. Potvin, and M. W. Savelsbergh. Transportation on Demand. In C. Barnhart and G. Laporte, editors, *Handbooks in Operations Research and Management Science*, number C, chapter 7. Elsevier, 2007. doi: 10.1016/S0927-0507(06)14007-4. URL https://linkinghub.elsevier.com/retrieve/pii/S0927050706140074.
- [36] S. Cullinane and K. Cullinane. The Logistics of Online Clothing Returns in Sweden and How to Reduce its Environmental Impact. *Journal of Service Science and Management*, 14(01):72–95, 2021. ISSN 1940-9893. doi: 10.4236/jssm.2021.141006. URL https://www.scirp.org/journal/doi.aspx?doi=10. 4236/jssm.2021.141006.
- [37] S. Cullinane, E. Karlsson, and Y. Wang. Retail clothing returns: A review of key issues. 2017. URL https://www.researchgate.net/publication/319528723.
- [38] Cushman & Wakefield. Winkelhuren Amsterdam 2022. Technical report, Gemeente Amsterdam Economische Zaken, Amsterdam, 2023.
- [39] N. F. da Cruz, S. Ferreira, M. Cabral, P. Simões, and R. C. Marques. Packaging waste recycling in Europe: Is the industry paying for it? *Waste Management*, 34(2):298–308, 2 2014. ISSN 0956053X. doi: 10.1016/j.wasman.2013.10.035. URL https://linkinghub.elsevier.com/retrieve/pii/ S0956053X13005217.
- [40] Davydenko I. and Hopman W. Effect of pick-up points and returns on CO2 emissions in last mile parcel delivery networks. Technical report, TNO, The Hague, 6 2020. URL www.tno.nl.
- [41] V. de Jonge, R. Koffrie, F. Mulder, K. Regtien, and K. Zagema. Routeradar Innovatiemonitor Marktontwikkeling Wegvervoer 2020, 7 2021. URL https://rwsduurzamemobiliteit. nl/beleid/routeradar/mmip-duurzaam-toekomstbestendig-mobiliteitssysteem/ routeradar-innovatiemonitor-marktontwikkeling/.
- [42] R. B. de Koster Marisa P. de Brito and M. A. van de Vendel. Return handling: an exploratory study with nine retailer warehouses. *International Journal of Retail & Distribution Management*, 30(8):407–421, 8 2002. ISSN 0959-0552. doi: 10.1108/09590550210435291. URL https://www.emerald.com/insight/content/doi/10.1108/09590550210435291/full/html.
- [43] DEFRA. Greenhouse gas reporting: conversion factors 2022, 6 2022. URL https://www.gov.uk/ government/publications/greenhouse-gas-reporting-conversion-factors-2022.

- [44] DEFRA. UK and England's carbon footprint to 2019, 11 2022. URL https://www.gov.uk/ government/statistics/uks-carbon-footprint.
- [45] R. Dekker, M. Fleischmann, K. Inderfurth, and L. Van Wassenhove. *Reverse Logistics*. Springer Berlin Heidelberg, Berlin, Heidelberg, 10 2004. ISBN 978-3-642-07380-9. doi: 10.1007/978-3-540-24803-3. URL http://link.springer.com/10.1007/978-3-540-24803-3.
- [46] J. Desrosiers, Y. Dumas, M. M. Solomon, and F. Soumis. Time Constrained Routing and Scheduling. In Handbooks in OR & MS, volume 8, chapter 2, pages 35–139. Elsevier Science B.V., 1995.
- [47] DHL. DHL Express' roadmap to decarbonization, 7 2022. URL https://www.dhl. com/discover/en-au/logistics-advice/sustainability-and-green-logistics/ roadmap-to-decarbonisation.
- [48] Directorate-General for Environment. EU strategy for sustainable and circular textiles, 3 2022. URL https://environment.ec.europa.eu/strategy/textiles-strategy\_en.
- [49] M. P. Dočekalová and A. Kocmanová. Composite indicator for measuring corporate sustainability. *Ecological Indicators*, 61:612–623, 2 2016. ISSN 1470160X. doi: 10.1016/j.ecolind.2015.10.012. URL https://linkinghub.elsevier.com/retrieve/pii/S1470160X15005476.
- [50] P. Dutta, A. Mishra, S. Khandelwal, and I. Katthawala. A multiobjective optimization model for sustainable reverse logistics in Indian E-commerce market. *Journal of Cleaner Production*, 249, 3 2020. ISSN 09596526. doi: 10.1016/j.jclepro.2019.119348.
- [51] J. B. Edwards, A. C. McKinnon, and S. L. Cullinane. Comparative analysis of the carbon footprints of conventional and online retailing. *International Journal of Physical Distribution & Logistics Management*, 40(1/2):103–123, 2 2010. ISSN 0960-0035. doi: 10.1108/09600031011018055. URL https: //www.emerald.com/insight/content/doi/10.1108/09600031011018055/full/html.
- [52] Ellen MacArthur Foundation. A new textiles economy: Redesigning fashion's future. Technical report, 2017. URL http://www.ellenmacarthurfoundation.org/publications.
- [53] D. Elrabaya, V. Marchenko, and D. Elrabay'a. Identifying the full cost to landfill municipal solid waste by incorporating emissions impact and land development lost opportunity: Case study, Sharjah-UAE. *International Journal of Engineering Sciences*, 10(6), 6 2021. doi: 10.35629/6734-1006023341. URL https://www.researchgate.net/publication/352464905.
- [54] S. Escursell, P. Llorach-Massana, and M. B. Roncero. Sustainability in e-commerce packaging: A review. *Journal of Cleaner Production*, 280:124314, 1 2021. ISSN 09596526. doi: 10.1016/j.jclepro.2020.124314. URL https://linkinghub.elsevier.com/retrieve/pii/S0959652620343596.
- [55] EuroDev. Cost of Employment in Spain, Portugal, and France, 12 2021. URL https://blog.eurodev. com/cost-of-employment-in-spain-portugal-france.
- [56] EuroDev. Cost of Employment in the Netherlands, Belgium and Luxembourg, 1 2022. URL https: //blog.eurodev.com/cost-of-employment-in-netherlands-belgium-luxembourg-benelux.
- [57] European Commission. About the EU Ecolabel. URL https://environment.ec.europa.eu/ topics/circular-economy/eu-ecolabel-home/about-eu-ecolabel\_en.
- [58] European Commission. Guidelines on non-financial reporting: Supplement on reporting climaterelated information. Technical report, Directorate-General for Financial Stability, Financial Services and Capital Markets Union, Brussels, 6 2019. URL https://ec.europa.eu/info/files/ 190618-climate-related-information-reporting-guidelines\_en.
- [59] European Commission. The European Green Deal, 12 2019. URL https://eur-lex.europa.eu/ legal-content/EN/ALL/?uri=CELEX:52019DC0640.
- [60] European Commission. 2030 Climate Target Plan, 9 2020. URL https://eur-lex.europa.eu/ legal-content/EN/TXT/?uri=CELEX:52020DC0562#document1.

- [61] European Commission. European Green Deal Delivering on our Targets. Technical report, European Commission, 2021. URL https://www.google.com/url?sa=t&rct=j&q=&esrc=s&source=web&cd=&ved=2ahUKEwjZtsi1idH9AhUUgf0HHa92C5kQFnoECAwQAQ&url=https%3A%2F%2Fec.europa.eu%2Fcommission%2Fpresscorner%2Fapi%2Ffiles%2Fattachment%2F869807%2FEGD\_brochure\_EN.pdf.pdf&usg=A0vVaw2liwvguvSd64yb0n3MQjq9.
- [62] European Commission. Proposal for a Directive of The European Parliament and of The Council. Technical report, European Commission, Brussels, 4 2021. URL https://eur-lex.europa.eu/ legal-content/EN/TXT/?uri=CELEX:52021PC0189.
- [63] European Commission. Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions: EU Strategy for Sustainable and Circular Textiles, 3 2022. URL https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX%3A52022DC0141.
- [64] European Commission. European Green Deal: Putting an end to wasteful packaging, boosting reuse and recycling, 11 2022. URL https://ec.europa.eu/commission/presscorner/detail/en/ip\_ 22\_7155.
- [65] European Commission. Landfill waste, 2022. URL https://environment.ec.europa.eu/topics/ waste-and-recycling/landfill-waste\_en.
- [66] European Commission. Proposal for a Regulation on Ecodesign for Sustainable Products, 3 2022.
- [67] European Commission. Questions and Answers on EU Certification of Carbon Removals, 11 2022. URL https://ec.europa.eu/commission/presscorner/detail/en/qanda\_22\_7159.
- [68] European Commission. Waste, 2022. URL https://ec.europa.eu/eurostat/web/waste/ targets.
- [69] European Environment Agency. Typical charge (gate fee and landfill tax) for legal landfilling of nonhazardous municipal waste in EU Member States and regions, 3 2013. URL https://www.eea. europa.eu/data-and-maps/figures/typical-charge-gate-fee-and.
- [70] European Environment Agency. Greenhouse gas emissions from transport in Europe, 10 2022. URL https://www.eea.europa.eu/ims/greenhouse-gas-emissions-from-transport.
- [71] European Union. Guarantees and returns, 10 2022. URL https://europa.eu/youreurope/ citizens/consumers/shopping/guarantees-returns/index\_en.htm.
- [72] European Union. The impact of textile production and waste on the environment (infographic), 4 2022. URL https://www.europarl.europa.eu/news/en/headlines/society/20201208ST093327/ the-impact-of-textile-production-and-waste-on-the-environment-infographic.
- [73] EuroStat. Recycling secondary material price indicator, 11 2023. URL https://ec.europa. eu/eurostat/statistics-explained/index.php?title=Recycling\_%E2%80%93\_secondary\_ material\_price\_indicator.
- [74] Eurostat. Monthly minimum wages bi-annual data, 1 2023. URL https://ec.europa.eu/ eurostat/databrowser/view/earn\_mw\_cur/default/table?lang=en.
- [75] J. Fichtinger, J. M. Ries, E. H. Grosse, and P. Baker. Assessing the environmental impact of integrated inventory and warehouse management. *International Journal of Production Economics*, 170:717–729, 12 2015. ISSN 09255273. doi: 10.1016/j.ijpe.2015.06.025. URL https://linkinghub.elsevier.com/ retrieve/pii/S0925527315002406.
- [76] S. Finnegan, S. Sharples, T. Johnston, and M. Fulton. The carbon impact of a UK safari park Application of the GHG protocol using measured energy data. *Energy*, 153:256–264, 6 2018. ISSN 03605442. doi: 10.1016/j.energy.2018.04.033. URL https://linkinghub.elsevier.com/retrieve/pii/S0360544218306376.

- [77] M. Fleischmann, H. R. Krikke, R. Dekker, and S. D. P. Flapper. A characterisation of logistics networks for product recovery. *Omega*, 28(6):653–666, 12 2000. ISSN 0305-0483. doi: 10.1016/S0305-0483(00) 00022-0.
- [78] M. Fleischmann, P. Beullens, J. M. Bloemhof-Ruwaard, and L. N. Van Wassenhove. The Impact of Product Recovery on Logistics Network Design. *Production and Operations Management*, 10(2), 2001.
- [79] M. Freudenberg. Composite Indicators of Country Performance: A Critical Assessment. 2003. URL https://www.researchgate.net/publication/273697960.
- [80] V.-A. Garrido Candela, K. Renata, K. Oyeshola, W. B. Oliver JRC Dir, R. Maria Rosa, H. Carme, F. Natalia, E. Marta, J. Gemma, J. Jaume, and B. Elisabet. Revision of the European Ecolabel Criteria for Lubricants Final Technical Report: Criteria proposal for revision of EU Ecolabel criteria. Technical report, European Union, Brussels, 6 2018. URL https://ec.europa.eu/jrc.
- [81] P. Gazzola, E. Pavione, R. Pezzetti, and D. Grechi. Trends in the Fashion Industry. The Perception of Sustainability and Circular Economy: A Gender/Generation Quantitative Approach. *Sustainability*, 12(7): 2809, 4 2020. ISSN 2071-1050. doi: 10.3390/su12072809. URL https://www.mdpi.com/2071-1050/ 12/7/2809.
- [82] A. Georgios and R. Srdan. The European Parliament's carbon footprint, Towards carbon neutrality, 9 2020. URL https://www.europarl.europa.eu/RegData/etudes/STUD/2020/652735/IPOL\_ STU(2020)652735\_EN.pdf.
- [83] GHG Protocol. A Corporate Accounting and Reporting Standard. Technical report, World Resources Institute and World Business Council for Sustainable Development, 2004. URL https://ghgprotocol. org/corporate-standard.
- [84] GHG Protocol. Technical Guidance for Calculating Scope 3 Emissions, 2013. URL https: //www.google.com/url?sa=t&rct=j&q=&esrc=s&source=web&cd=&cad=rja&uact=8&ved= 2ahUKEwidj\_rA67z9AhVAgfOHHcvcCNsQFnoECA0QAQ&url=https%3A%2F%2Fghgprotocol.org% 2Fsites%2Fdefault%2Ffiles%2Fstandards%2FScope3\_Calculation\_Guidance\_0.pdf&usg= A0vVaw33yIdeHazIP4adh4NRd-Ar.
- [85] A. Gialos, V. Zeimpekis, M. Madas, and K. Papageorgiou. Calculation and Assessment of CO2e Emissions in Road Freight Transportation: A Greek Case Study. *Sustainability*, 14(17):10724, 8 2022. ISSN 2071-1050. doi: 10.3390/su141710724. URL https://www.mdpi.com/2071-1050/14/17/10724.
- [86] D. I. Godil, Z. Yu, A. Sharif, R. Usman, and S. A. R. Khan. Investigate the role of technology innovation and renewable energy in reducing transport sector <scp> CO <sub>2</sub> </scp> emission in China: A path toward sustainable development. *Sustainable Development*, 29(4):694–707, 7 2021. ISSN 0968-0802. doi: 10.1002/sd.2167. URL https://onlinelibrary.wiley.com/doi/10.1002/sd.2167.
- [87] J. Gómez-Limón, M. Arriaza, and M. Guerrero-Baena. Building a Composite Indicator to Measure Environmental Sustainability Using Alternative Weighting Methods. *Sustainability*, 12(11):4398, 5 2020. ISSN 2071-1050. doi: 10.3390/sul2114398. URL https://www.mdpi.com/2071-1050/12/11/4398.
- [88] Greenhouse Gas Protocol. Life Cycle Databases, 2023. URL https://ghgprotocol.org/ life-cycle-databases.
- [89] E. Gustafsson, P. Jonsson, and J. Holmström. Reducing retail supply chain costs of product returns using digital product fitting. *International Journal of Physical Distribution & Logistics Management*, 51(8): 877–896, 8 2021. ISSN 0960-0035. doi: 10.1108/IJPDLM-10-2020-0334. URL https://www.emerald.com/insight/content/doi/10.1108/IJPDLM-10-2020-0334/full/html.
- [90] R. W. HALL. DISTANCE APPROXIMATIONS FOR ROUTING MANUAL PICKERS IN A WAREHOUSE. *IIE Transactions*, 25(4):76–87, 7 1993. ISSN 0740-817X. doi: 10.1080/07408179308964306. URL http: //www.tandfonline.com/doi/abs/10.1080/07408179308964306.
- [91] L. C. Harris. Fraudulent Return Proclivity: An Empirical Analysis. *Journal of Retailing*, 84(4):461–476, 12 2008. ISSN 0022-4359. doi: 10.1016/J.JRETAI.2008.09.003.

- [92] Z. He, P. Chen, H. Liu, and Z. Guo. Performance measurement system and strategies for developing low-carbon logistics: A case study in China. *Journal of Cleaner Production*, 156:395–405, 7 2017. ISSN 09596526. doi: 10.1016/j.jclepro.2017.04.071. URLhttps://linkinghub.elsevier.com/retrieve/ pii/S0959652617307898.
- [93] K. Hjort and B. Lantz. The impact of returns policies on profitability: A fashion e-commerce case. Journal of Business Research, 69(11):4980–4985, 11 2016. ISSN 01482963. doi: 10.1016/j.jbusres.2016. 04.064. URL https://linkinghub.elsevier.com/retrieve/pii/S0148296316302272.
- [94] A. Holding and A. Gendell. Polybags in the fashion industry: evaluating the options. Technical report, Fashion For Good, 12 2019.
- [95] Y. K. Hong and P. A. Pavlou. Product Fit Uncertainty in Online Markets: Nature, Effects, and Antecedents. *Information Systems Research*, 25(2):328–344, 6 2014. ISSN 1047-7047. doi: 10.1287/isre. 2014.0520. URL http://pubsonline.informs.org/doi/10.1287/isre.2014.0520.
- [96] A. Hübner, A. Holzapfel, and H. Kuhn. Distribution systems in omni-channel retailing. Business Research, 9(2):255–296, 8 2016. ISSN 2198-3402. doi: 10.1007/s40685-016-0034-7. URL https: //link.springer.com/10.1007/s40685-016-0034-7.
- [97] L. Hudrlíková and J. Kramulová. Do Transformation Methods Matter? The Case of Sustainability Indicators in Czech Regions. *Metodološki zvezki*, 10(1):31-48, 2013. URL http://epp.eurostat.ec. europa.eu/portal/page/portal/nuts\_nomenclature/local\_administrative\_units.
- [98] M. Ikram. Transition toward green economy: Technological Innovation's role in the fashion industry. *Current Opinion in Green and Sustainable Chemistry*, 37:100657, 10 2022. ISSN 2452-2236. doi: 10. 1016/J.COGSC.2022.100657.
- [99] E. Isaac and E. Chikweru. Test for Significance of Pearson's Correlation Coefficient (r). International Journal of Innovative Mathematics, Statistics & Energy Policies, 6(1):11–23, 1 2018. URL www.seahipaj. org.
- [100] A. Ishizaka and A. Labib. Review of the main developments in the analytic hierarchy process. *Expert Systems with Applications*, 38(11):14336–14345, 5 2011. ISSN 09574174. doi: 10.1016/j.eswa.2011.04. 143. URL https://linkinghub.elsevier.com/retrieve/pii/S0957417411006701.
- [101] N. Janakiraman, H. A. Syrdal, and R. Freling. The Effect of Return Policy Leniency on Consumer Purchase and Return Decisions: A Meta-analytic Review. *Journal of Retailing*, 92(2):226–235, 6 2016. ISSN 00224359. doi: 10.1016/j.jretai.2015.11.002. URL https://linkinghub.elsevier.com/retrieve/ pii/S0022435915000822.
- [102] I. Jestratijevic and U. Vrabič-Brodnjak. Sustainable and Innovative Packaging Solutions in the Fashion Industry: Global Report. Sustainability, 14(20):13476, 10 2022. ISSN 2071-1050. doi: 10.3390/ su142013476. URL https://www.mdpi.com/2071-1050/14/20/13476.
- [103] J. H. Kembro, A. Norrman, and E. Eriksson. Adapting warehouse operations and design to omnichannel logistics: A literature review and research agenda, 9 2018. ISSN 09600035.
- [104] L. A. Kerr and D. R. Goethel. Simulation Modeling as a Tool for Synthesis of Stock Identification Information. In *Stock Identification Methods*, pages 501–533. Elsevier, 2014. doi: 10. 1016/B978-0-12-397003-9.00021-7. URL https://linkinghub.elsevier.com/retrieve/pii/ B9780123970039000217.
- [105] H. Ketkale and S. Simske. A LifeCycle Analysis and Economic Cost Analysis of Corrugated Cardboard Box Reuse and Recycling in the United States. *Resources*, 12(2):22, 2 2023. ISSN 2079-9276. doi: 10. 3390/resources12020022. URL https://www.mdpi.com/2079-9276/12/2/22.
- [106] M. M. Khairul Akter, U. N. Haq, M. M. Islam, and M. A. Uddin. Textile-apparel manufacturing and material waste management in the circular economy: A conceptual model to achieve sustainable development goal (SDG) 12 for Bangladesh. *Cleaner Environmental Systems*, 4:100070, 3 2022. ISSN 2666-7894. doi: 10.1016/J.CESYS.2022.100070.

- [107] K. Khor and Z. Udin. Impact of Reverse Logistics Product Disposition towards Business Performance in Malaysian E& E Companies. *Journal of Supply Chain and Customer Relationship Management*, pages 1–19, 2 2012. ISSN 23267046. doi: 10.5171/2012.699469. URL http://www.ibimapublishing. com/journals/JSCCRM/2012/699469/699469.html.
- [108] R. Kiani Mavi, N. Kiani Mavi, D. Olaru, S. Biermann, and S. Chi. Innovations in freight transport: a systematic literature evaluation and COVID implications. *The International Journal of Logistics Management*, 33(4):1157–1195, 10 2022. ISSN 0957-4093. doi: 10.1108/IJLM-07-2021-0360. URL https://www.emerald.com/insight/content/doi/10.1108/IJLM-07-2021-0360/full/html.
- [109] J. H. Kim-Keung. Formulation of a Systemic PEST Analysis for Strategic Analysis. European Academic Research, 2(5), 8 2014. ISSN 2286-4822. URL www.euacademic.org.
- [110] A. Klein, D. Hilster, P. Scholten, L. Van Wijngaarden, E. Tol, and M. Otten. STREAM Goederenvervoer 2020 Emissies van modaliteiten in het goederenvervoer-Versie 2. Technical report, CE Delft, 2 2021. URL www.ce.nl.
- [111] M. Koszewska. Circular Economy Challenges for the Textile and Clothing Industry. Autex Research Journal, 18(4):337–347, 12 2018. ISSN 2300-0929. doi: 10.1515/aut-2018-0023. URL https://www. sciendo.com/article/10.1515/aut-2018-0023.
- [112] O. Lai. What is Fast Fashion?, 11 2021. URL https://earth.org/what-is-fast-fashion/.
- [113] E. K. Laitinen. A dynamic performance measurement system: evidence from small Finnish technology companies. *Scandinavian Journal of Management*, 18(1):65–99, 3 2002. ISSN 09565221. doi: 10.1016/S0956-5221(00)00021-X. URL https://linkinghub.elsevier.com/retrieve/pii/S095652210000021X.
- [114] D. M. Lambert and T. L. Pohlen. Supply Chain Metrics. The International Journal of Logistics Management, 12(1):1-19, 1 2001. ISSN 0957-4093. doi: 10.1108/09574090110806190. URL https://www.emerald.com/insight/content/doi/10.1108/09574090110806190/full/html.
- [115] K. Liu, X. Zeng, P. Bruniaux, J. Wang, E. Kamalha, and X. Tao. Fit evaluation of virtual garment tryon by learning from digital pressure data. *Knowledge-Based Systems*, 133:174–182, 10 2017. ISSN 09507051. doi: 10.1016/j.knosys.2017.07.007. URL https://linkinghub.elsevier.com/retrieve/ pii/S0950705117303234.
- [116] C. Lohman, L. Fortuin, and M. Wouters. Designing a performance measurement system: A case study. *European Journal of Operational Research*, 156(2):267–286, 7 2004. ISSN 0377-2217. doi: 10.1016/ S0377-2217(02)00918-9.
- [117] S. Mahar and P. D. Wright. In-Store Pickup and Returns for a Dual Channel Retailer. *IEEE Transac*tions on Engineering Management, 64(4):491-504, 11 2017. ISSN 0018-9391. doi: 10.1109/TEM.2017. 2691466. URL http://ieeexplore.ieee.org/document/7914623/.
- [118] R. Mangiaracina, A. Perego, S. Perotti, and A. Tumino. Assessing the environmental impact of logistics in online and offline B2C purchasing processes in the apparel industry. *International Journal of Logistics Systems and Management*, 23(1):98, 2016. ISSN 1742-7967. doi: 10.1504/IJLSM.2016.073300. URL http://www.inderscience.com/link.php?id=73300.
- [119] H. S. Matthews, E. Williams, T. Tagami, and C. T. Hendrickson. Energy implications of online book retailing in the United States and Japan. *Environmental Impact Assessment Review*, 22(5):493–507, 10 2002. ISSN 0195-9255. doi: 10.1016/S0195-9255(02)00024-0.
- [120] M. Mazziotta and A. Pareto. METHODS FOR CONSTRUCTING COMPOSITE INDICES: ONE FOR ALL OR ALL FOR ONE? 1. *Rivista Italiana di Economia Demografia e Statistica*, 2013.
- [121] S. McGuire. How much does it cost to rent a forklift?, 2020. URL https://hy-tekmaterialhandling. com/blog/how-much-it-cost-to-rent-a-forklift/#:~:text=Reach%20Truck%20Rental% 20Prices&text=Renting%20a%20reach%20truck%20with,%241%2C350%2D%241%2C500%20per% 20month.

- [122] McKinsey & Company. Scaling textile recycling in Europe-turning waste into value. Technical report, 7 2022.
- [123] A. J. Mehmet Sekip Altug, Tolga Aydinliyim. Managing Opportunistic Consumer Returns in Retail Operations. *Management Science*, 67(9), 2021. doi: https://doi.org/10.1287/mnsc.2020.3777.
- [124] M. Melacini, S. Perotti, M. Rasini, and E. Tappia. E-fulfilment and distribution in omni-channel retailing: a systematic literature review. *International Journal of Physical Distribution & Logistics Management*, 48(4):391–414, 5 2018. ISSN 0960-0035. doi: 10.1108/IJPDLM-02-2017-0101. URL https: //www.emerald.com/insight/content/doi/10.1108/IJPDLM-02-2017-0101/full/html.
- [125] P. Micheli and L. Mari. The theory and practice of performance measurement. Management Accounting Research, 25(2):147–156, 6 2014. ISSN 10445005. doi: 10.1016/j.mar.2013.07.005. URL https://linkinghub.elsevier.com/retrieve/pii/S104450051300070X.
- [126] S. Misra and D. Arivazhagan. Interactive Trial Room -A Solution to Reduce the Problem of Rampant Return of Sold Merchandise in Fashion E-Commerce Business. Indian Journal of Science and Technology, 10(16):1-4, 4 2017. ISSN 0974-5645. doi: 10.17485/ijst/2017/v10i16/106990. URL https://indjst.org/articles/ interactive-trial-room-a-solution-to-reduce-the-problem-of-rampant-return-of-sold-merchandise-in
- [127] S. K. Mukhopadhyay and R. Setaputra. A dynamic model for optimal design quality and return policies. *European Journal of Operational Research*, 180(3):1144–1154, 8 2007. ISSN 03772217. doi: 10.1016/J. EJOR.2006.05.016.
- [128] G. Munda. Choosing Aggregation Rules for Composite Indicators. Social Indicators Research, 109:337– 354, 9 2012. doi: https://doi.org/10.1007/s11205-011-9911-9.
- [129] G. Munda and M. Nardo. Noncompensatory/nonlinear composite indicators for ranking countries: a defensible setting. *Applied Economics*, 41(12):1513-1523, 5 2009. ISSN 0003-6846. doi: 10.1080/00036840601019364. URL http://www.tandfonline.com/doi/abs/10.1080/00036840601019364.
- [130] I. V. Muralikrishna and V. Manickam. Life Cycle Assessment. In *Environmental Management*, pages 57– 75. Elsevier, 2017. doi: 10.1016/B978-0-12-811989-1.00005-1. URL https://linkinghub.elsevier. com/retrieve/pii/B9780128119891000051.
- [131] S. S. Muthu, editor. Sustainability in the Textile Industry. Textile Science and Clothing Technology. Springer Singapore, Singapore, 2017. ISBN 978-981-10-2638-6. doi: 10.1007/978-981-10-2639-3. URL http://link.springer.com/10.1007/978-981-10-2639-3.
- [132] M. Nardo, M. Saisana, A. Saltelli, and S. Tarantola. State-of-the-Art Report on Simulation and Indicators. Technical report, Knowledge Economy Indicators, 2005. URL http://europa.eu.int/comm/ research/index\_en.cfmhttp://europa.eu.int/comm/research/fp6/ssp/kei\_en.htmhttp: //www.cordis.lu/citizens/kick\_off3.htmhttp://kei.publicstatistics.net/.
- [133] M. Nardo, M. Saisana, A. Saltelli, and S. Tarantola. Tools for Composite Indicators Building. Technical report, European Communities, 2005. URL https://www.researchgate.net/publication/ 277294848.
- [134] M. Nardo, M. Saisana, A. Saltelli, S. Tarantola, A. Hoffmann, E. Giovannini, and Organisation for Economic Co-operation and Development. *Handbook on constructing composite indicators : methodology and user guide*. OECD, 2008. ISBN 978-92-64-04345-9.
- [135] National Retail Federation. 2022 Retail Returns Rate Remains Flat at \$816 Billion, 12 2022. URL https://nrf.com/media-center/press-releases/ 2022-retail-returns-rate-remains-flat-816-billion.
- [136] A. Neely, M. Gregory, and K. Platts. Erratum. International Journal of Operations & Production Management, 25(12):1228-1263, 12 2005. ISSN 0144-3577. doi: 10.1108/01443570510633639. URL https://www.emerald.com/insight/content/doi/10.1108/01443570510633639/full/html.

- [137] Neil Z. Stern and Willard N. Ander. Greentailing and Other Revolutions in Retail. John Wiley & Sons, Inc., Hoboken, NJ, USA, 1 2012. ISBN 9781119197393. doi: 10.1002/9781119197393. URL http://doi. wiley.com/10.1002/9781119197393.
- [138] B. Ness, E. Urbel-Piirsalu, S. Anderberg, and L. Olsson. Categorising tools for sustainability assessment. *Ecological Economics*, 60(3):498–508, 1 2007. ISSN 09218009. doi: 10.1016/j.ecolecon.2006.07.023. URL https://linkinghub.elsevier.com/retrieve/pii/S0921800906003636.
- [139] D. W. Nutter, D.-S. Kim, R. Ulrich, and G. Thoma. Greenhouse gas emission analysis for USA fluid milk processing plants: Processing, packaging, and distribution. *International Dairy Journal*, 31(1):S57–S64, 4 2013. ISSN 09586946. doi: 10.1016/j.idairyj.2012.09.011. URL https://linkinghub.elsevier. com/retrieve/pii/S0958694612002063.
- [140] N. Olynec. 13 sustainability trends driving business in 2023, 1 2023. URL https://www.imd.org/ ibyimd/sustainability/13-sustainability-trends-driving-business-in-2023/.
- [141] Optoro. Making Returns Better for Customers, Retailers, and the Planet. Technical report, Optoro, 2022. URL https://4771362.fs1.hubspotusercontent-na1.net/hubfs/4771362/2022% 20Impact%20Report/Optoro\_2022%20Impact%20Report.pdf.
- [142] P. M. Orencio and M. Fujii. A localized disaster-resilience index to assess coastal communities based on an analytic hierarchy process (AHP). *International Journal of Disaster Risk Reduction*, 3(1):62–75, 3 2013. ISSN 22124209. doi: 10.1016/j.ijdrr.2012.11.006. URL https://linkinghub.elsevier.com/ retrieve/pii/S2212420912000428.
- [143] D. K. Pace. Modeling and Simulation Verification and Validation Challenges. *JOHNS HOPKINS APL TECHNICAL DIGEST*, 25(2), 2004.
- [144] M. Pakkar. A Hierarchical Aggregation Approach for Indicators Based on Data Envelopment Analysis and Analytic Hierarchy Process. *Systems*, 4(1):6, 1 2016. ISSN 2079-8954. doi: 10.3390/systems4010006. URL http://www.mdpi.com/2079-8954/4/1/6.
- [145] H. Palsson, F. Pettersson, and L. Winslott Hiselius. Energy consumption in e-commerce versus conventional trade channels - Insights into packaging, the last mile, unsold products and product returns. *Journal of Cleaner Production*, 164:765–778, 10 2017. ISSN 09596526. doi: 10.1016/j.jclepro.2017.06.242. URL https://linkinghub.elsevier.com/retrieve/pii/S0959652617314117.
- [146] Z. Pei and A. Paswan. Consumers' Legitimate and Opportunistic Product Return Behaviors in Online Shopping. *Journal of Electronic Commerce Research*, 19, 2018.
- [147] A. Pereira and R. Broed. Methods for Uncertainty and Sensitivity Analysis. Review and recommendations for implementation in Ecolego. Technical report, AlbaNova University Center, 1 2006.
- [148] R. Perera. The PESTLE Analysis. Nerdynaut, 9 2017. ISBN 9781790845323.
- [149] T. Phang, S. Way, D. Nabiha, and S. Jalaludin. Managing Environmental and Economic Performance: a review of theory and practice on performance measurement. *The International Journal of Accounting and Business Society*, 22(1), 2014.
- [150] N. Pollesch and V. Dale. Normalization in sustainability assessment: Methods and implications. *Ecological Economics*, 130:195–208, 10 2016. ISSN 09218009. doi: 10.1016/j.ecolecon.2016.06.018. URL https://linkinghub.elsevier.com/retrieve/pii/S0921800915305899.
- [151] PostNL. Annual Report Navigating a turbulent environment. Technical report, PostNL, The Hague, 2023. URL https://annualreport.postnl.nl/2022/.
- [152] PostNL. Locatiewijzer, 2023. URL https://www.postnl.nl/locatiewijzer/.
- [153] M. Reisi, L. Aye, A. Rajabifard, and T. Ngo. Transport sustainability index: Melbourne case study. *Ecological Indicators*, 43:288–296, 8 2014. ISSN 1470160X. doi: 10.1016/j.ecolind.2014.03.004. URL https://linkinghub.elsevier.com/retrieve/pii/S1470160X14000983.

- [154] J. M. Ries, E. H. Grosse, and J. Fichtinger. Environmental impact of warehousing: a scenario analysis for the United States. *International Journal of Production Research*, 55(21):6485–6499, 11 2017. ISSN 0020-7543. doi: 10.1080/00207543.2016.1211342. URL https://www.tandfonline.com/doi/full/ 10.1080/00207543.2016.1211342.
- [155] Rijksdienst voor Ondernemend Nederland. Energielabel C kantoren, 2018. URL https://www.rvo. nl/onderwerpen/wetten-en-regels-gebouwen/energielabel-c-kantoren.
- [156] Rijksoverheid. Bedragen minimumloon 2020. URL https://www.rijksoverheid.nl/ onderwerpen/minimumloon/bedragen-minimumloon/bedragen-minimumloon-2020.
- [157] D. S. Rogers, R. Lembke, and J. Benardino. A New Core Competency. *Supply Chain Management Review*, 5 2013. URL www.scmr.com.
- [158] H. Saarijärvi, U.-M. Sutinen, and L. C. Harris. Uncovering consumers' returning behaviour: a study of fashion e-commerce. *The International Review of Retail, Distribution and Consumer Research*, 27 (3):284–299, 5 2017. ISSN 0959-3969. doi: 10.1080/09593969.2017.1314863. URL https://www.tandfonline.com/doi/full/10.1080/09593969.2017.1314863.
- [159] T. L. Saaty. How to make a decision: The analytic hierarchy process. European Journal of Operational Research, 48(1):9–26, 9 1990. ISSN 03772217. doi: 10.1016/0377-2217(90)90057-I. URL https://linkinghub.elsevier.com/retrieve/pii/037722179090057I.
- [160] T. L. Saaty. Decision making with the analytic hierarchy process. *Int. J. Services Sciences*, 1(1):83–98, 2008.
- [161] M. Sadegh Pakkar. Using data envelopment analysis and analytic hierarchy process to construct composite indicators. *Journal of Applied Operational Research*, 6(3):174–187, 2014. ISSN 1927-0089. URL www.tadbir.ca.
- [162] F. Samara, S. Ibrahim, M. E. Yousuf, and R. Armour. Carbon Footprint at a United Arab Emirates University: GHG Protocol. *Sustainability*, 14(5):2522, 2 2022. ISSN 2071-1050. doi: 10.3390/su14052522. URL https://www.mdpi.com/2071-1050/14/5/2522.
- [163] T. Schmitz. Critical analysis of carbon dioxide emissions in a comparison of e-commerce and traditional retail. *Journal of Applied Leadership and Management*, 8:72–89, 2020. ISSN 2194-9522. URL https://www.journal-alm.org/article/view/21239.
- [164] A. H. Schrotenboer, S. Wruck, K. Jan Roodbergen, M. Veenstra, and A. S. Dijkstra. International Journal of Production Research Order picker routing with product returns and interaction delays Order picker routing with product returns and interaction delays. *International Journal of Production Research*, 55(21):6394–6406, 2017. ISSN 0020-7543. doi: 10.1080/00207543.2016.1206982. URL https: //www.tandfonline.com/action/journalInformation?journalCode=tprs20.
- [165] K. Selvi and R. Majumdar. Six Sigma-Overview of DMAIC and DMADV. 2014. ISSN 2319-6386. URL http://cert.asq.org/cert/six-sigma/index.
- [166] T. Semba, Y. Sakai, M. Ishikawa, and A. Inaba. Greenhouse Gas Emission Reductions by Reusing and Recycling Used Clothing in Japan. *Sustainability*, 12(19):8214, 10 2020. ISSN 2071-1050. doi: 10.3390/ su12198214. URL https://www.mdpi.com/2071-1050/12/19/8214.
- [167] R. K. Singh, H. Murty, S. Gupta, and A. Dikshit. Development of composite sustainability performance index for steel industry. *Ecological Indicators*, 7(3):565–588, 7 2007. ISSN 1470160X. doi: 10.1016/j.ecolind.2006.06.004. URL https://linkinghub.elsevier.com/retrieve/pii/ S1470160X06000550.
- [168] R. K. Singh, H. Murty, S. Gupta, and A. Dikshit. An overview of sustainability assessment methodologies. *Ecological Indicators*, 9(2):189–212, 3 2009. ISSN 1470160X. doi: 10.1016/j.ecolind.2008.05.011. URL https://linkinghub.elsevier.com/retrieve/pii/S1470160X08000678.
- [169] E. Smeets, R. Weterings, P. Bosch, M. Büchele, and D. Gee. Environmental indicators: Typology and overview. Technical report, EEA, 1999. URL http://europa.eu.int.

- [170] Solene Rauturier. Everything You Need to Know About Waste in the Fashion Industry, 2 2022. URL https://goodonyou.eco/waste-luxury-fashion/.
- [171] Statista. Online vs. offline purchases by category in the Netherlands in 2022, 2022. URL Onlinevs. offlinepurchasesbycategoryintheNetherlandsin2022.
- [172] Statista. Fashion worldwide, 2 2023. URL https://www.statista.com/outlook/dmo/ecommerce/ fashion/worldwide?currency=EUR.
- [173] Statista Research Department. Warehouse primary rent costs in Spain 2021, 6 2022. URL https://www.statista.com/statistics/527906/ warehouse-primary-rent-cost-logistics-market-spain-europe/#:~:text=The% 20annual%20primary%20rent%20costs,per%20square%20meter%20in%20Madrid.
- [174] J. R. Stock and J. P. Mulki. Product returns processing: an examination of practices of manufacturers, wholesalers/distributors, and retailers. *Journal of Business Logistics*, 30(1):33-62, 3 2009. ISSN 07353766. doi: 10.1002/j.2158-1592.2009.tb00098.x. URL https://onlinelibrary.wiley.com/ doi/10.1002/j.2158-1592.2009.tb00098.x.
- B. Stöcker, D. Baier, and B. M. Brand. New insights in online fashion retail returns from a customers' perspective and their dynamics. *Journal of Business Economics*, 91(8):1149–1187, 10 2021. ISSN 0044-2372. doi: 10.1007/s11573-021-01032-1. URL https://link.springer.com/10.1007/s11573-021-01032-1.
- [176] Y. Su, H. Duan, Z. Wang, G. Song, P. Kang, and D. Chen. Characterizing the environmental impact of packaging materials for express delivery via life cycle assessment. *Journal of Cleaner Production*, 274: 122961, 11 2020. ISSN 09596526. doi: 10.1016/j.jclepro.2020.122961. URL https://linkinghub. elsevier.com/retrieve/pii/S0959652620330067.
- [177] N. Suchek, C. I. Fernandes, S. Kraus, M. Filser, and H. Sjögrén. Innovation and the circular economy: A systematic literature review. *Business Strategy and the Environment*, 30(8):3686–3702, 12 2021. ISSN 0964-4733. doi: 10.1002/bse.2834. URL https://onlinelibrary.wiley.com/doi/10.1002/bse.2834.
- [178] A. Tait. Buy. Return. Repeat ... What really happens when we send back unwanted clothes?, 3 2023. URL https://www.theguardian.com/global-development/2023/mar/31/ what-happens-when-we-send-back-unwanted-clothes.
- [179] A. Tajbakhsh and E. Hassini. Performance measurement of sustainable supply chains: a review and research questions. *International Journal of Productivity and Performance Management*, 64(6):744– 783, 7 2015. ISSN 1741-0401. doi: 10.1108/IJPPM-03-2013-0056. URL https://www.emerald.com/ insight/content/doi/10.1108/IJPPM-03-2013-0056/full/html.
- [180] B. Talukder, K. W. Hipel, and G. W. vanLoon. Developing Composite Indicators for Agricultural Sustainability Assessment: Effect of Normalization and Aggregation Techniques. *Resources*, 6(4):66, 11 2017. ISSN 2079-9276. doi: 10.3390/resources6040066. URL http://www.mdpi.com/2079-9276/6/4/66.
- [181] B. H. Thacker, S. W. Doebling, F. M. Hemez, M. C. Anderson, J. E. Pepin, and E. A. Rodriguez. Concepts of Model Verification and Validation. Technical report, Los Alamos National Laboratory, 2004.
- [182] T. S. Thorisdottir and L. Johannsdottir. Sustainability within Fashion Business Models: A Systematic Literature Review. Sustainability, 11(8):2233, 4 2019. ISSN 2071-1050. doi: 10.3390/su11082233. URL https://www.mdpi.com/2071-1050/11/8/2233.
- [183] X. Tian and J. Sarkis. Emission burden concerns for online shopping returns. Nature Climate Change, 12(1):2-3, 1 2022. ISSN 1758-678X. doi: 10.1038/s41558-021-01246-9. URL https://www.nature. com/articles/s41558-021-01246-9.
- [184] C. U-Dominic and H. Godwin. A Review of Six Sigma and the Shared Relationship with Knowledge Management (KM) Discipline. Archives of Current Research International, 13(2):1–15, 3 2018. ISSN 24547077. doi: 10.9734/ACRI/2018/38262. URL https://journalacri.com/index.php/ACRI/ article/view/223.
- [185] ULine. 9 x 9 x 9" Corrugated Boxes. URL https://www.uline.com/Product/Detail/S-4094/ Corrugated-Boxes-200-Test/9-x-9-x-9-Corrugated-Boxes.
- [186] UPS. Five things to know about UPS's latest Sustainability Report, 8 2021. URL https://about.ups.com/us/en/social-impact/environment/climate/ 2020-ups-corporate-sustainability-report.html.
- [187] S. van der Meulen, T. Grijspaardt, W. Mars, W. van der Geest, A. Roest-Crollius, and J. Kiel. Cost Figures for Freight Transport-final report. Technical report, Panteia, 2023.
- [188] P. Van Loon, L. Deketele, J. Dewaele, A. McKinnon, and C. Rutherford. A comparative analysis of carbon emissions from online retailing of fast moving consumer goods. In *Journal of Cleaner Production*, volume 106, pages 478–486. Elsevier Ltd, 11 2015. doi: 10.1016/j.jclepro.2014.06.060.
- [189] T. Vanelslander, L. Deketele, and D. Van Hove. Commonly used e-commerce supply chains for fast moving consumer goods: comparison and suggestions for improvement. *International Journal of Logistics Research and Applications*, 16(3):243–256, 6 2013. ISSN 1367-5567. doi: 10.1080/13675567.2013.813444. URL http://www.tandfonline.com/doi/abs/10.1080/13675567.2013.813444.
- [190] L. G. Vargas. An overview of the analytic hierarchy process and its applications. European Journal of Operational Research, 48(1):2–8, 9 1990. ISSN 03772217. doi: 10.1016/0377-2217(90)90056-H. URL https://linkinghub.elsevier.com/retrieve/pii/037722179090056H.
- [191] R. Velazquez and S. M. Chankov. Environmental Impact of Last Mile Deliveries and Returns in Fashion E-Commerce: A Cross-Case Analysis of Six Retailers. In 2019 IEEE International Conference on Industrial Engineering and Engineering Management (IEEM), pages 1099–1103. IEEE, 12 2019. ISBN 978-1-7281-3804-6. doi: 10.1109/IEEM44572.2019.8978705. URL https://ieeexplore.ieee.org/ document/8978705/.
- [192] D. Vigo, J.-F. Cordeau, G. Laporte, and M. W. P. Savelsbergh. Vehicle Routing. volume 14. Transportation, handbooks in operations and management science, 1 2007. URL https://www.researchgate.net/ publication/233843551.
- [193] Vinted. Pakketdetails verzendopties, 2023. URL https://www.vinted.nl/help/ 51-het-juiste-pakketformaat-kiezen.
- [194] M. Von Zahn, K. Bauer, C. Mihale-Wilson, J. Jagow, M. Speicher, and O. Hinz. The Smart Green Nudge: Reducing Product Returns through Enriched Digital Footprints & Causal Machine Learning. 11 2022. URL https://ssrn.com/abstract=4262656.
- [195] L. Wang, E. Elahi, Y. Zhou, L. Wang, and S. Zhang. A Review of Packaging Materials' Consumption Regulation and Pollution Control. *Sustainability*, 14(23):15866, 11 2022. ISSN 2071-1050. doi: 10.3390/ su142315866. URL https://www.mdpi.com/2071-1050/14/23/15866.
- [196] P. A. Warhurst. Mining, Minerals and Sustainable Development Sustainability Indicators and Sustainability Performance Management. Technical report, IIED and WBCSD, 3 2002. URL www.wbs. warwick.ac.uk/ccu/.
- [197] West River Conveyors. Maximizing the lifespan of your belt conveyor. URL https://www. westriverconveyors.com/blog/maximizing-the-lifespan-of-your-conveyor-belts/#:~: text=TYPICAL%20CONVEYOR%20BELT%20LIFESPAN,is%20wrong%20for%20the%20application.
- [198] S. Widya Yudha, B. Tjahjono, and A. Kolios. A PESTLE Policy Mapping and Stakeholder Analysis of Indonesia's Fossil Fuel Energy Industry. *Energies*, 11(5):1272, 5 2018. ISSN 1996-1073. doi: 10.3390/ en11051272. URL http://www.mdpi.com/1996-1073/11/5/1272.
- [199] A. Wiese, W. Toporowski, and S. Zielke. Transport-related CO2 effects of online and brick-and-mortar shopping: A comparison and sensitivity analysis of clothing retailing. *Transportation Research Part D: Transport and Environment*, 17(6):473–477, 8 2012. ISSN 13619209. doi: 10.1016/j.trd.2012.05.007. URL https://linkinghub.elsevier.com/retrieve/pii/S1361920912000521.

- [200] S. L. Wood. Remote Purchase Environments: The Influence of Return Policy Leniency on Two-Stage Decision Processes. *Journal of Marketing Research*, 38(2):157–169, 5 2001. ISSN 0022-2437. doi: 10.1509/ jmkr.38.2.157.18847. URL http://journals.sagepub.com/doi/10.1509/jmkr.38.2.157.18847.
- [201] Y. Xu and W. Feng. Develop a cost model to evaluate the economic benefit of remanufacturing based on specific technique. Technical report. URL http://www.journalofremanufacturing.com/ content/4/1/4.
- [202] R.-N. Yan, S. Diddi, and B. Bloodhart. Predicting clothing disposal: The moderating roles of clothing sustainability knowledge and self-enhancement values. *Cleaner and Responsible Consumption*, 3:100029, 12 2021. ISSN 26667843. doi: 10.1016/j.clrc.2021.100029. URL https://linkinghub. elsevier.com/retrieve/pii/S2666784321000231.
- [203] Q. Zeng. Development of A Composite Indicator for Measuring Company Performance from Economic and Environmental Perspectives: A Study on Motor Vehicle Manufacturers, 2020. URL https://repository.tudelft.nl/islandora/object/uuid: 75d0ae4c-afbc-4b7a-814d-8bbb3642b95f?collection=research.
- [204] Q. Zeng, W. Beelaerts van Blokland, S. Santema, and G. Lodewijks. Benchmarking company performance from economic and environmental perspectives. *Benchmarking: An International Journal*, 27(3):1127–1158, 12 2019. ISSN 1463-5771. doi: 10.1108/BIJ-05-2019-0223. URL https://www. emerald.com/insight/content/doi/10.1108/BIJ-05-2019-0223/full/html.
- [205] N. Zhang, P. Zhou, and C.-C. Kung. Total-factor carbon emission performance of the Chinese transportation industry: A bootstrapped non-radial Malmquist index analysis. *Renewable and Sustainable Energy Reviews*, 41:584–593, 1 2015. ISSN 13640321. doi: 10.1016/j.rser.2014.08.076. URL https://linkinghub.elsevier.com/retrieve/pii/S136403211400762X.
- [206] L. Zhou, H. Tokos, D. Krajnc, and Y. Yang. Sustainability performance evaluation in industry by composite sustainability index. *Clean Technologies and Environmental Policy*, 14(5):789–803, 10 2012. ISSN 1618-954X. doi: 10.1007/s10098-012-0454-9. URL http://link.springer.com/10.1007/s10098-012-0454-9.
- [207] P. Zhou and L. P. Zhang. Composite Indicators for Sustainability Assessment: Methodological Developments. In *Energy, Environment and Transitional Green Growth in China*, pages 15–36. Springer Singapore, Singapore, 2018. doi: 10.1007/978-981-10-7919-1{\\_}2. URL http://link.springer.com/10. 1007/978-981-10-7919-1\_2.
- [208] P. Zhou, B. W. Ang, and K. L. Poh. Comparing aggregating methods for constructing the composite environmental index: An objective measure. *Ecological Economics*, 59(3):305–311, 9 2006. ISSN 0921-8009. doi: 10.1016/J.ECOLECON.2005.10.018.
- [209] Q. Zhu, J. Sarkis, and Y. Geng. Green supply chain management in China: Pressures, practices and performance. International Journal of Operations & Production Management, 23(5):449-468, 1 2004. URL https://www.researchgate.net/publication/303140100\_Green\_supply\_chain\_ management\_in\_China\_Pressures\_practices\_and\_performance.

## A

## Scientific research paper

The scientific paper starts on the next page.

#### Understanding the environmental impact of product returns in the fashion industry The development of a Composite Indicator to provide insights into the environmental and economic impact of product returns for fashion companies

#### Tijn Veen<sup>1</sup>

<sup>1</sup>MSc Candidate Delft University of Technology Dr. J.M. Vleugel, Dr. W.W.A. Beelaerts van Blokland, Prof. Dr. R.R. Negenborn, D. Visser, M. Rustemeijer, M. Natani

#### Abstract

**Purpose** - With the ongoing rise of product returns in the fashion industry, exacerbating its GHG footprint when it should be lowered, this paper aims to provide a measurement tool for companies to improve the environmental impact of product returns while considering the economic costs. Companies do not have an understanding of the impact of these returns on their performances, resulting from a gap in research where an accurate measurement method lacks that includes all activities and return methods.

**Design/methodology/approach** - The returns process and the return methods were analyzed using a literature study and interviews with experts, resulting in a set of environmental and economic indicators. These indicators were aggregated in a Composite Indicator (CI) by rescaling, analytical hierarchy process, and multiplicative aggregation techniques. A benchmark for target performance was developed based on reduction targets of the European Commission, using rescaling, equal weighting, and multiplication and integrated into the CI. An uncertainty- and sensitivity analysis confirmed the model's robustness. A case study of a fashion retailer verified the model. A scenario analysis based on the PESTLE framework was used to validate the CI and benchmark. A company's performance measured was compared against the verified benchmark by calculating the performance gap.

**Findings** - The environmental and economic impact of the return methods for the base case was measured. The impact of the return methods was compared against the benchmark, which shows that home pick-ups have the smallest performance gap with the benchmark for this retailer. Instore returns have the largest gap with the benchmark, indicating worse company performance and lower compliance with the EC targets. The scenario analysis showed how companies could use this CI to compare return improvement strategies with each other and provided potential solutions for companies to improve their performance.

**Research limitations/implications** - For fashion companies, this research provides a tool to measure, analyze, and improve the environmental impact of returns on their performances with EC reduction targets while considering their costs and providing transparency. Customers and governments are provided with transparent information on what- and how the environmental impact of returns is measured by a company. Governments can use the CI and benchmark proposed in this paper as a transparent tool for environmental reporting and compare the performances of companies sector-wide. For academics, this research provides multiple research opportunities.

**Originality/value** - This paper contributes by creating a new CI for company performance measurement for fashion companies on the environmental and economic impact of product returns. A new measurement method that assesses the environmental and economic impact of the return activities is developed and gives insight into the differences between the return methods. Also, a new benchmark expressing target performance is developed and incorporated in this CI based on reduction targets of the EC to combat climate change.

Keywords: Composite indicator, product returns, fashion industry, environmental performance, economic performance, performance measurement, target benchmark

#### ACRONYMS

- **3PL** Third Party Logistics Provider
- **AHP** Analytical Hierarchy Process
- **CI** Composite Indicator
- **CR** Consistency Ratio
- **UA** Uncertainty Analysis
- SA Sensitivity Analysis

- EC European CommissionEF Emisison factor
- **EW** Equal Weights

**GHG** Greenhouse Gas

- **EU** European Union
- ${\bf LCA}~$  Life Cycle Assessment
- **LIN** Linear aggregation
- ${\bf GME}$  Geometric mean

#### I. INTRODUCTION

The fashion industry is a worldwide driver for economic growth [34]. With a growth rate of 9.93% per year (CAGR 2023-2027) [67], further growth in demand is expected in the years to come, particularly within ecommerce [7, 34, 42]. At the same time, the fashion industry's impact ranks the fourth largest on climate change within the European Union (EU), and action is needed to lower its environmental footprint [24, 31, 71]. As part of the Green Deal, the European Commission (EC) plans on mandatory environmental reporting, in which companies need to provide transparency on their performances [23]. Most research and efforts focus on sustainability practices in the forward value chain [3]. Less attention is paid to returns, although these have become an increased problem in this industry, particularly [6, 11, 72].

High return rates are observed, which are expected to increase further due to the industry's growth [1, 11, 68]. Besides economic losses for companies, these returns create additional Greenhouse Gas (GHG) emissions and waste, thereby negatively affecting society and the environmental footprint of companies [11, 18, 59, 68, 75]. The EC recognizes the environmental impact of returned products as a widespread problem within the EU [27], and actions should be taken to combat climate change [24, 75].

To do so, the GHG emissions created by returns should be reduced by improving the return process and lowering the number of returns [11, 24], leading to environmental gains for society and potential cost savings for companies [77].

Companies need an understanding of the environmental and economic impact of returns and how this affects their performances. With this, companies can design effective strategies to improve their environmental impact while remaining economically viable [8, 11, 71, 80]. However, companies do not yet have the tool to analyze, assess, and compare the impact of product returns and return methods on their environmental and economic performances, as an accurate measurement method lacks [11, 72]. In the scarce attempts to assess product returns, not all relevant logistic activities involved in a return are considered, assess not all GHG emissions, or consider not all return methods available nowadays. As a result, companies cannot measure and monitor their performance, identify areas of improvement, and design effective return improvement efforts to improve the impact of returns. Also, companies lack insights into how far they are in reaching the goals of the EC to combat climate change, although they need to comply with this [21, 23, 31]. Moreover, as this knowledge is lacking for companies, companies cannot provide information on the environmental impact of their returns to customers or governments. This leaves reduction potential unexplored, as informed customers will be more likely to make more sustainable decisions for their choice of return method [4, 7, 80] and limit opportunistic returns, thereby returning fewer products [8, 77]. This is expected to lead to both environmental gains for society and potential cost savings for a company.

The research question answered in this paper is: How can a Composite Indicator (CI) be constructed to provide insight into the environmental and economic performance of product returns against a target benchmark for fashion companies and help improve their sustainability?

This research has developed a new performance measurement tool for fashion companies to measure, analyze, and evaluate returns' environmental and economic impact. A new CI was constructed to measure the performance of returns on environmental and economic dimensions [82]. The CI included all return activities, considered the differences in the return methods, and assessed the environmental impact on a company's performance, thereby filling the research gap. The designed CI expanded the field of company performance measurement by assessing both environmental and economic performance, which was not done frequently yet [81]. In addition, a new benchmark based on GHG reduction targets of the EC for 2030 was designed, developed, and integrated into this CI. With this, a transparent and sectorwide tool was developed that companies can use to see how closely their performances comply with EC targets. Furthermore, governments can use this transparent tool to compare companies' performances with each other, as it is known what- and how the impact is measured, and evaluate their progress as part of the prospected EU mandatory environmental reporting. Next, governments can use this new benchmark to set more concrete targets to combat climate change sector-wide.

First, the returns process is presented using a literature review supplemented by interviews with industry experts in section II. CI are tools to measure performance, which supports companies in measuring and tracking progress and identifying potential areas for improvement in processes [40, 47, 51]. The findings on the return process are combined with literature and reports for constructing CIs and benchmarking, which are used to construct the CI and the benchmark in section III. Next, the CI and benchmark are applied to a case study, and the results are presented in section IV. Next, a discussion on the model is given in section V. Last, conclusions and recommendations for further research are made in section VI.

#### **II. LITERATURE REVIEW**

First, the process of returns is outlined, followed by a review of the environmental impact of return activities and the economic impact.

#### A. How the process of returns works

This research focuses on the whole process of a return and considers all logistic activities that are crucial in analyzing the environmental and economic impact. The boundary defined for the process of product returns in this research starts from the customer requesting a return until the recovery and disposition of the product is executed [6, 15]. Four phases are distinguished: return preparation, collection, processing and handling of returns, and product recovery and disposition, which is visualized in Figure 1.

Fashion companies use in-store returns, collection point returns, or home pick-up returns. The availability of the return methods for a customer depends on whether the retailer offers this service [5]. Packaging is needed for safe transport when a collection point or home pick-up is selected. Product transport occurs to the final recovery point, depending on the return method. This final recovery point can be the brick-and-mortar store, where it will be restocked again, or the warehouse [6, 15]. If the product's final destination is the warehouse, it will be recovered. The choice for a recovery treatment is based on several criteria, including the cost, the product market value, added value of recovery, and the type of product [44].



FIG. 1: Return processes and sub-activities, adapted from [14]

### B. How a product return creates environmental impact

The GHG Protocol and the Life Cycle Assessment (LCA) are presented as assessment tools for environmental impact in literature. A LCA is used when measuring a more comprehensive set of environmental factors and evaluating the overall effect of a product throughout its entire life cycle [54]. The GHG Protocol is more useful when measuring specific activities and is commonly used as an accounting tool at the company levels for activities in its value chain [35]. This study aims to measure the GHG emissions created by all activities in the return process, making the GHG Protocol more suitable as a measurement tool to assess the environmental impact of product returns for this research.

The GHG Protocol is a recognized tool by the EC

for measuring and reporting GHG emissions and provides a standardized methodology by using Emission factor (EF)s that incorporate the GHG emissions per activity or process [35, 36].

The following sources of emissions are identified from the literature: packaging emissions, transport emissions, warehouse and equipment emissions, brick-and-mortar store emissions, and product recovery and disposal emissions.

Packaging emissions are created when a customer returns by home pick-up or collection point or when the product gets transported from the brick-and-mortar store to the warehouse. Packaging emissions consist of usage and waste emissions. The magnitude of the usage emissions depends on the type of packaging material, the mass of the packaging, and whether or not a customer reuses the packaging for their return [36, 69]. The packaging waste emissions depend on the waste treatment chosen, the type of packaging material, and the mass [36, 69].

Transport consists of short- and long-haul activity [36]. The transport emissions depend on how a product is returned, as for each return method, a different journey is considered [13, 19, 49, 59]. The transport emissions created by in-store returns depend firstly on the trip from the customers' location to the store and back [9, 19, 49, 79], and optionally the transport activity from the brick-and-mortar store to the warehouse [14]. This depends on the distance driven, product mass, and vehicle characteristics [36]. For home pick-ups, the whole pick-up journey needs to be considered. Based on the locations of the pick-up requests and the distribution center, a vehicle journey is constructed from the pick-up point to the initial recovery point [13]. After the vehicle arrives at the recovery point, the products are redistributed to their final recovery destination (i.e., the warehouse or brick-and-mortar store), creating additional transport emissions [36]. The transport emissions for collection point returns can be measured by distinguishing three transport activities. First, from the customer to the collection point; second, the journey of the Third Party Logistics Provider (3PL) from the collection point to the distribution center, where the journey distance depends on the location of the collection points served in that journey and the distribution center. Third, the product needs transportation to its final destination, intensifying the transport emissions of the collection-point return.

Warehousing includes multiple activities (see Figure 1), intensifying the environmental impact [49, 63]. The main contributors to warehouse emissions are the fuel and electricity consumed [49, 63]. In addition, internal transport and the energy used by warehouse equipment for handling facilities must be considered, depending on the equipment type used and operation time [63]. If products are handled in-store, the environmental impact created here must be considered [36]. This depends on the store size, location, and energy rating [49].

Products can be resold, repaired, recycled, remanufac-

tured, or sent to incineration or landfills [2, 15, 32, 44]. The chosen recovery or disposition strategy affects the GHG emissions of a product [6, 28, 36].

#### C. How a product return creates costs

The economic impact of product returns is measured based on the activities performed by a company to perform a return and the return methods [48]. According to an industry expert, this activity-based approach is commonly used in practice.

Costing of return-related activities includes transportation, labor, handling equipment, disposition, and storage of the products and packaging [39, 46], and is confirmed by two experts.

Depending on how the product is returned, transport costs are created. In-store returns only generate transport costs for a company when the decision is not to resell the product in-store. Also, the transport is usually outsourced to a 3PL but can be self-executed, affecting the cost structure [14]. The main components of transport costs are the distance, the vehicle load, the mass of the product, and fixed- and variable modality costs [15, 39, 73].

Warehousing and equipment costing depends on the space and resources needed, storage time, and leasing costs [14]. Operating costs of the physical store should be considered for returns handled in-store. According to an expert, the most significant cost factor is the leasing costs of the stores. Returns are highly labor intensive, and labor costs are incurred in brick-and-mortar stores and warehouses. Activities include receiving, inspecting, and restocking products [39]. The product return labor cost on an annual basis is the sum of these fees [39]. Additionally, this metric should include any overhead expenses and investments like training or other employee benefits in this metric [14].

Recovery costs are included for both packaging materials and the product itself. These recovery costs per item are calculated by the mass of the item, the recovery charge, and optional transport charges, depending on the country of disposal [20, 29]. Governmental charges are associated with landfill and incineration, which come at the cost of a company [41].

#### **III. CONSTRUCTION OF THE CI**

A new CI was constructed for a multi-dimensional assessment of product returns on the performance of fashion companies, including the development and integration of a target benchmark. The CI should give a complete assessment of the impact of returns, including all return methods and activities; it needs to be straightforward to use by companies and transparent; the results should be easy to be communicated; and companies should know how far they are in reaching certain goals, thereby increasing the value of this measurement tool [82]. The construction was split into four phases, as can be seen in Figure 2. A case study was used to complete phase 4.



FIG. 2: Conceptual model construction CI with benchmark, source: (Author)

#### A. Phase 1. Theoretical framework with variables

The theoretical framework for assessing product returns is presented in Table I. The final selection avoided double-counting of emissions or costs between the indicators and fulfilled the objective and scope of the research. Variables 1-5 are the environmental set of indicators, and variables 6-10 are the indicators for the economic dimension. All variables have a negative indicator variability, which indicates that a lower value of that respective variable is preferred to obtain higher performance [62, 83].

 

 TABLE I: Theoretical framework of the variables, source: (Author)

	Description	Unit of measurement
$V_1$	Packaging emissions	$\frac{kgCO2e}{returned \ product}$
$V_2$	Transport emissions	$\frac{kgCO2e}{returned \ product}$
$V_3$	Warehouse and equipment	kgCO2e
13	emissions	returned product
$V_4$	Product recovery emissions	$\frac{kgCO2e}{returned \ product}$
$V_5$	Brick-and-mortar emissions	$\frac{kgCO2e}{returned \ product}$
$V_6$	Transport costs	euro returned product
$V_7$	Labor costs	$\frac{euro}{returned \ product}$
$V_8$	Warehouse and equipment	euro
V8	costs	returned product
$V_9$	Recovery costs	euro returned product
$V_{10}$	Brick-and-mortar costs	$\frac{euro}{returned \ product}$

#### B. Phase 2. Development of the CI

The CI was constructed using several techniques that were selected based on the requirements for this CI, its objective, and scope [50, 56]. For aggregation, the multiplication technique was used. This technique is easy to interpret and allows for partial compensability between the indicators, meaning that, to some extent, trade-offs can be made between the variables [52]. A partial compensatory approach was most appropriate rather than complete compensability considering the requirements and theoretical framework in this assessment [50, 52, 53]. The weights were constructed with Analytical Hierarchy Process (AHP), which allows for relative weighting of the variables. Also, it allowed decision-makers to make company-specific trade-offs between the variables but provided a statistical method to construct the weights [38, 55]. As AHP was selected, [50] suggests using either a ranking, standardization, or rescaling transformation method for normalization. The ranking was unsuitable in this study as it cannot provide differences in performances [70], z-score was unfeasible with multiplication aggregation, so rescaling was selected [56]. Rescaling is simple and can widen ranges variable intervals but is less capable of maintaining proportionally [56].

#### 1. Measurement of the variables

Variables were measured using methods derived from literature and interviews.

The emissions of packaging waste  $(V_1)$  were determined by the total mass of packaging material  $m_m$  multiplied by the waste-treatment-specific EF, see Equation 3.1 [36, 69].

$$V_1 = m_m * EF_{\rm wt} \tag{3.1}$$

The transport emissions were calculated by the total distance driven with a modality [36, 37], represented in Equation 3.2. Here, d represents the distance,  $m_p$  the mass, and EF an EF based on the modality [19, 49, 79]. A constant fuel consumption per traveled kilometer is assumed [13]. For the in-store vehicle journey and the collection point journey, a general vehicle routing problem with a single warehouse was assumed to be the underlying transport model, in which several conditions were respected [10, 13, 16, 76]. The capacity of the vehicle or time limitations limits the vehicle's journey. Additionally, it was assumed that a customer's request must be served in one visit [10, 15, 76]. The vehicle departs at the warehouse and ends its journey at the same location, of which the origin and destination were represented by nodes  $\theta$  and n+1, and n the demand nodes. For home pick-ups, the vehicle journey was calculated by Equation 3.3, with  $i, j \in P$ , with P the set of pick-up points [10, 13, 16, 76]. To calculate the transport emissions of a collection point journey, Equation 3.4 was used, in which the distance was measured between the collection points,  $i, j \in C$ , with C [13]. The return requests were represented by demand nodes n, and the arcs represent the actual distance  $d_{ij}$  from node i to node j.

$$V_2 = \sum d * m_p * EF_m \tag{3.2}$$

$$V2_{pick-up} = \left(\sum_{(i,j)\in P} d_{ij} + d_{0,i} + d_{j,n+1}\right) * m_p * EF_m \quad (3.3)$$

$$V2_{collection-point} = \left(\sum_{(i,j)\in C} d_{ij} + d_{0,i} + d_{j,n+1}\right) * m_p * EF_m$$
(3.4)

An environmental impact study of warehousing of [63] links the size of the warehouse to energy consumption and the geographical location of the warehouse. The warehouse and equipment emissions were calculated by Equation 3.5, where A is the area dedicated to returns at the warehouse,  $EF_{wh}$  the EF for warehousing,  $c_{eq}$  the costs of equipment for a year,  $EF_{eq}$  the EF for equipment usage, and n the number of products handled at the warehouse over the considered time period [35, 36].

$$V_3 = \frac{(A_w * EF_{wh}) + (c_{eq} * EF_{eq})}{n}$$
(3.5)

The calculation of GHG emissions for this phase of the return process was defined for the chosen recovery method [6]. Recovery emissions are measured with Equation 3.6, with an EF for the waste treatment chosen and  $m_p$  the mass of the product [28, 36].

$$V_4 = m_p * EF_{wt} \tag{3.6}$$

Based on the average time a product spends at the store, and the number of products handled in the store over time, the in-store emissions allocated to a product return can be determined [36]. This is captured in Equation 3.7, where  $A_s$  is the area of the brick-and-mortar store,  $EF_s$  a store-specific emission factor that expresses the energy usage in GHG emissions, and n the number of products handled in store over a considered time period [49].

$$V_5 = \frac{A_s * EF_s}{n} \tag{3.7}$$

The transport costs were calculated with Equation 3.8, where d is the distance driven,  $m_p$  the product mass, and  $c_t$  includes modality-specific factors expressed per ton-km distance driven [15, 39, 73].

$$V_6 = \sum d * m_p * c_t \tag{3.8}$$

The calculation of [39] was adapted to Equation 3.9, which expresses the total labor costs per returned item. Here,  $c_{\rm res}$  is the costs for inspection and restocking per product,  $c_{\rm rec}$  is the cost of receiving per item, and  $c_{\rm overh}$  is the overhead expenses per product [14, 39].

$$V_7 = c_{\rm res} + c_{\rm rec} + c_{\rm overh} \tag{3.9}$$

The warehouse and equipment costs are determined with Equation 3.10, where  $c_{wh}$  are the leasing cost of the warehouse over a considered period [14].

$$V_8 = \frac{(A_w * c_{wh}) + c_{eq}}{n}$$
(3.10)

The recovery costs were calculated by Equation 3.11, where both the product return and packaging material are considered. The costs for recovery depend on the chosen treatment of waste and the country where it is processed [12, 20, 29].

$$V_9 = (m_p * c_{rec,p}) + (m_m * c_{rec,m})$$
(3.11)

Operating costs of the physical brick-and-mortar store should be considered for in-store returns. From interviews, it follows that the most significant cost factor within this metric is the leasing costs of the store. The costs per unit product can be estimated by obtaining the total number of products handled at the store over a certain period, as presented in Equation 3.12. Here,  $c_s$  are the costs per m2,  $A_s$  the total store area, and n is the number of products handled in store over a time period.

$$V_{10} = \frac{c_s * A_s}{n} \tag{3.12}$$

#### 2. Normalization

The indicators are normalized so they can be compared regardless of their different unit of measurement. This is done by rescaling using Equation 3.13 [81]. Here,  $min_i(x_{ij})$  and  $max_i(x_{ij})$  are the minimum and maximum values for indicator j considering all return methods i[70].

$$I_{ij} = 1 + \frac{max_i(x_{ij}) - x_{ij}}{max_i(x_{ij}) - min_i(x_{ij})}$$
(3.13)

Where:

- *i* The return method  $i, i = \{1, ..., n\}$
- j The value of indicator  $j, j = \{1, ..., m\}$
- $x_{ij}$  The raw value of indicator j for return method i
- $I_{ij}$  The normalized value of  $x_{ij}$

#### 3. Weighting

The variables were relatively weighted with the AHP technique, which allows decision-makers to make tradeoffs between pairs of variables[56]. First, the decision maker decides which variable in the considered pair is more important. Second, the decision maker expresses how much more preferred this variable is by assigning a score on a semantic scale that ranges from 1 to 9 [55, 58, 64, 65]. A value of 1 (weak preference) is given when both indicators are equally important, a five corresponds to a strong preference, and a 9 expresses an extreme preference for one variable over another [57].

These preferences are combined in a reciprocal pairwise comparison matrix A  $(n \ge n)$ , where the  $a_{ii} = 1$ and  $a_{ji} = 1/a_{ij}$  should be met [74]. Each element  $a_{ij}$ expresses the relative preference assigned by the decision maker for alternative i to j, where the alternatives represent the variables [57, 64]. Next, the relative weights for each variable will be calculated by the principal right eigenvector approach of matrix A [57, 64]. Then, the consistency in the judgments of the weights needs to be ensured by reaching an acceptable level of consistency, which is determined by Equation 3.15 [43, 66]. Here,  $\lambda_{max}$  is the maximum eigenvalue, *n* is the number of variables, and RI can be derived from the random consistency index of [57, 81]. The Consistency Ratio (CR) should be below a threshold value, which is 0.10 with  $n \ge 5 [57, 66].$ 

$$A = [a_{ij}] = \begin{bmatrix} 1 & a_{ij} & \dots & a_{1n} \\ 1/a_{ij} & 1 & \dots & a_{2n} \\ \vdots & \vdots & \vdots & \vdots \\ 1/a_{1n} & 1/a_{2n} & \dots & 1 \end{bmatrix}$$
(3.14)

$$CR = \frac{\lambda_{max} - n}{(n-1) * RI} \tag{3.15}$$

### TABLE II: RI values corresponding to the number of variables n, from [57]

#### 4. Aggregation

The indicators are aggregated with multiplication aggregation (also Geometric mean (GME)), which provides an in-between solution between the full and noncompensability of the different variables. By multiplying the weights and normalized indicators, a value of the CI is obtained that reflects the relative importance of each indicator and how much an individual indicator contributes to the total score of the CI [81, 83]. The previously calculated functions are used as input for the formula for aggregation (Equation 3.16). In this equation,  $w_j$  expresses the weight of indicator j for product return method i.  $I_{ij}$  is the normalized value of indicator j for return method i.  $CI_i$  expresses the value of the CI for return method i [81].

$$CI_i = \prod_{j=1}^{10} I_{ij}^{w_j} \tag{3.16}$$

#### C. Phase 3. Construction of the benchmark

A new benchmark is designed, constructed, and integrated into this CI, expressing target performance. Benchmarking can identify performance gaps and is, therefore, a commonly used method in addition to performance measurement [81, 82]. The benchmark is designed using the Political and Legal criteria from the PESTLE analysis (Political (P), Economic (E), Sociological (S), Technological (T), Legal (L), and Environmental (E)) [61, 78]. European companies should comply with their performances with legislation and policy-making decisions, which therefore drive company improvements.

The benchmark is set for the year 2030. This will represent the shorter-term goals of a company in its net-zero journey for 2050, which are more tangible and more accessible for companies to achieve as it is less than a decade away. Also, shorter-term goals allow companies more flexibility in their strategies for achieving them. The reduction targets are obtained through reports and official publications of the EC. Companies need to reduce their emissions according to reduction targets set by the EU. These targets are translated into company-specific values for the benchmark with data from that company. There are no targets set for the economic indicators, as making any accurate assumptions on the benchmark costs for 2030 is impossible. These costs are embedded with uncertainty, and making any assumptions to this end would negatively affect the reliability of the benchmark. For this reason, the base case costs for the economic benchmarks are applied in this research.

- Packaging should be made of recycled materials by 2030. The packaging recycling rate for cardboard should be 85% and for plastic 55% in 2030. Packaging materials sent to landfills should decrease to 9.6% by 2030 [25, 28]. It is assumed that the remaining packaging material is combusted.
- The target value for transport emission reduction by the EC is set at 55% by 2030 compared to the base year 1990, which was accepted in 2020 [30]. Assuming a linear reduction rate, this would mean a yearly GHG-emission reduction of 1.57% from 1995-2030, which is applied.
- The EC has put a benchmark of 32% of renewable energy sources in buildings by 2030, which

will be the target value for warehouse- and brickand-mortar energy usage in the benchmark in this research [22]. Electricity produced by renewables should power warehousing equipment as well in 2030.

• Incineration and landfill should be limited to 1% of the total waste treatments, and a reduction of overall disposal emissions of 30% by 2030 should be met [24, 26, 28, 45].

The benchmark is constructed by rescaling, Equal Weights (EW), and aggregation by multiplication. Instead of AHP, EW is applied to improve the generalizability and transparency of the benchmark. The aggregated results of the CI, representing the company's performance, were compared with the target performance by analyzing the performance gap M. The performance gap is determined with Equation 3.17, where a lower value of M represents better performance, where  $CI_b$  is the benchmark performance and  $CI_a$  is the actual performance of a company [17].

$$M = CI_b - CI_a \tag{3.17}$$

#### D. Phase 4. Robustness evaluation of the CI

Insight into how- and to what extent the final value of the CI was affected by a source of uncertainty for building the CI [33, 70] is obtained by inspecting uncertainty triggers by uncertainty- and sensitivity analysis [56]. The triggers analyzed were  $X_1$  rescaling weights,  $X_2$  use of a different aggregation method, and  $X_3$  varying the input data. The normalization method was not analyzed as the rescaling method is the only appropriate normalization method for this CI. The weights were rescaled, Linear aggregation (LIN) was adopted for aggregation, and uncertainty sources in the input data were increased- and decreased by 25%, while the rest remained unchanged.

The model was tested with a case study and a scenario analysis based on the PESTLE framework [61, 78]. This scenario analysis served as a tool to test the models' performance.

#### IV. RESULTS

First, an introduction was given to the experiment settings. Next, the results were presented of the performance of the different return methods.

#### A. Experiment settings

#### 1. Case study

Data was retrieved from an international fashion retailer, mainly active in the downstream supply chain. The data consisted of 6388 returned products from customers in the Netherlands in the base year. The retailer offers in-store returns, home pick-ups, and collection point return services. The retailer's brick-andmortar store is located in the Netherlands, and the warehouse for returns is in Spain.

#### 2. Scenario set-up

Scenarios were created with PESTLE. A moderate and extreme variant was developed for each scenario based on literature studies, reports, and expert insights. The scenarios are implemented in the model to test its performance.

- Scenario 1. Shifting customer behavior (PESTLE) Drive by more conscious customer behavior, less opportunistic returns, choice of return method based on environmental impact, reuse packaging box, recycle packaging, financially discouraging customers by a return fee.
- Scenario 2. Green supply chain management (PESTLE)

Driven by stakeholder engagement, adoption of electric vehicles, short-haul transportation, LNGfueled long-haul transportation, renewable energy for buildings and warehouses, financial compensation costs in a return fee.

• Scenario 3. Economic attractiveness of renewables (PESTLE)

The rise in fossil fuel prices, increased use of renewables and green transportation, and financial government support.

• Scenario 4. Governmental support for circularity (**PESTLE**)

Extended Producer Responsibility, higher recycling rates for packaging material, lower land-fill/incineration rates, and financial support from the government.

• Scenario 5. Technological innovations (PESTLE) More efficient transport, better online shopping experiences, fewer in-store operations, and more e-commerce increased the recycling potential of clothing.

#### B. Performance measurement with the CI

The raw values for each indicator for the return methods and the benchmark are shown in Table III, normalized in Table IV. The weights were derived from the reciprocal matrix in Table V, which expressed the relative preferences of the decision makers between each pair of variables. This resulted in the weights presented in Table VI, satisfying the reciprocal constraints with CR below 0.1.

Table VIII showed the aggregated value of the CI scores of each return method. A higher value of the CI

represented better performance, and a lower value indicated worse performance. Based on the data of this retailer, the in-store return method scored lowest among the return alternatives, and the home pick-up scored best in terms of environmental and economic performance. The largest differences between the return methods are within the environmental performance, as seen in Table VII. The base case overall performance is the average of the return methods, which comes close to the CI score of the collection point returns. This is because almost all returned items are returned through collection point returns for this retailer (92.45%).

Next, the performance gaps with the benchmark are calculated, where a lower value represents better performance [17]. As seen in Table IX, if home pick-up returns a product, the company's performance comes closer to the target performance than when a customer uses other return methods. The differences between the performance of the return methods is visualized in Figure 3. Higher values result in better performance, corresponding to values more on the outside of the axis in the radar. Values closer to the center are less preferred in terms of performance. This shows that the benchmark outperforms almost all return methods on the environmental indicators, identifying areas for improvement for the company to comply with EC targets. Companies use these insights to define improvement strategies that target specific measures. Although in-store returns score best on packaging emissions, transport costs, and brickand-mortar costs, the aggregated results show that this return method is not advantageous to improve the overall performance of a company. When a company focuses on improving transport emissions, it is interesting to use home pick-ups more often for returns, as this method lowers the performance gap with the benchmark.



FIG. 3: Radar chart base case with the benchmark, source: (Author)

TABLE III: Raw indicator values base case with the benchmark, source: (Author)

	$V_1$	$V_2$	$V_3$	$V_4$	$V_5$	$V_6$	V7	$V_8$	$V_9$	$V_{10}$
	[kgCO2e/	[kgCO2e/	[kgCO2e/	[kgCO2e/	[kgCO2e/	[euro/	[euro/	[euro/	[euro/	[euro/
	returned product]									
Benchmark	0.016	14.498	0.743	0.007	0.484	0.230	12.379	5.096	0.523	1.903
Base case overall performance	0.075	32.217	1.092	0.012	0.712	0.230	12.379	5.096	0.523	1.903
Return method 1. Home pick-up	0.077	23.721	0.624	0.005	0.784	0.432	7.076	4.721	0.280	2.093
Return method 2. In-store	0.013	42.567	1.559	0.016	0.642	0.073	17.666	5.470	0.755	1.715
Return method 3. Collection point	0.078	32.592	1.118	0.010	0.710	0.217	12.673	5.117	0.539	1.896

### TABLE IV: Normalized values base case with the benchmark, source: (Author)

	$V_1$					$V_6$				
Benchmark	1.950	2.000	1.873	1.860	2.000	1.563	1.499	1.499	1.487	1.504
						1.563				
						1.000				
Return method 2. In-store	2.000	1.000	1.000	1.000	1.473	2.000	1.000	1.000	1.000	2.000
Return method 3. Collection point	1.000	1.355	1.471	1.507	1.246	1.598	1.471	1.471	1.454	1.520

#### TABLE V: Reciprocal comparison matrix, source: (Author)

	$V_1$	$V_2$	$V_3$	$V_4$	$V_5$	$V_6$	$V_7$	$V_8$	$V_9$	$V_{10}$
$V_1$	1	$\frac{1}{4}$	$\frac{1}{2}$	$\frac{1}{2}$	1	$\frac{1}{4}$	$\frac{1}{3}$	$\frac{1}{2}$	$\frac{1}{2}$	1
$V_2$	4	1	$\overline{2}$		3	1	$\frac{1}{2}$			6
$V_3$	2	$\frac{1}{2}$	1	3	2	$\frac{1}{2}$	$\frac{1}{2}$	1	2	3
$V_4$	2	$\frac{\overline{1}}{2}$	$\frac{1}{3}$	1	$2 \\ \frac{1}{3} \\ 1$	$\frac{\overline{1}}{2}$	$\frac{\overline{1}}{3}$	$\frac{1}{2}$	$\frac{1}{2}$	3
$V_2$ $V_3$ $V_4$ $V_5$ $V_6$ $V_7$ $V_8$	1	$\frac{1}{3}$	$\frac{1}{2}$	$egin{array}{c} 1 \\ 3 \\ 2 \end{array}$	ĭ	$\frac{1}{2}$	$\frac{1}{4}$	$\begin{array}{c} 1 \\ rac{1}{2} \\ 3 \\ 2 \end{array}$	$\tilde{2}$	1
$V_6$	4	ĭ	$\overline{2}$	2	2	Ī	$\frac{1}{3}$	2	2	3
$V_7$	3	2	2	3	4	3	ľ	3	6	2
$V_8$	2	$\frac{1}{2}$	1	2	$\frac{1}{3}$	$\frac{1}{2}$	$\frac{1}{3}$	1	<b>2</b>	1
$V_9$	2	$\frac{1}{2}$	$\frac{1}{2}$	2	$\frac{\frac{1}{3}}{\frac{1}{2}}$	$\frac{1}{2}$	$\frac{1}{6}$	$\frac{1}{2}$	1	2
$V_{10}$	1	$\frac{1}{4}$ 1 $\frac{1}{2}$ $$	$\frac{1}{2} \frac{2}{2} \frac{1}{1} \frac{1}{3} \frac{1}{2} \frac{2}{2} \frac{2}{2} \frac{1}{1} \frac{1}{2} \frac{1}{3} \frac{1}$	$\frac{1}{3}$	$\tilde{1}$	$\frac{1}{4}$ 1 $\frac{1}{2}$ 1 $\frac{1}$	13121213141311131131612	$\tilde{1}$	$\frac{1}{2}$	1

#### C. Scenario testing

This section evaluates the performance of the different scenario designs (see section IVA2). The results of the performance gaps with the benchmark are visualized in Figure 4, where the height of the column represents the performance gap relative to the benchmark. The largest gap is found in the extreme variant of scenario 3; the economic attractiveness of renewables. The extreme variant of scenario 1 has the smallest performance gap with the benchmark relative to the other scenarios, indicating better performance.

The results show that the most considerable improvements in overall performance can be achieved when fewer products are returned due to more conscious behavior and, on the other hand, a higher recycling rate of packaging materials is achieved. Additionally, a return fee to discourage customers from returning items also leads to performance improvements relative to the base case, closing the gap with the benchmark. This is a crucial component to lower the costs of returns when no governmental support is given. The scenario analysis shows the importance of the financial support of governments for companies to ensure feasibility in company efforts to improve their environmental impact. Adopting less harmful waste

TABLE VI: Weights derived by AHP, source: (Author)

	Weight
$V_1$	0.044
$V_2$	0.158
$V_3$	0.104
$V_4$	0.060
$V_5$	0.085
$V_6$	0.134
$V_7$	0.228
$V_8$	0.075
$V_9$	0.063
$V_{10}$	0.048

 TABLE VII: Environmental performance score and
 economic performance score of the return methods,

 source:
 (Author)

	Environmental performance score	Economic performance score
Base case overall performance	1.334	1.513
Return method 1. Home pick-up	1.545	1.588
Return method 2. In-store	1.150	1.260
Return method 3. Collection point	1.339	1.504

treatments that enhance circularity shows improvements in environmental areas but are slightly higher in costs, affecting the overall performance. The gains achieved by technological innovations depend on the availability of technology and economies of scale that might help to lower costs, shrinking the performance gap with the benchmark.

#### V. DISCUSSION

#### A. Robustness evaluation of the CI

Each uncertainty trigger was implemented in the CI, and its effect on the final value of the CI using base case data were inspected and statistically tested using the Pearson correlation coefficient [60, 81]. The Uncertainty Analysis (UA) has concluded that a strong correlation ex-

TABLE VIII: Aggregated CI scores with benchmark, source: (Author)

	CI score
Benchmark	1.710
Base case overall performance	1.430
Return method 1. Home pick-up	1.568
Return method 2. In-store	1.209
Return method 3. Collection point	1.427



TABLE IX: Performance gap base case with the benchmark, source: (Author)

FIG. 4: Performance gaps scenarios relative to the benchmark, source: (Author)

ists between the constructed model and the alternative construction methods, indicating that using a different method leads to similar results. Also, uncertainty in the CI resulting from assumptions in the input data was analyzed using a Sensitivity Analysis (SA). This evaluation revealed that the model's output is robust to changes in data, reflecting the resilience of the CI.

#### B. Verification

The CI was successfully verified through a structured walkthrough, and the model's output was inspected by testing it with a base case. First, the calculation of the variables was inspected and checked on how they were computed, and a consistency check was done on their units of measurement. Second, the formulation and equations for rescaling, weighting, and aggregation were reviewed. Thirdly, the reliability of the model's output and set-up of the benchmark was inspected. The benchmark's set-up was verified using the verification framework PESTLE and through expert consultation. Next, it was checked whether the benchmark was implemented correctly to verify the reliability of the model's output. The benchmark was tested using performance analysis tools from the literature, which showed plausible results aligned with the expectations, and no surprises were found. The model's output showed that the model could find what needs to be understood, and the CI can be successfully verified.

#### C. Validation

Conceptual validation of the model took place by investigating the model's behavior and correctness through a scenario analysis. Five scenarios were developed with the verification tool PESTLE and expert insights, which improved the reliability of the scenarios with real-life developments. Implementing the scenarios in the model has shown that it can provide desired outputs, serve its purposes, and create valuable insights, validating the model and benchmark.

#### VI. CONCLUSION AND FUTURE RESEARCH

#### A. Conclusion

This research has created a new CI to measure the impact of product returns on the environmental and economic performance of fashion companies and created an understanding of this to improve their environmental footprint. A new benchmark was designed based on GHG reduction targets of the EC for 2030 which was developed and integrated into this CI to help companies improve sustainability.

The first scientific contribution of this research was that scientific research was added to the underexplored topic of product returns in the fashion industry, focusing on the effects and potential solutions of consumer product returns' environmental and economic impact. The second contribution was developing a measurement method for a complete assessment of the GHG emissions and costs of product returns, including all activities and return methods. Third, an overview was created to compare return methods based on their performance, which was lacking [7]. Fourth, a new CI was constructed for fashion companies to measure their performance, including designing and developing a new benchmark expressing environmental target performance. Last, both environmental and economic dimensions were included in this CI, thereby expanding research on this new type of company performance measurement by CIs [81].

Additionally, multiple practical contributions were made. Fashion companies can lower their environmental footprint- and costs by improving the returns process using the insights from the CI. Insight into the sources of GHG emissions and the costs of product returns were created, and how companies can measure them. Also, differences between the return methods and their performance were made explicit. Companies can use these combined insights to define, compare, and implement feasible strategies and well-informed decisions to improve sustainability, as both environmental improvements and economic risks were considered. Further, these insights created transparency on what- and how the environmental impact of returns was measured. This information can inform customers and create awareness of their return behavior. This leads to environmental and economic benefits for a company, as more conscious return choices are likely to be made, and opportunistic return behavior can be limited. Further, knowledge was created for companies to know how far they are in reaching the goals of the EC to combat climate change as they can measure and track their progress relative to these targets. With this, a transparent and sector-wide tool was developed that companies can use to see how closely their performances comply with EC targets, and governments are provided insight into how- and what environmental is measured. Governments can use this tool to compare companies' performances with each other and evaluate their progress as part of the prospected EU mandatory environmental reporting or to set more concrete targets to combat climate change sector-wide.

#### B. Future research

This study has created the information needed to provide transparency and inform customers, but it does not analyze how- and where companies in the purchasingand return process can use this information the best to encourage customers' conscious return decisions- and behavior most effectively. Therefore, it is recommended to develop a dashboard or digital twin where insights are obtained where- and how customers can best be encouraged using the insights obtained from this study. Further, it is recommended that companies mitigate the number of returns by adjusting the leniency of return policies. However, the potential effect on customer sales was left unconsidered in this study, forming the basis

- [1] J. Ader, J. Chai, M. Singer, P. Adhi, and H. Yankelevich. Returning to order: Improving returns management for apparel companies. *McKin*sey & Company, 4 2021. URL https://www. mckinsey.com/industries/retail/our-insights/ returning-to-order-improving-returns-management-
- S. Agrawal, R. K. Singh, and Q. Murtaza. Disposition decisions in reverse logistics: Graph theory and matrix approach. *Journal of Cleaner Production*, 137:93-104, 11 2016. ISSN 09596526. doi:10.1016/j.jclepro.2016.07.045. URL https://linkinghub.elsevier.com/retrieve/ pii/S0959652616309337.
- [3] T. A. Banihashemi, J. Fei, and P. S.-L. Chen. Exploring the relationship between reverse logistics and sustainability performance. *Modern Supply Chain Research and Applications*, 1(1):2–27, 2 2019. ISSN 2631-3871. doi:10.1108/MSCRA-03-2019-0009. URL https://www.emerald.com/insight/content/doi/10.1108/MSCRA-03-2019-0009/full/html.
- [4] A. Berg, A. Granskog, L. Lee, and K.-H. Magnus. Fashion on climate: How the fashion industry can urgently act to reduce its greenhouse-gas emissions. Technical report, McKinsey & Company and Global Fashion Agenda, 2020.

for the following possibilities for future research, which have both practical and scientific implications. First, the findings of this study should be incorporated to design new return policies, targeting lowering returns using financial discouragements based on the environmental impact created when returning a product. Research should be conducted into optimal financial discouragement while considering its effect on customer demand and a company's profitability. Also, exploring the efficiency of targeting specific product categories with their return policies is recommended, as some categories cause higher returns than others. The effect of the policies can be analyzed and tracked using the CI developed in this research. These insights may help companies adjust their return policies, contributing to lower returns and improved environmental impact. Following, investigating a company's decision-making process on the final destination of a returned product is recommended, as this affects performance. Future research should be done to increase the adoption potential of textile recycling and overcome its current limitations, where both the technical and practical feasibility should be considered for fashion companies. The final recommendation for future research is to investigate the most effective governmental support for companies in lowering their performance gap with the benchmark. Using the benchmark and CI to investigate the effectiveness of implementing these policies and monitor the effects on companies' performance is recommended.

#### BIBLIOGRAPHY

URL https://www.mckinsey.com/industries/retail/ our-insights/fashion-on-climate.

- [5] Bernon and J. Gorst. Online retail returns management: Integration within an omni-channel distribution context. *International Journal of Physical Distribution*
- returning-to-order-improving-returns-management-for-app&feLogisticsnMesnagement, 46(6/7), 2016. URL http: S. Agrawal, R. K. Singh, and Q. Murtaza. Disposition //shura.shu.ac.uk/11656/.
  - [6] M. Bernon, S. Rossi, and J. Cullen. Retail reverse logistics: A call and grounding framework for research. *International Journal of Physical Distribution and Logistics Management*, 41(5):484-510, 6 2011. ISSN 09600035. doi:10.1108/09600031111138835. URL https://www.emerald.com/insight/content/doi/10. 1108/09600031111138835/full/html.
  - [7] C. Bozzi, M. Neves, and C. Mont'alvão. Fashion E-Tail and the Impact of Returns: Mapping Processes and the Consumer Journey towards More Sustainable Practices. *Sustainability (Switzerland)*, 14(9), 2022. ISSN 20711050. doi:10.3390/su14095328. URL https://doi. org/10.3390/su14095328.
  - [8] H. Buldeo Rai. The net environmental impact of online shopping, beyond the substitution bias. *Journal of Transport Geography*, 93(October 2020):103058, 5 2021.

ISSN 09666923. doi:10.1016/j.jtrangeo.2021.103058. URL https://linkinghub.elsevier.com/retrieve/ pii/S0966692321001113.

- [9] I. D. Cardenas, W. Dewulf, T. Vanelslander, C. Smet, and J. Beckers. The e-commerce parcel delivery market and the implications of home B2C deliveries vs pick-up points. *International Journal of Transport Economics*, 44(2):235–256, 6 2017. ISSN 03035247. doi: 10.19272/201706702004.
- [10] J.-F. Cordeau, G. Laporte, J.-Y. Potvin, and M. W. Savelsbergh. Transportation on Demand. In C. Barnhart and G. Laporte, editors, *Handbooks in Operations Research and Management Science*, number C, chapter 7. Elsevier, 2007. doi:10.1016/S0927-0507(06)14007-4. URL https://linkinghub.elsevier.com/retrieve/ pii/S0927050706140074.
- [11] S. Cullinane and K. Cullinane. The Logistics of Online Clothing Returns in Sweden and How to Reduce its Environmental Impact. Journal of Service Science and Management, 14(01):72-95, 2021. ISSN 1940-9893. doi:10.4236/jssm.2021.141006. URL https://www.scirp.org/journal/doi.aspx?doi= 10.4236/jssm.2021.141006.
- [12] N. F. da Cruz, S. Ferreira, M. Cabral, P. Simões, and R. C. Marques. Packaging waste recycling in Europe: Is the industry paying for it? *Waste Management*, 34(2):298-308, 2 2014. ISSN 0956053X. doi:10.1016/j.wasman.2013.10.035. URL https://linkinghub.elsevier.com/retrieve/pii/ S0956053X13005217.
- [13] Davydenko I. and Hopman W. Effect of pick-up points and returns on CO2 emissions in last mile parcel delivery networks. Technical report, TNO, The Hague, 6 2020. URL www.tno.nl.
- [14] R. B. de Koster Marisa P. de Brito and M. A. van de Vendel. Return handling: an exploratory study with nine retailer warehouses. *International Journal of Retail & Distribution Management*, 30(8):407-421, 8 2002. ISSN 0959-0552. doi:10.1108/09590550210435291. URL https://www.emerald.com/insight/content/doi/10. 1108/09590550210435291/full/html.
- [15] R. Dekker, M. Fleischmann, K. Inderfurth, and L. Van Wassenhove. *Reverse Logistics*. Springer Berlin Heidelberg, Berlin, Heidelberg, 10 2004. ISBN 978-3-642-07380-9. doi:10.1007/978-3-540-24803-3. URL http: //link.springer.com/10.1007/978-3-540-24803-3.
- [16] J. Desrosiers, Y. Dumas, M. M. Solomon, and F. Soumis. Time Constrained Routing and Scheduling. In *Handbooks* in OR & MS, volume 8, chapter 2, pages 35–139. Elsevier Science B.V., 1995.
- M. P. Dočekalová and A. Kocmanová. Composite indicator for measuring corporate sustainability. *Ecological Indicators*, 61:612-623, 2 2016. ISSN 1470160X. doi:10.1016/j.ecolind.2015.10.012. URL https://linkinghub.elsevier.com/retrieve/pii/ S1470160X15005476.
- [18] P. Dutta, A. Mishra, S. Khandelwal, and I. Katthawala. A multiobjective optimization model for sustainable reverse logistics in Indian E-commerce market. *Journal of Cleaner Production*, 249, 3 2020. ISSN 09596526. doi: 10.1016/j.jclepro.2019.119348.
- [19] J. B. Edwards, A. C. McKinnon, and S. L. Cullinane. Comparative analysis of the carbon footprints of conventional and online retailing. *Inter-*

national Journal of Physical Distribution & Logistics Management, 40(1/2):103-123, 2 2010. ISSN 0960-0035. doi:10.1108/09600031011018055. URL https://www.emerald.com/insight/content/doi/10. 1108/09600031011018055/full/html.

- [20] D. Elrabaya, V. Marchenko, and D. Elrabay'a. Identifying the full cost to landfill municipal solid waste by incorporating emissions impact and land development lost opportunity: Case study, Sharjah-UAE. International Journal of Engineering Sciences, 10(6), 6 2021. doi:10.35629/6734-1006023341. URL https:// www.researchgate.net/publication/352464905.
- [21] European Commission. The European Green Deal, 12 2019. URL https://eur-lex.europa.eu/ legal-content/EN/ALL/?uri=CELEX:52019DC0640.
- [22] European Commission. European Green Deal Delivering on our Targets. Technical report, European Commission, 2021. URL https://www.google.com/ url?sa=t&rct=j&q=&esrc=s&source=web&cd=&ved= 2ahUKEwjZtsi1idH9AhUUgf0HHa92C5kQFnoECAwQAQ& url=https%3A%2F%2Fec.europa.eu%2Fcommission% 2Fpresscorner%2Fapi%2Ffiles%2Fattachment% 2F869807%2FEGD\_brochure\_EN.pdf.pdf&usg= A0vVaw2liwvguvSd64yb0n3MQjq9.
- [23] European Commission. Proposal for a Directive of The European Parliament and of The Council. Technical report, European Commission, Brussels, 4 2021. URL https://eur-lex.europa.eu/legal-content/EN/ TXT/?uri=CELEX:52021PC0189.
- [24] European Commission. Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions: EU Strategy for Sustainable and Circular Textiles, 3 2022. URL https://eur-lex.europa.eu/legal-content/EN/ TXT/?uri=CELEX%3A52022DC0141.
- [25] European Commission. European Green Deal: Putting an end to wasteful packaging, boosting reuse and recycling, 11 2022. URL https://ec.europa.eu/ commission/presscorner/detail/en/ip\_22\_7155.
- [26] European Commission. Landfill waste, 2022. URL https://environment.ec.europa.eu/topics/ waste-and-recycling/landfill-waste\_en.
- [27] European Commission. Proposal for a Regulation on Ecodesign for Sustainable Products, 3 2022.
- [28] European Commission. Waste, 2022. URL https://ec. europa.eu/eurostat/web/waste/targets.
- [29] European Environment Agency. Typical charge (gate fee and landfill tax) for legal landfilling of non-hazardous municipal waste in EU Member States and regions, 3 2013. URL https://www.eea.europa.eu/data-and-maps/ figures/typical-charge-gate-fee-and.
- [30] European Environment Agency. Greenhouse gas emissions from transport in Europe, 10 2022. URL https://www.eea.europa.eu/ims/ greenhouse-gas-emissions-from-transport.
- [31] European Union. The impact of textile production and waste on the environment (infographic), 4 2022. URL https://www.europarl.europa.eu/ news/en/headlines/society/20201208ST093327/ the-impact-of-textile-production-and-waste-on-the-environment textile-production-and-waste-on-the-environment textile-production-and-waste-on-the-environment textile-production-and-waste-on-the-environment textile-production-and-waste-on-the-environment textile-production-and-waste-on-the-environment textile-production-and-waste-on-the-environment
- [32] M. Fleischmann, H. R. Krikke, R. Dekker, and S. D. P. Flapper. A characterisation of logistics networks for product recovery. *Omega*, 28(6):653–666, 12 2000. ISSN

0305-0483. doi:10.1016/S0305-0483(00)00022-0.

- [33] M. Freudenberg. Composite Indicators of Country Performance: A Critical Assessment. 2003. URL https: //www.researchgate.net/publication/273697960.
- [34] P. Gazzola, E. Pavione, R. Pezzetti, and D. Grechi. Trends in the Fashion Industry. The Perception of Sustainability and Circular Economy: A Gender/Generation Quantitative Approach. Sustainability, 12(7):2809, 4 2020. ISSN 2071-1050. doi:10.3390/su12072809. URL https://www.mdpi.com/2071-1050/12/7/2809.
- [35] GHG Protocol. A Corporate Accounting and Reporting Standard. Technical report, World Resources Institute and World Business Council for Sustainable Development, 2004. URL https://ghgprotocol.org/ corporate-standard.
- [36] GHG Protocol. Technical Guidance for Calculating Scope 3 Emissions, 2013. URL https: //www.google.com/url?sa=t&rct=j&q=&esrc=s& source=web&cd=&cad=rja&uact=8&ved=2ahUKEwidj\_ rA67z9AhVAgf0HHcvcCNsQFnoECA0QAQ&url=https%3A% 2F%2Fghgprotocol.org%2Fsites%2Fdefault%2Ffiles% 2Fstandards%2FScope3\_Calculation\_Guidance\_0.pdf& usg=A0vVaw33yIdeHazIP4adh4NRd-Ar.
- [37] A. Gialos, V. Zeimpekis, M. Madas, and K. Papageorgiou. Calculation and Assessment of CO2e Emissions in Road Freight Transportation: A Greek Case Study. *Sustainability*, 14(17):10724, 8 2022. ISSN 2071-1050. doi:10.3390/su141710724. URL https://www.mdpi.com/ 2071-1050/14/17/10724.
- [38] J. Gómez-Limón, M. Arriaza, and M. Guerrero-Baena. Building a Composite Indicator to Measure Environmental Sustainability Using Alternative Weighting Methods. *Sustainability*, 12(11):4398, 5 2020. ISSN 2071-1050. doi:10.3390/su12114398. URL https://www.mdpi.com/ 2071-1050/12/11/4398.
- [39] E. Gustafsson, P. Jonsson, and J. Holmström. Reducing retail supply chain costs of product returns using digital product fitting. International Journal of Physical Distribution & Logistics Management, 51(8):877-896, 8 2021. ISSN 0960-0035. doi:10.1108/IJPDLM-10-2020-0334. URL https://www.emerald.com/insight/content/ doi/10.1108/IJPDLM-10-2020-0334/full/html.
- [40] Z. He, P. Chen, H. Liu, and Z. Guo. Performance measurement system and strategies for developing low-carbon logistics: A case study in China. *Journal* of Cleaner Production, 156:395-405, 7 2017. ISSN 09596526. doi:10.1016/j.jclepro.2017.04.071. URL https://linkinghub.elsevier.com/retrieve/pii/ S0959652617307898.
- [41] A. Holding and A. Gendell. Polybags in the fashion industry: evaluating the options. Technical report, Fashion For Good, 12 2019.
- [42] M. Ikram. Transition toward green economy: Technological Innovation's role in the fashion industry. *Current Opinion in Green and Sustainable Chemistry*, 37:100657, 10 2022. ISSN 2452-2236. doi: 10.1016/J.COGSC.2022.100657.
- [43] A. Ishizaka and A. Labib. Review of the main developments in the analytic hierarchy process. *Expert Systems with Applications*, 38(11):14336-14345, 5 2011. ISSN 09574174. doi:10.1016/j.eswa.2011.04.143. URL https://linkinghub.elsevier.com/retrieve/pii/S0957417411006701.

- [44] K. Khor and Z. Udin. Impact of Reverse Logistics Product Disposition towards Business Performance in Malaysian E&E Companies. Journal of Supply Chain and Customer Relationship Management, pages 1-19, 2 2012. ISSN 23267046. doi:10.5171/2012.699469. URL http://www.ibimapublishing.com/journals/ JSCCRM/2012/699469/699469.html.
- [45] M. Koszewska. Circular Economy Challenges for the Textile and Clothing Industry. Autex Research Journal, 18(4):337-347, 12 2018. ISSN 2300-0929. doi: 10.1515/aut-2018-0023. URL https://www.sciendo. com/article/10.1515/aut-2018-0023.
- [46] D. M. Lambert and T. L. Pohlen. Supply Chain Metrics. The International Journal of Logistics Management, 12(1):1-19, 1 2001. ISSN 0957-4093. doi:10.1108/09574090110806190. URL https://www.emerald.com/insight/content/doi/ 10.1108/09574090110806190/full/html.
- [47] C. Lohman, L. Fortuin, and M. Wouters. Designing a performance measurement system: A case study. *European Journal of Operational Research*, 156(2):267–286, 7 2004. ISSN 0377-2217. doi:10.1016/S0377-2217(02)00918-9.
- [48] S. Mahar and P. D. Wright. In-Store Pickup and Returns for a Dual Channel Retailer. *IEEE Transactions* on Engineering Management, 64(4):491-504, 11 2017. ISSN 0018-9391. doi:10.1109/TEM.2017.2691466. URL http://ieeexplore.ieee.org/document/7914623/.
- [49] R. Mangiaracina, A. Perego, S. Perotti, and A. Tumino. Assessing the environmental impact of logistics in online and offline B2C purchasing processes in the apparel industry. *International Journal of Logistics Systems and Management*, 23(1):98, 2016. ISSN 1742-7967. doi:10.1504/IJLSM.2016.073300. URL http: //www.inderscience.com/link.php?id=73300.
- [50] M. Mazziotta and A. Pareto. METHODS FOR CON-STRUCTING COMPOSITE INDICES: ONE FOR ALL OR ALL FOR ONE? 1. Rivista Italiana di Economia Demografia e Statistica, 2013.
- [51] P. Micheli and L. Mari. The theory and practice of performance measurement. *Management Accounting Re*search, 25(2):147-156, 6 2014. ISSN 10445005. doi: 10.1016/j.mar.2013.07.005. URL https://linkinghub. elsevier.com/retrieve/pii/S104450051300070X.
- [52] G. Munda. Choosing Aggregation Rules for Composite Indicators. Social Indicators Research, 109:337–354, 9 2012. doi:https://doi.org/10.1007/s11205-011-9911-9.
- [53] G. Munda and M. Nardo. Noncompensatory/nonlinear composite indicators for ranking countries: a defensible setting. *Applied Economics*, 41(12):1513-1523, 5 2009. ISSN 0003-6846. doi:10.1080/00036840601019364. URL http://www.tandfonline.com/doi/abs/10.1080/ 00036840601019364.
- [54] I. V. Muralikrishna and V. Manickam. Life Cycle Assessment. In *Environmental Management*, pages 57–75. Elsevier, 2017. doi:10.1016/B978-0-12-811989-1.00005-1. URL https://linkinghub.elsevier.com/retrieve/pii/B9780128119891000051.
- [55] M. Nardo, M. Saisana, A. Saltelli, and S. Tarantola. Tools for Composite Indicators Building. Technical report, European Communities, 2005. URL https://www. researchgate.net/publication/277294848.
- [56] M. Nardo, M. Saisana, A. Saltelli, S. Tarantola, A. Hoffmann, E. Giovannini, and Organisation for Economic Co-operation and Development. *Handbook on construct*-

- [57] P. M. Orencio and M. Fujii. A localized disasterresilience index to assess coastal communities based on an analytic hierarchy process (AHP). International Journal of Disaster Risk Reduction, 3(1):62-75, 3 2013. ISSN 22124209. doi:10.1016/j.ijdrr.2012.11.006. URL https://linkinghub.elsevier.com/retrieve/pii/ S2212420912000428.
- [58] M. Pakkar. A Hierarchical Aggregation Approach for Indicators Based on Data Envelopment Analysis and Analytic Hierarchy Process. Systems, 4(1):6, 1 2016. ISSN 2079-8954. doi:10.3390/systems4010006. URL http://www.mdpi.com/2079-8954/4/1/6.
- [59] H. Palsson, F. Pettersson, and L. Winslott Hiselius. Energy consumption in e-commerce versus conventional trade channels - Insights into packaging, the last mile, unsold products and product returns. *Journal* of Cleaner Production, 164:765–778, 10 2017. ISSN 09596526. doi:10.1016/j.jclepro.2017.06.242. URL https://linkinghub.elsevier.com/retrieve/pii/ S0959652617314117.
- [60] A. Pereira and R. Broed. Methods for Uncertainty and Sensitivity Analysis. Review and recommendations for implementation in Ecolego. Technical report, AlbaNova University Center, 1 2006.
- [61] R. Perera. *The PESTLE Analysis*. Nerdynaut, 9 2017. ISBN 9781790845323.
- [62] N. Pollesch and V. Dale. Normalization in sustainability assessment: Methods and implications. *Ecological Economics*, 130:195–208, 10 2016. ISSN 09218009. doi:10.1016/j.ecolecon.2016.06.018. URL https://linkinghub.elsevier.com/retrieve/pii/ S0921800915305899.
- [63] J. M. Ries, E. H. Grosse, and J. Fichtinger. Environmental impact of warehousing: a scenario analysis for the United States. *International Journal of Production Research*, 55(21):6485–6499, 11 2017. ISSN 0020-7543. doi:10.1080/00207543.2016.1211342. URL https://www.tandfonline.com/doi/full/10.1080/00207543.2016.1211342.
- [64] T. L. Saaty. How to make a decision: The analytic hierarchy process. European Journal of Operational Research, 48(1):9-26, 9 1990. ISSN 03772217. doi:10.1016/0377-2217(90)90057-I. URL https://linkinghub.elsevier. com/retrieve/pii/037722179090057I.
- [65] M. Sadegh Pakkar. Using data envelopment analysis and analytic hierarchy process to construct composite indicators. Journal of Applied Operational Research, 6(3): 174-187, 2014. ISSN 1927-0089. URL www.tadbir.ca.
- [66] R. K. Singh, H. Murty, S. Gupta, and A. Dikshit. Development of composite sustainability performance index for steel industry. *Ecological Indicators*, 7(3):565–588, 7 2007. ISSN 1470160X. doi:10.1016/j.ecolind.2006.06.004. URL https://linkinghub.elsevier.com/retrieve/pii/S1470160X06000550.
- [67] Statista. Fashion worldwide, 2 2023. URL https://www.statista.com/outlook/dmo/ecommerce/ fashion/worldwide?currency=EUR.
- [68] B. Stöcker, D. Baier, and B. M. Brand. New insights in online fashion retail returns from a customers' perspective and their dynamics. *Journal of Business Economics*, 91(8):1149–1187, 10 2021. ISSN 0044-2372. doi:10.1007/s11573-021-01032-1. URL https://link.

springer.com/10.1007/s11573-021-01032-1.

- [69] Y. Su, H. Duan, Z. Wang, G. Song, P. Kang, and D. Chen. Characterizing the environmental impact of packaging materials for express delivery via life cycle assessment. *Journal of Cleaner Production*, 274:122961, 11 2020. ISSN 09596526. doi:10.1016/j.jclepro.2020.122961. URL https://linkinghub.elsevier.com/retrieve/ pii/S0959652620330067.
- [70] B. Talukder, K. W. Hipel, and G. W. vanLoon. Developing Composite Indicators for Agricultural Sustainability Assessment: Effect of Normalization and Aggregation Techniques. *Resources*, 6(4):66, 11 2017. ISSN 2079-9276. doi:10.3390/resources6040066. URL http: //www.mdpi.com/2079-9276/6/4/66.
- T. S. Thorisdottir and L. Johannsdottir. Sustainability within Fashion Business Models: A Systematic Literature Review. Sustainability, 11(8):2233, 4 2019. ISSN 2071-1050. doi:10.3390/su11082233. URL https://www. mdpi.com/2071-1050/11/8/2233.
- [72] X. Tian and J. Sarkis. Emission burden concerns for online shopping returns. Nature Climate Change, 12 (1):2-3, 1 2022. ISSN 1758-678X. doi:10.1038/s41558-021-01246-9. URL https://www.nature.com/articles/ s41558-021-01246-9.
- [73] S. van der Meulen, T. Grijspaardt, W. Mars, W. van der Geest, A. Roest-Crollius, and J. Kiel. Cost Figures for Freight Transport-final report. Technical report, Panteia, 2023.
- [74] L. G. Vargas. An overview of the analytic hierarchy process and its applications. *European Journal of Operational Research*, 48(1):2-8, 9 1990. ISSN 03772217. doi:10.1016/0377-2217(90)90056-H. URL https://linkinghub.elsevier.com/retrieve/ pii/037722179090056H.
- [75] R. Velazquez and S. M. Chankov. Environmental Impact of Last Mile Deliveries and Returns in Fashion E-Commerce: A Cross-Case Analysis of Six Retailers. In 2019 IEEE International Conference on Industrial Engineering and Engineering Management (IEEM), pages 1099-1103. IEEE, 12 2019. ISBN 978-1-7281-3804-6. doi:10.1109/IEEM44572.2019.8978705. URL https:// ieeexplore.ieee.org/document/8978705/.
- [76] D. Vigo, J.-F. Cordeau, G. Laporte, and M. W. P. Savelsbergh. Vehicle Routing. volume 14. Transportation, handbooks in operations and management science, 1 2007. URL https://www.researchgate.net/ publication/233843551.
- [77] M. Von Zahn, K. Bauer, C. Mihale-Wilson, J. Jagow, M. Speicher, and O. Hinz. The Smart Green Nudge: Reducing Product Returns through Enriched Digital Footprints & Causal Machine Learning. 11 2022. URL https://ssrn.com/abstract=4262656.
- S. Widya Yudha, B. Tjahjono, and A. Kolios. A PESTLE Policy Mapping and Stakeholder Analysis of Indonesia's Fossil Fuel Energy Industry. *Energies*, 11(5):1272, 5 2018. ISSN 1996-1073. doi:10.3390/en11051272. URL http: //www.mdpi.com/1996-1073/11/5/1272.
- [79] A. Wiese, W. Toporowski, and S. Zielke. Transportrelated CO2 effects of online and brick-and-mortar shopping: A comparison and sensitivity analysis of clothing retailing. *Transportation Research Part D: Transport and Environment*, 17(6):473–477, 8 2012. ISSN 13619209. doi: 10.1016/j.trd.2012.05.007. URL https://linkinghub. elsevier.com/retrieve/pii/S1361920912000521.

- [80] R.-N. Yan, S. Diddi, and B. Bloodhart. Predicting clothing disposal: The moderating roles of clothing sustainability knowledge and self-enhancement values. *Cleaner and Responsible Consumption*, 3:100029, 12 2021. ISSN 26667843. doi:10.1016/j.clrc.2021.100029. URL https://linkinghub.elsevier.com/retrieve/ pii/S2666784321000231.
- [81] Q. Zeng. Development of A Composite Indicator for Measuring Company Performance from Economic and Environmental Perspectives: A Study on Motor Vehicle Manufacturers, 2020. URL https://repository.tudelft.nl/islandora/object/ uuid:75d0ae4c-afbc-4b7a-814d-8bbb3642b95f?

collection=research.

- [82] Q. Zeng, W. Beelaerts van Blokland, S. Santema, and G. Lodewijks. Benchmarking company performance from economic and environmental perspectives. *Benchmarking: An International Journal*, 27(3):1127– 1158, 12 2019. ISSN 1463-5771. doi:10.1108/BIJ-05-2019-0223. URL https://www.emerald.com/insight/ content/doi/10.1108/BIJ-05-2019-0223/full/html.
- [83] P. Zhou and L. P. Zhang. Composite Indicators for Sustainability Assessment: Methodological Developments. In Energy, Environment and Transitional Green Growth in China, pages 15–36. Springer Singapore, Singapore, 2018. doi:10.1007/978-981-10-7919-1"2. URL http: //link.springer.com/10.1007/978-981-10-7919-1\_2.

## B

## Preference scale of AHP

The AHP uses a preference scale to assign importance to a variable in each pair considered. The scale is expressed from 1 to 9, where each factor corresponds to a degree of preference [168, 160].

Table B.1: Preference/importance scale AHP, from [167]
--

Factor of preference	Importance
1	Equally preferred
2	Equally to moderately preferred
3	Moderately preferred
4	Moderately to strongly preferred
5	Strongly preferred
6	Strongly to very strongly preferred
7	Very strongly preferred
8	Very to extremely strongly preferred
9	Extremely preferred

## C

### Dimensions of a package

This Appendix gives insight into different packaging dimensions used for e-commerce in fashion. The measurement units of the variables are expressed per product, and the emissions generated in product returns must be allocated to one product return. Therefore it is necessary to specify the average dimensions used for a unit product return in this research. This will be done for an average fashion item. Small, medium, and large boxes are typically used for shipment in fashion e-commerce. For example, small boxes are mainly used for belts, jewelry, socks, and other small accessories. Medium-sized boxes are commonly used when transporting jeans, dresses, sweaters, or shoes. Larger parcels are suitable, for example, oversized winter coats, backpacks, or jackets [193]. Although jackets and coats can fall into larger-sized packages, most fashion items are relatively small and lightweight. For this reason, it is more likely that the average packaging size of a fashion product is a medium-sized package. This corresponds to a package mass of approximately 1 kilogram [193]. This research assumes the size of a cardboard box for product return of 22.9 x 22.9 x 22.9 cm (9 x 9 x 9 inches) with a mass of 0.279 kg [34]. For corrugated cardboard boxes, a mass of 0.227 kg is handled [185]. Although these box dimensions will be used as a standard for this research, it should be considered that the value for the product size can vary per company and product type.

These dimensions are also considered for plastic polybags with an average mass of 0.011 kg [188].

Packaging type	Box dimensions [cm]	Weight [kg]	Database/Source
Cardboard	15.24 x 15.24 x 15.24	0.127	[34]
Cardboard	22.86 x 22.86 x 22.86	0.291	[34]
Cardboard	30.86 x 30.86 x 30.86	0.49	[34]
Corrugated cardboard	22.86 x 22.86 x 22.86	0.279	[185]
LDPE polybag	22.86 x 22.86 x 22.86	0.011	[188]

Table C.1: Dimensions for the most common sizes of packaging, adapted from [34, 185, 188]

## D

## Context data

This Appendix provides an overview of the input data used to perform base case calculations and measurements.

#### Table D.1: Context data

Data description	Figures	Source	Assumptions/additional comments
Retailer features			
Warehouse size [m2]	6500	Client data	This warehouse is solely dedicated to handling product returns.
Store size [m2]	120	Client data	
Warehouse energy rating	n/a	-	
Store energy rating [A-G]	С	Client data	
Returns handled at the warehouse/year [#]	1.090.000	Client data	
Warehouse employees [#]	208	Client data	
Location store	NL	Client data	
Location warehouse	ES	Client data	
Products handled in-store/year [#]	-	Client data	
Logistics and transport features			
Returned products/year from the Netherlands [#]	6388	Client data	This number corresponds to the number of products delivered in the Dutch market of this retailer.
Transport modalities used	Road transportation	Client data	No further specifications given on the type of road vehicles.
Transport providers used	UPS, PostNL	Client data	
			The time to handle a return is between 5-15 minutes based
Average time for handling returns in store [min]	10	Expert consultation	on expert consultation, so the average of this is assumed as the input value.
			This number is also in line with estimations done by [118]
Average time product spends at storage in warehouse	3-5 days	Expert consultation	
Energy consumption			
Annual energy consumption [kWh/m2/year]	n/a	-	
Annual energy consumption store [kWh/m2/year]	225	RVO [155]	This number is assumed based on the energy label of the store.
Equipment			
Forklifts	3	Client data	
Hand pallet truck	5	Client data	
Conveyor belt	1	Client data	
Packaging			
Packaging type used by retailer	Cardboard and polybags	Client data	A product is packaged with a light polybag, next the product is packaged with either cardboard or a poly bag.

## Ε

## Environmental data

In this Appendix, the data is provided on the various sources of inputs for the environmental calculations. In the column named 'assumptions/additional comments', information is given on the scope of the EF.

Table E.1: Data for the calculation and measurement of emissions

Data description	Figures	Database/source	Assumptions/additional comments
Buildings			
Warehouse and storage facility leased (scope 3) [kgCO2e/m2]	86.344	GHG Protocol [88]	
Machinery and equipment [kgCO2e/EUR]	0.373	DEFRA [44]	
Electricity unknown [kgCO2e/kWh]	0.337	CO2emissiefactoren [31]	Well-to-wheel
Electricity renewables [kgCO2e/kWh]	0	CO2emissiefactoren [31]	Well-to-wheel
Clothing - waste treatements			
Reuse/restocking of clothing [kgCO2e/kg]	0.152	DEFRA [43]	
Landfill of clothing [kgCO2e/kg]	0.450	DEFRA [43]	Collection, transportation, and landfill emissions are included [43].
Incineration of clothing [kgCO2e/kg]	0.021	DEFRA [43]	Includes the process of incineration and generation of electricity [44]
Recycling of clothing closed loop [kgCO2e/kg]	0.021	DEFRA [43]	Including transport to a recovery facility [43].
Packaging material			
Cardboard [kgCO2e/kg]	1.53	[34]	Cradle to grave [34]
Plastic (LDPE) [kgCO2e/kg]	2.10	[139]	Cradle to raw material [139]
Plastic (HDPE) [kgCO2e/kg]	1.92	[139]	Cradle to raw material [139]
Corrugated cardboard [kgCO2e/kg]	0.94	[34]	Box with recycled material of 61.4%, cradle to grave [34].
Packaging - waste treatements			
Cardboard landfill [kgCO2e/kg]	1.042	DEFRA [43]	Including collection, transportation, and landfill emissions [43].
Cardboard incineration [kgCO2e/kg]	0.021	DEFRA [43]	Includes the process of incineration and generation of electricity [44]
Plastic average landfill [kgCO2e/kg]	0.089	DEFRA [43]	Including collection, transportation, and landfill emissions [43].
Plastic average incineration [kgCO2e/kg]	0.021	DEFRA [43]	Includes the process of incineration and generation of electricity [44]
Plastic HDPE landfill [kgCO2e/kg]	0.089	DEFRA [43]	Including collection, transportation, and landfill emissions [43].
Plastic HDPE incineration [kgCO2e/kg]	0.021	DEFRA [43]	Includes the process of incineration and generation of electricity [44]
Plastic LDPE and LLDPE landfill [kgCO2e/kg]	0.089	DEFRA [43]	Including collection, transportation, and landfill emissions [43].
Plastic LDPE and LLDPE incineration [kgCO2e/kg]	0.021	DEFRA [43]	Includes the process of incineration and generation of electricity [44]
Passenger travel			
Bike (non-electric) [kgCO2e/km]	0	CO2emissiefactoren [31]	Well-to-wheel
Walk [kgCO2e/km]	0	CO2emissiefactoren [31]	Well-to-wheel
PT (metro, bus, tram) [kgCO2e/km]	0.075	CO2emissiefactoren [31]	Well-to-wheel
Car (fuel type unknown) [kgCO2e/km]	0.193	CO2emissiefactoren [31]	Well-to-wheel
Car (petrol, medium size) [kgCO2e/km]	0.204	CO2emissiefactoren [31]	Well-to-wheel
Car (diesel, medium size) [kgCO2e/km]	0.18	CO2emissiefactoren [31]	Well-to-wheel
Car (electric, green electricity, medium size) [kgCO2e/km]	0.002	CO2emissiefactoren [31]	Well-to-wheel
Freight transport			
Truck 10-20 ton, average laden [kgCO2e/kg-km]	0.256	CO2emissiefactoren [31]	Well-to-wheel
Truck trailer diesel [kgCO2e/kg-km]	0.117	CE Delft Stream [110]	Well-to-wheel
Tractor trailer electric [kgCO2e/kg-km]	0.117	CE Delft Stream [110]	Well-to-wheel
Truck trailer light diesel [kgCO2e/kg-km]	0.260	CE Delft Stream [110]	Well-to-wheel
Truck trailer light LNG [kgCO2e/kg-km]	0.236	CE Delft Stream [110]	Well-to-wheel
Vans average diesel (up to 3.5 tonnes) [kgCO2e/kg-km]	0.579	DEFRA [43]	By transport of a 3PL, well-to-wheel
Vans average petrol (up to 3.5 tonnes) [kgCO2e/kg-km]	0.754	DEFRA [43]	By transport of a 3PL, well-to-wheel
	0.251	DEFRA [43]	By transport of a 3PL, well-to-wheel
Vans average BEV (up to 3.5 tonnes) [kgCO2e/kg-km]	0.231	DEF104 [45]	by transport of a 51 E, well to wheel

# F

## Cost data

#### This appendix provides information for the calculation of the economic indicators.

#### Table F.1: Cost data

Data description	Figures	Database/source	Assumptions/additional comments
Retailer costs			
Warehouse lease costs [EU/m2/year]	1044	Statista [173]	This is an average value for large warehouses (area >5000 m2)
Marchouse rease costs (207 m27 year)	1011	Statista [110]	at specified location.
Brick-and-mortar rental costs [EUR/m2/year]	2025	Cushman & Wakefield [38]	This is the average of the range 1600-2450 EUR/m2/year,
briek and moral femal costs [2010/m2/year]	2020	Susimum et Mukenend [56]	specified for the store location.
Equipment costs			
Costs forklift [EUR/unit/year]	11328	[121]	A commonly used type of forklift is a reach truck. Price ranges from 600-1500
costs torkint [EOR/ unit/ year]	11520	[121]	dollars per month. Price data of base year (2020) used for conversion of dollars to euros.
			A conveyor belt costs 1570-5060 dollars on average, assuming the belt
Costs conveyor belt [EUR/unit/year]	995	[1, 197]	lasts three years. The average value of this range is assumed. Price data of
			base year used for conversion dollar to euros.
Containing and a state of the s	2100	1101	Price range between 4000-10000 dollar per equipment. Assumed replacement after
Costs manual picker [EUR/unit/year]	2100	[13]	3 years. Price data of base year used for conversion dollar to euros.
Transport costs			
Van - average [EUR/kg-km]	0.007601	KiM [187]	Diesel fueled, includes fixed, variable, staff, mode-specific, and general operating costs.
Electric van [EU/kg-km]	0.011934	Adapted from RWS, KiM [41, 187]	Cost are assumed to be 57% higher compared to an average diesel truck
Truck trailer [EU/kg-km]	0.000166	KiM [187]	Diesel fueled truck trailer for bulk cargo, includes fixed, variable, staff, mode-specific, and general operating costs
Electric truck trailer [EU/kg-km]	0.002606	Adapted from RWS, KiM [41, 187]	Cost are assumed to be 57% higher compared to an average diesel truck
Hydrogen truck trailer (7.5-19t) [EU/kg-km]	0.000498	Adapted from RWS, KiM [41, 187]	Cost are assumed to be three times higher compared to an average diesel truck.
Transport fee parcel at collection point [eu/parcel]	12.3-21.6	[89]	Ranges from 1-2 items to 9-10 items per returned parcel.
Labor costs			
Minimum hourly wage (Spain) [EUR/hour]	5.94	Statista [74]	Assumed a 40-hour workweek. The value is adapted to the base year.
Minimum hourly wage (Netherlands) [EUR/hour]	9.54	Rijksoverheid [156]	Assumed a 40-hour workweek. The value is adapted to the base year.
Overhead expenses (Spain) [%]	29.9	Eurodev [55]	Percent of the gross-based salary.
Overhead expenses (Netherlands) [%]	24	Eurodev [56]	Percent of the gross-based salary.
Cost of receiving a product [EUR/item]	2.25	[89]	recent of the gross-based sharty.
Cost of inspection and storage returns [EUR/item]	4.5	[89]	
Disposal costs			
Landfill charge (Spain) [EUR/tonne]	47.1	European Commission [69]	Landfill charge for base year
Incineration charge (Spain) [EUR/tonne]	23.6	European Commission [69]	Disposal charge for base year
Cardboard recycling [EUR/tonne]	104	Eurostat [73]	
Plastic recycling [EUR/tonne]	247	Eurostat [73]	
Mechanical recycling closed loop traditional [EU/kg-output]	0.7	[122]	Ranges from 500-900 euros per ton, so the average value is assumed.

## G

### Performance gaps

This section presents the scenarios' relative performance *RP* values and the performance gaps *M* based on the scores from section 7.3.

The relative performance *RP* and performance gap are calculated using the theory described in section 3.5. The relative performance expresses the ratio between the actual and target performance. So, the actual performance of a scenario in section 7.3 is divided by the target performance, which is the benchmark score on the CI. A higher value of *RP* illustrates a closer actual performance relative to the target performance, expressed by the benchmark.

The performance gap *M* represents the difference in performance between the actual and target values, calculated by subtracting the actual performance from the target performance. The best performance is achieved when the values of all indicators are minimized, so a lower value for this represents a better performance score [49]. As seen in Table G.1, the extreme variant of scenario 1 has the lowest performance gap in this table, and the extreme variant of scenario 3 is the largest.

Table G.1: Overview of relative performance and performance gap, source: (Author)

	Relative performance RP	Performance gap M
Benchmark	100.00%	0.00
Base case	75.58%	0.45
Scenario 1. Moderate shifting customer behavior	78.13%	0.40
Scenario 1. Extreme shifting customer behavior	85.80%	0.26
Scenario 2. Moderate green supply chain management	71.41%	0.53
Scenario 2. Extreme green supply chain management	74.95%	0.46
Scenario 3. Moderate economic attractiveness of renewables	72.99%	0.50
Scenario 3. Extreme economic attractiveness of renewables	69.91%	0.56
Scenario 4. Moderate governmental support to a circular business model	77.61%	0.41
Scenario 4. Extreme governmental support to a circular business model	77.71%	0.41
Scenario 5. Moderate technological innovations	75.33%	0.46
Scenario 5. Extreme technological innovations	80.51%	0.36

Н

## Uncertainty and sensitivity analysis

#### H.1. Rescaled pairwise comparison matrix

In subsection 7.2.5, the weights of the CI were rescaled by asking different experts. The weights presented in Table 7.4 are derived based on the pairwise comparison matrix presented in Table H.1. The data from the comparison matrix originates from a group of 3 experts. The consistency of this matrix is checked and below the threshold value of 0.10, indicating statistically consistent weights [142].

With  $V_1$ : packaging emissions,  $V_2$ : transport emissions,  $V_3$ : warehouse and equipment emissions,  $V_4$ : product recovery emissions,  $V_5$ : brick-and-mortar emissions,  $V_6$ : transport costs,  $V_7$ : warehouse and equipment costs,  $V_8$ : labor costs,  $V_9$ : recovery costs,  $V_{10}$ : brick-and-mortar costs.

	Vl	V2	V3	V4	V5	V6	V7	V8	V9	V10
V1	1	1/5	1/3	1/2	1/2	1/4	1/4	1/2	1/4	1/2
V2	5	1	5	4	5	1/2	1/2	3	2	2
<b>V3</b>	3	1/5	1	1/4	2	1/5	1/3	1	1/3	1
V4	2	1/4	4	1	4	1/5	1/4	1/2	1/2	1
V5	2	1/5	1/2	1/4	1	1/6	1/3	1/3	1/4	1/3
V6	4	2	5	5	6	1	2	3	1	3
V7	4	2	3	4	3	1/2	1	1/2	1	1
<b>V8</b>	2	1/3	1	2	3	1/3	2	1	3	2
V9	4	1/2	3	2	4	1	1	1/3	1	1/2
V10	2	1/2	1	1	3	1/3	1	1/2	2	1

Table H.1: Pairwise comparison matrix rescaled weights, source: (Author)

#### H.2. Data on the sensitivity analysis

This appendix contains the data of the SA performed by changing several input variables. The following variables were changed: equipment costs, warehouse emission value, incineration costs, and landfill costs. First, the raw data is presented, followed by the aggregated values in subsection H.2.2.

#### H.2.1. Raw data after increasing and decreasing the variables

The following section gives the raw data for each of the values  $V_1$ : packaging emissions,  $V_2$ : transport emissions,  $V_3$ : warehouse and equipment emissions,  $V_4$ : product recovery emissions,  $V_5$ : brick-and-mortar emissions,  $V_6$ : transport costs,  $V_7$ : warehouse and equipment costs,  $V_8$ : labor costs,  $V_9$ : recovery costs,  $V_{10}$ : brick-and-mortar costs.

- Scenario 1. Shifting customer behavior
- Scenario 2. Green supply chain management
- Scenario 3. Economic attractiveness of renewables
- Scenario 4. Governmental support for circularity
- Scenario 5. Technological innovations

The equipment costs are increased by 25% in Table H.2 and decreased by 25% in Table H.3. The transport cost for electric vans is increased by 25% in Table H.4 and decreased accordingly in Table H.5. The warehouse emission value is increased by 25% in Table H.6 and decreased by 25% in Table H.7. In Table H.8, the landfill costs are increased by 25% and decreased by 25% in Table H.9. Lastly, in Table H.10, the incineration costs are increased by 25%. In Table H.11, the raw data is visible after decreasing these values by 25%.

Table H.2: Raw data after increasing the equipment costs by 25%, source: (Author)

	V1	V2	V3	V4	V5	V6	V7	V8	V9	V10
	[kgCO2e/	[kgCO2e/	[kgCO2e/	[kgCO2e/	[kgCO2e/	[euro	[euro/	[euro/	[euro/	[euro/
	returned product]									
Benchmark	0.0161	14.4975	0.7427	0.0080	0.4844	0.2302	12.3998	5.0962	0.4967	1.9026
Base case overall performance	0.0790	32.2167	1.0922	0.0114	0.7124	0.2302	12.3998	5.0962	0.4967	1.9026
Scenario 1. Moderate	0.0639	32.3545	1.0834	0.0107	0.7184	0.2683	11.2998	5.0891	0.4941	1.9187
Scenario 1. Extreme	0.0452	31.5089	1.0786	0.0110	0.7192	0.2272	10.2459	5.0882	0.5071	1.9207
Scenario 2. Moderate	0.0632	30.5256	0.8396	0.0103	0.5343	0.2683	14.0891	5.0962	0.5149	2.4734
Scenario 2. Extreme	0.0457	29.0887	0.5870	0.0130	0.3562	0.3068	13.3188	5.0962	0.5016	2.6636
Scenario 3. Moderate	0.0752	30.2375	0.9911	0.0109	0.6412	0.2554	13.0198	5.3510	0.5249	1.9977
Scenario 3. Extreme	0.0765	27.2687	0.8396	0.0109	0.5343	0.2892	13.6398	5.6058	0.5568	2.0929
Scenario 4. Moderate	0.0306	32.2167	0.9455	0.0099	0.6087	0.2302	13.0822	5.1007	0.5709	2.0081
Scenario 4. Extreme	0.0157	32.2167	0.7932	0.0095	0.5013	0.2302	13.3913	5.1007	0.6231	2.0559
Scenario 5. Moderate	0.0739	32.4684	1.1168	0.0118	0.7131	0.2334	12.6796	5.1159	0.5196	1.9044
Scenario 5. Extreme	0.0760	29.0975	1.0813	0.0098	0.7188	0.2314	12.2762	5.0875	0.4913	1.9196

#### Table H.3: Raw data after decreasing the equipment costs by 25%, source: (Author)

	V1	V2	V3	V4	V5	V6	V7	V8	V9	V10
	[kgCO2e/	[kgCO2e/	[kgCO2e/	[kgCO2e/	[kgCO2e/	[euro	[euro/	[euro/	[euro/	[euro/
	returned product]									
Benchmark	0.0161	14.4975	0.7427	0.0074	0.4844	0.2302	12.3589	5.0962	0.5115	1.9026
Base case overall performance	0.0748	32.2167	1.0922	0.0106	0.7124	0.2302	12.3589	5.0962	0.5115	1.9026
Scenario 1. Moderate	0.0661	32.3545	1.0834	0.0109	0.7184	0.2294	11.2592	5.0890	0.5167	1.9187
Scenario 1. Extreme	0.0512	31.5089	1.0786	0.0111	0.7192	0.2272	10.2055	5.0882	0.5324	1.9207
Scenario 2. Moderate	0.0623	30.5256	0.8396	0.0112	0.5343	0.2683	14.0481	5.0960	0.5026	2.4734
Scenario 2. Extreme	0.0455	29.0887	0.5870	0.0105	0.3562	0.3068	13.2779	5.0962	0.5102	2.6636
Scenario 3. Moderate	0.0755	30.2375	0.9911	0.0132	0.6412	0.2554	12.9768	5.3510	0.5420	1.9977
Scenario 3. Extreme	0.0766	27.2687	0.8396	0.0129	0.5343	0.2892	13.5948	5.6058	0.5335	2.0929
Scenario 4. Moderate	0.0315	32.2167	0.9460	0.0106	0.6087	0.2310	13.0411	5.1007	0.5869	2.0081
Scenario 4. Extreme	0.0162	32.2167	0.0097	0.0097	0.5013	0.2302	13.3501	5.1007	0.6132	2.0559
Scenario 5. Moderate	0.0789	32.4684	1.1168	0.0101	0.7131	0.2334	12.6378	5.1159	0.5181	1.9044
Scenario 5. Extreme	0.0786	29.0975	1.0813	0.0101	0.7188	0.2310	12.2357	5.0875	0.5109	1.9196

#### Table H.4: Raw data after increasing the transport cost of electric vans by 25%, source: (Author)

	V1	V2	V3	V4	V5	V6	V7	V8	V9	V10
	[kgCO2e/	[kgCO2e/	[kgCO2e/	[kgCO2e/	[kgCO2e/	[euro	[euro/	[euro/	[euro/	[euro/
	returned product]									
Benchmark	0.016	14.498	0.743	0.008	0.484	0.230	12.400	5.096	0.497	1.903
Base case overall performance	0.079	32.217	1.092	0.011	0.712	0.230	12.400	5.096	0.497	1.903
Scenario 1. Moderate	0.062	32.355	1.083	0.012	0.718	0.229	11.280	5.089	0.508	1.919
Scenario 1. Extreme	0.046	31.509	1.079	0.010	0.719	0.227	10.226	5.085	0.517	1.921
Scenario 2. Moderate	0.063	30.526	0.840	0.010	0.534	0.295	14.089	5.096	0.515	2.473
Scenario 2. Extreme	0.046	29.089	0.587	0.013	0.356	0.361	13.319	5.096	0.502	2.664
Scenario 3. Moderate	0.078	30.237	0.991	0.010	0.641	0.263	12.998	5.351	0.511	1.998
Scenario 3. Extreme	0.077	27.269	0.840	0.012	0.534	0.308	13.617	5.606	0.559	2.093
Scenario 4. Moderate	0.031	32.217	0.945	0.010	0.609	0.230	13.082	5.101	0.571	2.008
Scenario 4. Extreme	0.016	32.217	0.793	0.010	0.501	0.230	13.391	5.101	0.623	2.056
Scenario 5. Moderate	0.072	32.468	1.117	0.012	0.713	0.233	12.659	5.116	0.502	1.904
Scenario 5. Extreme	0.077	29.097	1.081	0.011	0.719	0.231	12.256	5.087	0.514	1.920

#### H.2.2. Aggregated values CI

The SA increases and decreases the equipment costs, electric van costs, warehouse emission value, incineration costs, and landfill costs by 25%. In the main text, the correlation coefficient is presented based on the outcomes of this analysis. The values of the CI are presented in Table H.12.

	V1	V2	V3	V4	V5	V6	V7	V8	V9	V10
	[kgCO2e/	[kgCO2e/	[kgCO2e/	[kgCO2e/	[kgCO2e/	[euro	[euro/	[euro/	[euro/	[euro/
	returned product]									
Benchmark	0.016	14.498	0.743	0.008	0.484	0.230	12.400	5.096	0.508	1.903
Base case overall performance	0.079	32.217	1.092	0.011	0.712	0.230	12.400	5.096	0.508	1.903
Scenario 1. Moderate	0.062	32.355	1.083	0.012	0.718	0.229	11.280	5.089	0.500	1.919
Scenario 1. Extreme	0.046	31.509	1.079	0.010	0.719	0.227	10.226	5.085	0.491	1.921
Scenario 2. Moderate	0.063	30.526	0.840	0.010	0.534	0.295	14.089	5.096	0.515	2.473
Scenario 2. Extreme	0.046	29.089	0.587	0.013	0.356	0.361	13.319	5.096	0.502	2.664
Scenario 3. Moderate	0.078	30.237	0.991	0.010	0.641	0.263	12.998	5.351	0.511	1.998
Scenario 3. Extreme	0.077	27.269	0.840	0.012	0.534	0.308	13.617	5.606	0.559	2.093
Scenario 4. Moderate	0.031	32.217	0.945	0.010	0.609	0.230	13.082	5.101	0.571	2.008
Scenario 4. Extreme	0.016	32.217	0.793	0.010	0.501	0.230	13.391	5.101	0.623	2.056
Scenario 5. Moderate	0.072	32.468	1.117	0.012	0.713	0.233	12.659	5.116	0.514	1.904
Scenario 5. Extreme	0.077	29.097	1.081	0.011	0.719	0.231	12.256	5.087	0.515	1.920

#### Table H.5: Raw data after decreasing the transport cost of electric vans by 25%, source: (Author)

#### Table H.6: Raw data after increasing the warehouse emissions value by 25%, source: (Author)

	V1	V2	V3	V4	V5	V6	V7	V8	V9	V10
	[kgCO2e/	[kgCO2e/	[kgCO2e/	[kgCO2e/	[kgCO2e/	[euro	[euro/	[euro/	[euro/	[euro/
	returned product]									
Benchmark	0.0160	14.4975	0.9144	0.0078	0.4844	0.2302	12.3793	5.0962	0.5243	1.9026
Base case overall performance	0.0724	32.2167	1.3447	0.0111	0.7124	0.2302	12.3793	5.0962	0.5243	1.9026
Scenario 1. Moderate	0.0631	32.3545	1.3339	0.0118	0.7184	0.2294	11.2795	5.0891	0.5039	1.9187
Scenario 1. Extreme	0.0483	31.5089	1.3280	0.0096	0.7192	0.2272	10.2257	5.0853	0.5328	1.9207
Scenario 2. Moderate	0.0616	30.5256	1.0290	0.0113	0.5343	0.2683	14.0686	5.0962	0.4937	2.4734
Scenario 2. Extreme	0.0457	29.0887	0.7133	0.0112	0.3562	0.3068	13.2983	5.0962	0.4835	2.6636
Scenario 3. Moderate	0.0779	30.2375	1.2184	0.0126	0.6412	0.2554	12.9768	5.3510	0.5492	1.9977
Scenario 3. Extreme	0.0767	27.2687	1.0290	0.0128	0.5343	0.2892	13.6173	5.6058	0.5374	2.0929
Scenario 4. Moderate	0.0297	32.2167	1.1613	0.0103	0.6087	0.2302	13.0617	5.1007	0.5708	2.0081
Scenario 4. Extreme	0.0159	32.2167	0.9709	0.0094	0.5013	0.2302	13.3707	5.1007	0.6260	2.0559
Scenario 5. Moderate	0.0759	32.4684	1.3751	0.0103	0.7131	0.2334	12.6587	5.1159	0.5469	1.9044
Scenario 5. Extreme	0.0787	29.0975	1.3313	0.0101	0.7188	0.2314	12.2560	5.0875	0.5102	1.9196

#### Table H.7: Raw data after decreasing the warehouse emissions value by 25%, source: (Author)

	V1	V2	V3	V4	V5	V6	V7	V8	V9	V10
	[kgCO2e/	[kgCO2e/	[kgCO2e/	[kgCO2e/	[kgCO2e/	[euro	[euro/	[euro/	[euro/	[euro/
	returned product]									
Benchmark	0.0157	14.4975	0.5709	0.0073	0.4844	0.2302	12.3793	5.0962	0.5291	1.9026
Base case overall performance	0.0762	32.2167	0.8396	0.0105	0.7124	0.2302	12.3793	5.0962	0.5291	1.9026
Scenario 1. Moderate	0.0620	32.3545	0.8328	0.0128	0.7184	0.2294	11.3791	5.0891	0.5290	1.9187
Scenario 1. Extreme	0.0463	31.5089	0.8292	0.0109	0.7192	0.2272	10.2257	5.0853	0.5110	1.9207
Scenario 2. Moderate	0.0616	30.5256	0.6502	0.0114	0.5343	0.2683	14.0686	5.0962	0.5134	2.4734
Scenario 2. Extreme	0.0465	29.0887	0.4607	0.0113	0.3562	0.3068	13.2983	5.0962	0.4887	2.6636
Scenario 3. Moderate	0.0741	30.2375	0.7638	0.0111	0.6412	0.2554	12.9768	5.3510	0.5157	1.9977
Scenario 3. Extreme	0.0800	27.2687	0.6502	0.0116	0.5343	0.2892	13.6173	5.6058	0.5584	2.0929
Scenario 4. Moderate	0.0296	32.2167	0.7297	0.0102	0.6087	0.2302	13.0617	5.1007	0.5803	2.0081
Scenario 4. Extreme	0.0156	32.2167	0.6154	0.0094	0.5013	0.2302	13.3707	5.1007	0.6224	2.0559
Scenario 5. Moderate	0.0760	32.4684	0.8585	0.0111	0.7131	0.2334	12.6587	5.1159	0.5203	1.9044
Scenario 5. Extreme	0.0782	29.0975	0.8312	0.0096	0.7188	0.2314	12.2560	5.0875	0.4979	1.9196

#### Table H.8: Raw data after increasing the landfill cost by 25%, source: (Author)

	V1	V2	V3	V4	V5	V6	V7	V8	V9	V10
	[kgCO2e/	[kgCO2e/	[kgCO2e/	[kgCO2e/	[kgCO2e/	[euro	[euro/	[euro/	[euro/	[euro/
	returned product]									
Benchmark	0.0161	14.4975	0.7427	0.0071	0.4844	0.2302	12.3793	5.0962	0.5121	1.9026
Base case overall performance	0.0765	32.2167	1.0922	0.0102	0.7124	0.2302	12.3793	5.0962	0.5121	1.9026
Scenario 1. Moderate	0.0613	32.3545	1.0834	0.0121	0.7184	0.2294	11.2795	5.0891	0.5032	1.9187
Scenario 1. Extreme	0.0478	31.5089	1.0786	0.0108	0.7192	0.2272	10.2257	5.0853	0.5045	1.9207
Scenario 2. Moderate	0.0610	30.5256	0.8396	0.0132	0.5343	0.2683	14.0686	5.0962	0.5048	2.4734
Scenario 2. Extreme	0.0474	29.0887	0.5870	0.0115	0.3562	0.3068	13.2983	5.0962	0.5186	2.6636
Scenario 3. Moderate	0.0750	30.2375	0.9911	0.0114	0.6412	0.2554	12.9768	5.3510	0.5354	1.9977
Scenario 3. Extreme	0.0799	27.2687	0.8396	0.0122	0.5343	0.2892	13.6173	5.6058	0.5379	2.0929
Scenario 4. Moderate	0.0282	32.2167	0.9455	0.0099	0.6087	0.2302	13.0617	5.1007	0.5834	2.0081
Scenario 4. Extreme	0.0156	32.2167	0.7932	0.0095	0.5013	0.2302	13.3707	5.1007	0.6254	2.0559
Scenario 5. Moderate	0.0759	32.4684	1.1168	0.0100	0.7131	0.2334	12.6587	5.1159	0.5019	1.9044
Scenario 5. Extreme	0.0753	29.0975	1.0813	0.0095	0.7188	0.2314	12.2560	5.0875	0.4884	1.9196

#### Table H.9: Raw data after decreasing the landfill cost by 25%, source: (Author)

	V1	V2	V3	V4	V5	V6	V7	V8	V9	V10
	[kgCO2e/	[kgCO2e/	[kgCO2e/	[kgCO2e/	[kgCO2e/	[euro	[euro/	[euro/	[euro/	[euro/
	returned product]									
Benchmark	0.0161	14.4975	0.7427	0.0071	0.4844	0.2302	12.3793	5.0962	0.4743	1.9026
Base case overall performance	0.0765	32.2167	1.0922	0.0102	0.7124	0.2302	12.3793	5.0962	0.4743	1.9026
Scenario 1. Moderate	0.0613	32.3545	1.0834	0.0121	0.7184	0.2294	11.2795	5.0891	0.4695	1.9187
Scenario 1. Extreme	0.0478	31.5089	1.0786	0.0108	0.7192	0.2272	10.2257	5.0853	0.4761	1.9207
Scenario 2. Moderate	0.0610	30.5256	0.8396	0.0132	0.5343	0.2683	14.0686	5.0962	0.4832	2.4734
Scenario 2. Extreme	0.0474	29.0887	0.5870	0.0115	0.3562	0.3068	13.2983	5.0962	0.5145	2.6636
Scenario 3. Moderate	0.0750	30.2375	0.9911	0.0114	0.6412	0.2554	12.9768	5.3510	0.5293	1.9977
Scenario 3. Extreme	0.0799	27.2687	0.8396	0.0122	0.5343	0.2892	13.6173	5.6058	0.5560	2.0929
Scenario 4. Moderate	0.0282	32.2167	0.9455	0.0099	0.6087	0.2302	13.0617	5.1007	0.5648	2.0081
Scenario 4. Extreme	0.0156	32.2167	0.7932	0.0095	0.5013	0.2302	13.3707	5.1007	0.6051	2.0559
Scenario 5. Moderate	0.0759	32.4684	1.1168	0.0100	0.7131	0.2334	12.6587	5.1159	0.5187	1.9044
Scenario 5. Extreme	0.0753	29.0975	1.0813	0.0095	0.7188	0.2314	12.2560	5.0875	0.5135	1.9196

#### Table H.10: Raw data after increasing the incineration cost by 25%, source: (Author)

	V1	V2	V3	V4	V5	V6	V7	V8	V9	V10
	[kgCO2e/	[kgCO2e/	[kgCO2e/	[kgCO2e/	[kgCO2e/	[euro	[euro/	[euro/	[euro/	[euro/
	returned product]									
Benchmark	0.0159	14.4975	0.7427	0.0076	0.4844	0.2302	12.3793	5.0962	0.5060	1.9026
Base case overall performance	0.0753	32.2167	1.0922	0.0108	0.7124	0.2302	12.3793	5.0962	0.5060	1.9026
Scenario 1. Moderate	0.0646	32.3545	1.0834	0.0123	0.7184	0.2294	11.2795	5.0891	0.5046	1.9187
Scenario 1. Extreme	0.0455	31.5089	1.0786	0.0114	0.7192	0.2272	10.2257	5.0853	0.5193	1.9207
Scenario 2. Moderate	0.0617	30.5256	0.8396	0.0111	0.5343	0.2683	14.0686	5.0962	0.5036	2.4734
Scenario 2. Extreme	0.0454	29.0887	0.5870	0.0107	0.3562	0.3068	13.2983	5.0962	0.5177	2.6636
Scenario 3. Moderate	0.0753	30.2375	0.9911	0.0120	0.6412	0.2554	12.9768	5.3510	0.5228	1.9977
Scenario 3. Extreme	0.0754	27.2687	0.8396	0.0114	0.5343	0.2892	13.6173	5.6058	0.5348	2.0929
Scenario 4. Moderate	0.0301	32.2167	0.9455	0.0099	0.6087	0.2302	13.0617	5.1007	0.5745	2.0081
Scenario 4. Extreme	0.0157	32.2167	0.7932	0.0095	0.5013	0.2302	13.3707	5.1007	0.6229	2.0559
Scenario 5. Moderate	0.0778	32.4684	1.1168	0.0101	0.7131	0.2334	12.6587	5.1159	0.5346	1.9044
Scenario 5. Extreme	0.0750	29.0975	1.0813	0.0097	0.7188	0.2314	12.2560	5.0875	0.5062	1.9196

Benchmark         0.0161         14.4975         0.7427         0.0081         0.4844         0.2902         12.3793         5.0962         0.4969         1.9028           Base case overall performance         0.0010         32.2167         1.0922         0.0116         0.7124         0.2302         12.3793         5.0962         0.4969         1.9028           Scenario 1, Moderate         0.0626         32.3545         1.0824         0.0116         0.7124         0.2294         11.2795         5.0961         0.4969         1.9028           Scenario 1, Extreme         0.0494         31.5089         1.0786         0.0107         0.7182         0.2272         10.2277         5.0853         0.5206         1.9207           Scenario 2, Moderate         0.0460         23.0876         0.8376         0.0118         0.5343         0.2083         14.0686         5.0952         0.4983         2.4734           Scenario 2, Extreme         0.0460         23.087         0.8911         0.0108         0.6412         0.2554         12.9983         5.3510         0.5645         1.9977           Scenario 3, Moderate         0.0749         27.2687         0.8396         0.0112         0.5642         1.2983         5.5068         0.5577         2.0		V1	V2	V3	V4	V5	V6	V7	V8	V9	V10
Benchmark         0.0161         14.4975         0.7427         0.00161         0.4844         0.2302         12.3793         5.0962         0.4969         1.9026           Base case overall performance         0.0610         32.2167         1.0822         0.0116         0.7124         0.2302         12.3793         5.0962         0.4969         1.9026           Scenario 1. Moderate         0.0626         32.3545         1.0834         0.0126         0.7184         0.2294         11.2795         5.0891         0.5034         1.9026           Scenario 1. Extreme         0.0494         31.5099         10.786         0.0117         0.7182         0.2272         10.2257         5.0831         0.5206         1.3207           Scenario 2. Moderate         0.0464         0.3566         0.0118         0.5343         0.2683         14.0686         5.0962         0.4983         2.4734           Scenario 2. Moderate         0.04661         29.0887         0.6370         0.0112         0.3542         0.0088         13.2983         5.0962         0.4983         2.4734           Scenario 2. Moderate         0.04672         0.3375         0.9911         0.0109         0.6412         0.2554         12.9993         5.3510         0.5645 <td< td=""><td></td><td>[kgCO2e/</td><td>[kgCO2e/</td><td>[kgCO2e/</td><td>[kgCO2e/</td><td>[kgCO2e/</td><td>[euro</td><td>[euro/</td><td>[euro/</td><td>[euro/</td><td>[euro/</td></td<>		[kgCO2e/	[kgCO2e/	[kgCO2e/	[kgCO2e/	[kgCO2e/	[euro	[euro/	[euro/	[euro/	[euro/
Base case overall performance         0.0810         32.2167         1.0922         0.0116         0.7124         0.2392         12.3793         5.0962         0.4969         1.9025           Scenario 1. Moderate         0.0626         32.3545         1.0634         0.0126         0.7184         0.2294         11.2795         5.0891         0.5034         1.9187           Scenario 1. Extreme         0.0599         30.5256         0.0396         0.0107         0.7192         0.2272         10.2257         5.0853         0.5206         1.9207           Scenario 2. Moderate         0.0599         30.5256         0.0396         0.0112         0.3562         0.3663         14.0686         5.0962         0.4983         2.4734           Scenario 3. Moderate         0.0763         30.2375         0.9911         0.0108         0.6412         0.2554         12.9983         5.0502         0.4983         2.4734           Scenario 3. Moderate         0.0763         30.2375         0.9911         0.0108         0.6412         0.2554         12.9983         5.3510         0.5645         1.9997           Scenario 3. Moderate         0.0790         32.2167         0.9455         0.0117         0.6087         0.2302         13.6171         5.1007		returned product]									
Scenario 1. Moderate         0.0626         32.3545         1.0834         0.0126         0.7184         0.2294         11.2795         5.0891         0.5034         1.9187           Scenario 1. Entreme         0.0626         32.3545         1.0834         0.0107         0.7192         0.2272         10.2257         5.0891         0.5034         1.9187           Scenario 1. Entreme         0.0699         30.5256         0.4396         0.0107         0.7192         0.2272         10.2257         5.0853         0.5206         1.9207           Scenario 2. Entreme         0.0460         29.0887         0.5870         0.0112         0.3562         0.3068         13.2883         5.0962         0.4983         2.4734           Scenario 3. Moderate         0.0763         30.2375         0.9911         0.0106         0.6412         0.2554         12.9803         5.3510         0.5645         1.9977           Scenario 3. Extreme         0.0749         27.2687         0.4396         0.0127         0.50492         13.6173         5.6058         0.5577         2.0492           Scenario 4. Moderate         0.0254         32.2167         0.4345         0.0117         0.6067         0.2302         13.3017         5.1007         0.5770	Benchmark	0.0161	14.4975	0.7427	0.0081	0.4844	0.2302	12.3793	5.0962	0.4969	1.9026
Scenario I. Extreme         0.0494         31.5089         1.0786         0.0107         0.7192         0.2272         10.2257         5.0853         0.5206         1.9207           Scenario 2. Moderate         0.0599         30.5256         0.8396         0.0118         0.5343         0.2683         14.0686         5.0962         0.4983         2.4734           Scenario 2. Kreme         0.0464         29.0887         0.0570         0.0112         0.3562         0.3068         13.2983         5.0962         0.5088         2.6638           Scenario 3. Moderate         0.0763         30.2375         0.9911         0.0108         0.6412         0.2554         12.9983         5.3510         0.5645         1.9997           Scenario 4. Noderate         0.0763         30.2375         0.9911         0.0108         0.6412         0.2554         12.9983         5.3510         0.5645         1.9977           Scenario 4. Noterate         0.0290         32.2167         0.9455         0.0117         0.6087         0.2302         13.6173         5.1007         0.5710         2.0081           Scenario 4. Noderate         0.0154         32.2167         0.93455         0.5113         0.2302         13.3017         5.1007         0.6175	Base case overall performance	0.0810	32.2167	1.0922	0.0116	0.7124	0.2302	12.3793	5.0962	0.4969	1.9026
Scenario 2. Moderate         0.0599         30.5256         0.8396         0.0118         0.5343         0.2683         14.0686         5.0952         0.4983         2.4734           Scenario 2. Moderate         0.0669         29.0807         0.5870         0.0112         0.3562         0.3066         13.2893         5.0952         0.5298         2.6536           Scenario 3. Moderate         0.0763         30.2375         0.9911         0.0108         0.6412         0.2554         12.9983         5.3510         0.5645         1.9977           Scenario 3. Extreme         0.0749         27.2687         0.6386         0.0122         0.5343         0.2892         13.6173         5.6058         0.5577         2.0829           Scenario 4. Moderate         0.0290         32.2167         0.9455         0.0117         0.6067         0.2302         13.0617         5.1007         0.5770         2.0681           Scenario 4. Moderate         0.0154         32.2167         0.79452         0.0095         0.5013         0.2302         13.307         5.1007         0.6175         2.0584           Scenario 4. Moderate         0.0154         32.2167         0.7932         0.0095         0.5013         0.2302         13.307         5.1007 <td< td=""><td>Scenario 1. Moderate</td><td>0.0626</td><td>32.3545</td><td>1.0834</td><td>0.0126</td><td>0.7184</td><td>0.2294</td><td>11.2795</td><td>5.0891</td><td>0.5034</td><td>1.9187</td></td<>	Scenario 1. Moderate	0.0626	32.3545	1.0834	0.0126	0.7184	0.2294	11.2795	5.0891	0.5034	1.9187
Scenario 2. Extreme         0.0460         29.0887         0.5870         0.0112         0.3562         0.3068         13.2983         5.0962         0.5028         2.6836           Scenario 3. Moderate         0.0763         30.2375         0.9911         0.0108         0.6412         0.2554         12.9983         5.3510         0.5645         1.9977           Scenario 4. Kureme         0.0794         27.2687         0.8396         0.0125         0.5343         0.2892         13.6173         5.6058         0.5777         2.0929           Scenario 4. Kureme         0.0290         32.2167         0.9455         0.0117         0.6087         0.2302         13.0617         5.1007         0.5710         2.0081           Scenario 4. Extreme         0.0154         32.2167         0.7932         0.0095         0.5013         0.2302         13.3017         5.1007         0.6175         2.0581           Scenario 4. Extreme         0.0154         32.2167         0.7932         0.0095         0.5013         0.2302         13.3077         5.1007         0.6175         2.0581           Scenario 5. Moderate         0.0787         32.4684         1.1168         0.0100         0.7131         2.0334         12.6587         5.1159         0.51	Scenario 1. Extreme	0.0494	31.5089	1.0786	0.0107	0.7192	0.2272	10.2257	5.0853	0.5206	1.9207
Scenario 3. Moderate         0.0763         30.2375         0.9911         0.0108         0.6412         0.2554         12.9883         5.3510         0.5645         1.9977           Scenario 3. Extreme         0.0749         27.2687         0.8396         0.0125         0.5343         0.2892         13.6173         5.6058         0.5577         2.9329           Scenario 4. Moderate         0.0250         32.2167         0.9455         0.0117         0.6067         0.2302         13.0617         5.1007         0.5710         2.0681           Scenario 4. Extreme         0.0154         32.2167         0.7932         0.0095         0.513         0.2302         13.307         5.1007         0.6175         2.0559           Scenario 4. Extreme         0.0154         32.2167         0.7932         0.0095         0.513         0.2302         13.307         5.1007         0.6175         2.0559           Scenario 5. Moderate         0.0787         32.4684         1.1168         0.0100         0.7311         2.0324         5.108         1.9044	Scenario 2. Moderate	0.0599	30.5256	0.8396	0.0118	0.5343	0.2683	14.0686	5.0962	0.4983	2.4734
Scenario 3. Extreme         0.0749         27.2687         0.8396         0.0125         0.5343         0.2892         13.6173         5.6058         0.5577         2.0929           Scenario 4. Moderate         0.0290         32.2167         0.9455         0.0117         0.6067         0.2302         13.0617         5.1007         0.5710         2.0081           Scenario 4. Extreme         0.0154         32.2167         0.7932         0.0095         0.5013         0.2302         13.3707         5.1007         0.6175         2.0059           Scenario 5. Moderate         0.0787         32.4684         1.1168         0.0100         0.7131         0.2334         12.6587         5.1159         0.5108         1.9044	Scenario 2. Extreme	0.0460	29.0887	0.5870	0.0112	0.3562	0.3068	13.2983	5.0962	0.5028	2.6636
Scenario 4. Moderate         0.0290         32.2167         0.9455         0.0117         0.6087         0.2302         13.0617         5.1007         0.5710         2.0819           Scenario 4. Extreme         0.0154         32.2167         0.7332         0.0095         0.5013         0.2302         13.3617         5.1007         0.5710         2.0819           Scenario 4. Extreme         0.0154         32.2167         0.7332         0.0095         0.5013         0.2302         13.3707         5.1007         0.6175         2.0859           Scenario 5. Moderate         0.0787         32.4684         1.1168         0.0100         0.7311         0.2334         12.6587         5.1159         0.5108         1.9044	Scenario 3. Moderate	0.0763	30.2375	0.9911	0.0108	0.6412	0.2554	12.9983	5.3510	0.5645	1.9977
Scenario 4. Extreme         0.0154         32.2167         0.0995         0.0095         0.5013         0.2302         13.3707         5.1007         0.6175         2.0559           Scenario 5. Moderate         0.0787         32.4684         1.1168         0.0100         0.7131         0.2334         12.6587         5.1159         0.5108         1.9044	Scenario 3. Extreme	0.0749	27.2687	0.8396	0.0125	0.5343	0.2892	13.6173	5.6058	0.5577	2.0929
Scenario 5. Moderate 0.0787 32.4684 1.1168 0.0100 0.7131 0.2334 12.6587 5.1159 0.5108 1.9044	Scenario 4. Moderate	0.0290	32.2167	0.9455	0.0117	0.6087	0.2302	13.0617	5.1007	0.5710	2.0081
	Scenario 4. Extreme	0.0154	32.2167	0.7932	0.0095	0.5013	0.2302	13.3707	5.1007	0.6175	2.0559
	Scenario 5. Moderate	0.0787	32.4684	1.1168	0.0100	0.7131	0.2334	12.6587	5.1159	0.5108	1.9044
Scenario 5. Extreme 0.0754 29.0975 1.0813 0.0100 0.7188 0.2314 12.2560 5.0875 0.5048 1.9196	Scenario 5. Extreme	0.0754	29.0975	1.0813	0.0100	0.7188	0.2314	12.2560	5.0875	0.5048	1.9196

Table H.11: Raw data after decreasing the incineration cost by 25%, source: (Author)

Table H.12: Aggregated scores CI after performing the sensitivity analysis, source: (Author)

	Equipment costs + 25%	Equipment cost - 25%	Transport cost electric van + 25%	Transport cost electric van -25%	Warehouse emission value +25%	Warehouse emission value -25%	Incineration cost +25%	Incineration cost -25%	Landfill cost +25%	Landfill cost -25%
Benchmark	1.857	1.809	1.863	1.841	1.819	1.820	1.837	1.858	1.860	1.862
Base case	1.386	1.392	1.389	1.371	1.374	1.376	1.390	1.394	1.389	1.381
Scenario 1. Moderate	1.412	1.450	1.442	1.457	1.442	1.411	1.444	1.438	1.433	1.434
Scenario 1. Extreme	1.535	1.513	1.544	1.544	1.529	1.527	1.533	1.532	1.519	1.528
Scenario 2. Moderate	1.325	1.299	1.328	1.348	1.319	1.309	1.296	1.321	1.321	1.317
Scenario 2. Extreme	1.363	1.353	1.365	1.426	1.391	1.388	1.373	1.365	1.384	1.388
Scenario 3. Moderate	1.343	1.295	1.368	1.342	1.301	1.344	1.331	1.328	1.327	1.318
Scenario 3. Extreme	1.267	1.238	1.282	1.225	1.250	1.252	1.261	1.244	1.271	1.246
Scenario 4. Moderate	1.406	1.365	1.407	1.405	1.399	1.389	1.392	1.398	1.398	1.382
Scenario 4. Extreme	1.438	1.468	1.439	1.438	1.435	1.432	1.432	1.420	1.431	1.436
Scenario 5. Moderate	1.342	1.360	1.361	1.352	1.336	1.343	1.365	1.356	1.345	1.366
Scenario 5. Extreme	1.451	1.434	1.431	1.437	1.430	1.442	1.451	1.426	1.444	1.446

## Scenario analysis

This appendix gives an in-depth overview of the performance of the different scenarios with the benchmark. A radar chart does this. The indicators are defined on the spokes of the radar. For each indicator, better performance is obtained when the value is higher, corresponding to values more on the outside of the axis in the radar. Values closer to the center are less preferred in terms of performance.  $V_1$ : packaging emissions,  $V_2$ : transport emissions,  $V_3$ : warehouse and equipment emissions,  $V_4$ : product recovery emissions,  $V_5$ : brick-and-mortar emissions,  $V_6$ : transport costs,  $V_7$ : warehouse and equipment costs,  $V_8$ : labor costs,  $V_9$ : recovery costs,  $V_{10}$ : brick-and-mortar costs.

- Scenario 1. Shifting customer behavior
- · Scenario 2. Green supply chain management
- Scenario 3. Economic attractiveness of renewables
- Scenario 4. Governmental support for circularity
- Scenario 5. Technological innovations



Figure I.1: Radar chart benchmark and base case performance, source: (Author)



Figure I.2: Radar chart benchmark and scenario 1, source: (Author)



Figure I.3: Radar chart benchmark and scenario 2, source: (Author)



Figure I.4: Radar chart benchmark and scenario 3, source: (Author)



Figure I.5: Radar chart benchmark and scenario 4, source: (Author)



Figure I.6: Radar chart benchmark and scenario 5, source: (Author)